

From elliptic multiple zeta values to modular graph functions: open and closed strings at one loop

Johannes Broedel, Oliver Schlotterer, Federico Zerbini

▶ To cite this version:

Johannes Broedel, Oliver Schlotterer, Federico Zerbini. From elliptic multiple zeta values to modular graph functions: open and closed strings at one loop. 2018. cea-01729196

HAL Id: cea-01729196 https://cea.hal.science/cea-01729196

Preprint submitted on 12 Mar 2018

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

From elliptic multiple zeta values to modular graph functions: open and closed strings at one loop

Johannes Broedel^{a,b}, Oliver Schlotterer^{b,c,d}, Federico Zerbini^{e,f}

^aInstitut für Mathematik und Institut für Physik, Humboldt-Universität zu Berlin IRIS Adlershof, Zum Großen Windkanal 6, 12489 Berlin, Germany

> ^bKAVLI Institute for Theoretical Physics, Kohn Hall, University of California, Santa Barbara, CA 93106, USA

^cMax-Planck-Institut für Gravitationsphysik, Albert-Einstein-Institut Am Mühlenberg 1, 14476 Potsdam, Germany

> ^dPerimeter Institute for Theoretical Physics 31 Caroline St N, Waterloo, ON N2L 2Y5, Canada

^eInstitut de Physique Théorique (IPhT), CEA-Saclay Orme des Merisiers batiment 774, 91191 Gif-sur-Yvette, France

^fMax-Planck-Institut für Mathematik, Vivatsgasse 7, 53111 Bonn, Germany

Abstract

We relate one-loop scattering amplitudes of massless open- and closed-string states at the level of their low-energy expansion. The modular graph functions resulting from integration over closed-string punctures are observed to follow from symmetrized open-string integrals through a tentative generalization of the single-valued projection known from genus zero.

Contents

1.1 Summary of results 1.2 1.2 Outline 1.2 2 Basics 2.1 2.1 Single-valued projection at tree level 2.2 A- and B-cycle eMZVs and iterated Eisenstein integrals 1.2 2.3 Modular graph functions 1 3 An open-string setup for graph functions 1 3.1 Definition of A- and B-cycle graph functions 1 3.2 Evaluating A-cycle graph functions 1 3.3 Evaluating B-cycle graph functions 1 3.3 Evaluating B-cycle graph functions 2 4 Open versus closed strings 2 4.1 Comparing relations among A-cycle graph functions with relations among modular graph functions 2 4.2 Modular graph functions from A-cycle graph functions 3 4.3 Modular graph functions from B-cycle graph functions 3 5 Non-planar A-cycle graph function 3
1.2 Outline 1.2 2 Basics 2.1 Single-valued projection at tree level 1.2 2.2 A- and B-cycle eMZVs and iterated Eisenstein integrals 1.2 2.3 Modular graph functions 1 3 An open-string setup for graph functions 1 3.1 Definition of A- and B-cycle graph functions 1 3.2 Evaluating A-cycle graph functions 1 3.3 Evaluating B-cycle graph functions 2 4 Open versus closed strings 2 4.1 Comparing relations among A-cycle graph functions with relations among modular graph functions from A-cycle graph functions 3 4.2 Modular graph functions from B-cycle graph functions 3 5 Non-planar A-cycle graph function 4
2 Basics 2.1 Single-valued projection at tree level 2.2 A- and B-cycle eMZVs and iterated Eisenstein integrals 2.3 Modular graph functions 2.3 Modular graph functions 1 3 An open-string setup for graph functions 3.1 Definition of A- and B-cycle graph functions 3.2 Evaluating A-cycle graph functions 3.3 Evaluating B-cycle graph functions 3.4 Open versus closed strings 4 Open versus closed strings 4.1 Comparing relations among A-cycle graph functions with relations among modular graph functions 2.2 Modular graph functions from A-cycle graph functions 3.3 Modular graph functions from B-cycle graph functions
 2 Basics 2.1 Single-valued projection at tree level
2.1 Single-valued projection at tree level
 2.2 A- and B-cycle eMZVs and iterated Eisenstein integrals
 2.3 Modular graph functions
 3 An open-string setup for graph functions 3.1 Definition of A- and B-cycle graph functions 3.2 Evaluating A-cycle graph functions 3.3 Evaluating B-cycle graph functions 4 Open versus closed strings 4.1 Comparing relations among A-cycle graph functions with relations among modular graph functions 4.2 Modular graph functions from A-cycle graph functions 3 Modular graph functions from B-cycle graph functions 5 Non-planar A-cycle graph function
 3.1 Definition of A- and B-cycle graph functions
 3.2 Evaluating A-cycle graph functions
 3.3 Evaluating <i>B</i>-cycle graph functions
 4 Open versus closed strings 4.1 Comparing relations among A-cycle graph functions with relations among modular graph functions
 4 Open versus closed strings 4.1 Comparing relations among A-cycle graph functions with relations among modular graph functions
 4.1 Comparing relations among A-cycle graph functions with relations among modular graph functions
ular graph functions 2 4.2 Modular graph functions from A-cycle graph functions 3 4.3 Modular graph functions from B-cycle graph functions 3 5 Non-planar A-cycle graph function
 4.2 Modular graph functions from A-cycle graph functions
 4.3 Modular graph functions from <i>B</i>-cycle graph functions
5 Non-planar 4-cycle graph function
5.1 Non-planar open-string integrals
5.2 Examples of non-planar A-cycle graph functions
5.2 Comparing α' -expansions of planar and non-planar integrals
5.5 Comparing a expansions of planar and non planar moegrais
6 Conclusions 4
A Translating between graphs and notations for modular graph functions 4
B Constant term of B-cycle eMZVs
C Expanding S-transformed A-cycle eMZVs 5
C.1 The Laurent polynomial
C.2 The q-expansion $\ldots \ldots 5$
D. Different Assess of iterated Discussion intermals
D Different navors of iterated Eisenstein integrals 3
D.1 Another convention for iterated Eisenstein integrals
D.2 Conversion between \mathcal{E}_0 and \mathcal{E}
D.3 Conversion between \mathcal{E} and \mathcal{G}
D.4 Examples of modular transformations
E A-cycle graph functions at weight five 5
F Relations between modular graph functions at weight six 5
G Explicit modular graph forms and modular graph functions
G 1 Cauchy Riemann derivatives
G.2 Modular graph functions at weight six

1 Introduction

Modular graph functions are building blocks for one-loop scattering amplitudes in closed-string theories at the one-loop level. They have been thoroughly investigated by D'Hoker, Green, Vanhove and other authors during the last couple of years [1–13] and arise from Feynman graphs of certain conformal scalar fields on the torus. Each modular graph function depends on the modular parameter of the torus and its modular invariance is inherited from the underlying closed-string setup. While the computation of their asymptotic expansion ¹ is itself cumbersome, they exhibit a variety of mathematical structures: modular graph functions are related by a network of algebraic identities and related to holomorphic Eisenstein series by differential equations with respect to the modular parameter. Even more, they satisfy certain eigenvalue equations involving the modular invariant Laplace operator.

Most interestingly for the purpose of this article, however, a first connection between elliptic multiple polylogarithms (as defined in refs. [14–16]) and modular graph functions was established in ref. [6]: The latter were written as special values of infinite sums of single-valued multiple polylogarithms, and these infinite sums are proposed in the reference to be a single-valued analogue of elliptic multiple polylogarithms². This connection extends an observation made for genus-zero (tree level) open- and closed-string amplitudes: closed-string tree amplitudes are conjectured to be obtained by acting with the so-called single-valued projection on the multiple zeta values appearing in their open-string counterparts [17–19]. The single-valued projection maps generic multiple zeta values to those instances which descend from single-valued polylogarithms at genus zero [20, 21].

At genus one (one-loop level), Enriquez's elliptic multiple zeta values [22] were shown to capture the low-energy expansion of the open superstring [23–25]. The results of [6] suggest to expect that modular graph functions are single-valued versions of Enriquez's elliptic multiple zeta values. However, the precise matching and thus the relation between open- and closed-string results at one-loop level is an open problem: First, the closed-string [6] and open-string literature [23–25] use different notions of elliptic polylogarithms. Second, the dependence of modular graph functions and elliptic multiple zeta values on the modular parameters of the respective genus-one surface is realized in rather different languages.

In the current article we are going to bridge the leftover gap between one-loop open- and closed-string amplitudes before integration over the respective modular parameters. We propose a setup which allows to relate certain building blocks of open-string amplitudes with modular graph functions. This accumulates evidence for a conjectural elliptic generalization of the single-valued projection known from genus zero. Simultaneously, this leads to a conjectural formalism to explicitly construct modular graph functions starting from open-string quantities. The results thus obtained pass all consistency checks and match previous partial expressions.

The main idea is to define open-string graph functions within an abelian version of one-loop open-string amplitudes. Despite the fact that the permissible string spectrum of Type-I opensuperstring theory does not contain an abelian gauge boson [26], we will consider a kinematical building block of the putative amplitude, which is non-trivial and well-defined for auxiliary abelian particles. In order to implement the abelian character of the auxiliary particles, the integration regions for open-string punctures are symmetrized in a convenient manner. The symmetrized open-string integrals of the abelian setup are the key to lining up the properties

¹As the modular parameter τ tends to $i\infty$ such that a homology cycle of the Riemann surfaces pinches.

 $^{^{2}}$ It is not demonstrated that the infinite sums studied in ref. [6] can be called single-valued elliptic multiple polylogarithms in the usual mathematical sense. This would be true if one can write them as finite linear combinations of products of elliptic multiple polylogarithms and their complex conjugates.

of the open-string genus-one Green function with its closed-string counterpart. In particular, the graphical organization of the low-energy expansion of open- and closed-string amplitudes in terms of open-string and modular graph functions agrees, which allows for direct comparison between constituents.



Figure 1: Context of a tentative generalization "esv" of the single-valued projection to genus one.

1.1 Summary of results

The notion of a *single-valued projection* applies to a variety of situations [27]. The most common examples are multiple zeta values (MZVs),

$$\zeta_{n_1, n_2, \dots, n_r} := \sum_{0 < k_1 < k_2 < \dots < k_r}^{\infty} k_1^{-n_1} k_2^{-n_2} \dots k_r^{-n_r} , \quad n_i \in \mathbb{N}^+, \ n_r \ge 2,$$
(1.1)

of weight $n_1+n_2+\ldots+n_r$ and depth r, which can be represented as multiple polylogarithms evaluated at unit argument. In contrast, single-valued MZVs³ descend from single-valued multiple polylogarithms at unit argument [21]. As explained in the reference, the single-valued projection formally denoted as

$$\operatorname{sv}(\zeta_{n_1,\dots,n_r}) = \zeta_{n_1,\dots,n_r}^{\operatorname{sv}}$$
(1.2)

maps generic MZVs (1.1) to their single-valued counterparts, e.g.

$$\zeta_{2k}^{\text{sv}} = 0, \qquad \zeta_{2k+1}^{\text{sv}} = 2\zeta_{2k+1}, \quad k \in \mathbb{N}^+$$

$$\zeta_{3,5}^{\text{sv}} = -10\zeta_3\zeta_5, \qquad \zeta_{3,5,3}^{\text{sv}} = 2\zeta_{3,5,3} - 2\zeta_3\zeta_{3,5} - 10\zeta_3^2\zeta_5.$$
(1.3)

As will be reviewed in the next section, the single-valued projection of MZVs appears naturally in relating tree-level scattering amplitudes of open and closed strings: the single-valued map acts on the MZVs arising in the low-energy expansion of open-string disk integrals and yields the closed-string integral over a punctured sphere. Correspondingly, it would be desirable to identify a similar map called "esv" for the elliptic version of multiple zeta values ω (to be defined and discussed below)

$$\operatorname{esv}(\omega(n_1,\ldots,n_r|\tau)) = \omega^{\operatorname{esv}}(n_1,\ldots,n_r|\tau).$$
(1.4)

at the one-loop level. As will be shown in this article, one-loop open- and closed-string amplitudes – expressed as open-string and modular graph functions, respectively – can be taken as a

 $^{^{3}}$ While the concept of single-valuedness is well defined for a function, the notion is – by slight abuse of nomenclature – also used for MZVs which are numbers.

starting point to propose an analogous single-valued projection of elliptic multiple zeta values (eMZVs), see figure 1. Accordingly, we are going to describe suitable operations on open-string graph functions, which conjecturally yield modular graph functions as their one-loop closed-string counterparts,

$$\operatorname{esv}\left(\begin{array}{c}\operatorname{open-string}\\\operatorname{graph}\operatorname{function}\end{array}\right) = \left(\begin{array}{c}\operatorname{closed-string}\operatorname{mo-}\\\operatorname{dular}\operatorname{graph}\operatorname{function}\end{array}\right).$$
(1.5)

We will provide examples of this correspondence up to and including the seventh subleading order in the low-energy expansion. In particular, starting from eq. (1.5), we will establish a new connection between building blocks of open- and closed-string four-point amplitudes

esv
$$M_4^{\text{open}}(s_{ij}|-\frac{1}{\tau}) = M_4^{\text{closed}}(s_{ij}|\tau)$$
. (1.6)

These functions of the modular parameters τ of the underlying Riemann surfaces result from integrating over the open- and closed-string punctures and yield the respective building blocks for amplitudes upon integration over τ . We will furthermore provide evidence that the planar open-string integral on the left hand side can be replaced by any of its non-planar counterparts, irrespective on how the four state insertions are distributed over the boundary of the worldsheet.

It is important to mention that a way to produce a single-valued projection of eMZVs (and therefore of open-string graph functions) already exists in the literature: it is based on their representation in terms of iterated integrals of Eisenstein series (as will be explained later in section 2), followed by the construction given in Francis Brown's papers [28] and [29]. Brown's construction maps iterated integrals of Eisenstein series to certain modular-invariant real-analytic functions whose coefficients are single-valued MZVs. So far, however, it remains conjectural that modular graph functions are contained in the image of this elliptic single-valued projection. We postpone the investigation of the relation between our single-valued projection and Brown's map to a sequel of the present work.

1.2 Outline

Several techniques and previous results entering the construction of this work are reviewed in section 2. First, a short review is given on the single-valued projection in the context of regular multiple zeta values, which appear at string tree level. Second, A- and B-cycle versions of eMZVs will be reviewed. As it will turn out, modular transformations are facilitated by representing A- and B-cycle elliptic multiple zeta values in the language of iterated integrals over Eisenstein series. Modular graph functions including some of their properties are introduced briefly.

In section 3, open-string graph functions are introduced. While starting from the so called A-cycle graph functions, it will turn out that finally B-cycle functions are the objects necessary for the construction of modular graph functions.

Once open-string graph functions are properly introduced, the comparison with modular graph functions can happen, and it is presented in section 4. Using several examples, we will finally arrive at a set of rules relating open-string graph functions to modular graph functions. As this is done at the level of relating eMZVs to what is believed to be a single-valued version thereof, the construction should constitute a representation of an elliptic single-valued projection.

Finally, non-planar analogues of the above open-string graph functions are introduced in section 5, generalizing our main result eq. (1.6) to admit the integrals for arbitrary non-planar four-point open-string amplitudes on the left hand side.

Various details and examples can be found in the appendices. In appendix A we provide a table allowing to translate our graphical notation to different notations for modular graph functions appearing in earlier articles on the subject.

2 Basics

2.1 Single-valued projection at tree level

In this section, we provide a brief review of the tree-level relations between open- and closedstring amplitudes and identify them as the single-valued projection in eq. (1.2).

Tree amplitudes among n massless open-string states can be represented by moduli-space integrals over punctured disks accompanied by partial amplitudes of the Yang–Mills field theory [30, 31]. The moduli-space integrals read

$$Z(\rho(1,2,\ldots,n) \mid \sigma(1,2,\ldots,n)) = \int_{D(\rho(1,2,\ldots,n))} \frac{\mathrm{d}z_1 \,\mathrm{d}z_2 \cdots \mathrm{d}z_n}{\mathrm{vol}(\mathrm{SL}(2,\mathbb{R}))} \frac{\prod_{i< j}^n |z_{ij}|^{-s_{ij}}}{\sigma(z_{12}z_{23}\ldots z_{n-1,n}z_{n,1})}, \qquad (2.1)$$

where z_i are the positions of the punctures on the boundary of a disk. The integral $Z(\cdot|\cdot)$ in eq. (2.1) is labeled by two permutations $\sigma, \rho \in S_n$ of the external legs $1, 2, \ldots, n$ which govern the cyclic product of $z_{ij} = z_i - z_j$ in the denominator and the integration domain

$$D(1, 2, \dots, n) = \{ (z_1, z_2, \dots, z_n) \in \mathbb{R}^n, -\infty < z_1 < z_2 < \dots < z_n < \infty \}.$$
 (2.2)

The division by the inverse volume vol(SL(2, \mathbb{R})) of the conformal Killing group can be implemented by dropping any three integrations, fixing the respective positions such as $(z_1, z_{n-1}, z_n) = (0, 1, \infty)$, and inserting the compensating Jacobian $z_{1,n-1}z_{1,n}, z_{n-1,n}$. Finally, the disk integrals eq. (2.1) depend on the lightlike momenta k_j of the external states $j = 1, 2, \ldots, n$ through the dimensionless Mandelstam variables⁴

$$s_{ij} := -\frac{\alpha'}{2} k_i \cdot k_j \tag{2.3}$$

involving the inverse string tension α' .

Tree-level amplitudes among massless closed-string states, in turn, comprise moduli-space integrals over punctured spheres,

$$W(\rho(1,2,\ldots,n) \,|\, \sigma(1,2,\ldots,n)) := \pi^{3-n} \int_{\mathbb{C}^n} \frac{\mathrm{d}^2 z_1 \,\mathrm{d}^2 z_2 \cdots \mathrm{d}^2 z_n}{\mathrm{vol}(\mathrm{SL}(2,\mathbb{C}))} \frac{\prod_{i
(2.4)$$

where both permutations $\rho, \sigma \in S_n$ label a cyclic product of z_{ij} or their complex conjugates. The inverse volume vol(SL(2, \mathbb{C})) suppresses three complex integrations and the normalization factor π^{3-n} is chosen for later convenience.

The low-energy regime of string amplitudes is encoded in the Taylor expansion of the disk and sphere integrals around small values of the inverse string tension α' and thus small values of the Mandelstam variables (2.3). The *w*'th order in the low-energy expansion beyond the

⁴Throughout this work, we will follow the normalization convention for α' which is tailored to the closed-string setup. The fully accurate normalization of open-string quantities can be restored by rescaling $\alpha' \to 4\alpha'$ [32].

respective field-theory amplitudes gives rise to MZVs eq. (1.1) of weight w [33, 34], for instance

$$s_{12}Z(1,2,3,4 \mid 1,2,4,3) = \exp\left(\sum_{n=2}^{\infty} \frac{\zeta_n}{n} \left[s_{12}^n + s_{23}^n - (s_{12} + s_{23})^n\right]\right)$$
(2.5)

$$s_{12}W(1,2,3,4|1,2,4,3) = \exp\left(2\sum_{k=1}^{\infty} \frac{\zeta_{2k+1}}{2k+1} \left[s_{12}^{2k+1} + s_{23}^{2k+1} - (s_{12}+s_{23})^{2k+1}\right]\right).$$
(2.6)

Generic examples of multiplicity $n \ge 5$ also involve MZVs of higher depth $r \ge 2$ [35,17], and the explicit polynomial dependence on the Mandelstam invariants can for instance be computed⁵ via polylogarithm manipulations [31], the Drinfeld associator [42] or a Berends–Giele recursion for a putative effective field theory of bi-colored scalars [43]. A machine-readable form of such results is available for download on the website [44].

Closed-string integrals (2.4) can in principle be assembled from squares of open-string integrals (2.1) through the Kawai–Lewellen–Tye (KLT) relations [45]. However, the KLT formula obscures the cancellation of various MZVs from the open-string constituents: From the all-order conjectures of ref. [17], closed-string integrals (2.4) are expected to be single-valued open-string integrals [18, 19],

$$W(\rho(1, 2, \dots, n) \,|\, \sigma(1, 2, \dots, n)) = \operatorname{sv} Z(\rho(1, 2, \dots, n) \,|\, \sigma(1, 2, \dots, n)) \,. \tag{2.7}$$

The MZVs in the image of the single-valued projection sv(...) are precisely the single-valued MZVs described in eqs. (1.2) and (1.3) above – in agreement with the four-point examples eqs. (2.5) and (2.6). As can be seen from eq. (2.7), the sv-projection trades the integration domain of the disk integral eq. (2.1) for an antiholomorphic cyclic denominator of a sphere integral (2.4).

2.2 A- and B-cycle eMZVs and iterated Eisenstein integrals

Several versions of eMZVs have been used in different contexts: when represented as special values of multiple elliptic polylogarithms (defined by Brown and Levin in [16]), they have made an appearance in the evaluation of the sunrise integral, see for instance [46–55], while when represented as the coefficients of the elliptic associator (defined by Enriquez in [56]), they have made an appearance in the one-loop open-string amplitudes. The latter is the context that we consider in this article; therefore our conventions are inspired by the string-theory setup in refs. [23–25]. A further comprehensive reference on eMZVs is Matthes's PhD thesis [57]. A-cycle eMZVs are defined as iterated integrals over the unit interval

$$\omega_A(n_1, n_2, \dots, n_r | \tau) := \int_{\substack{0 \le z_1 \le z_2 \le \dots \le z_r \le 1}} f^{(n_1)}(z_1, \tau) \, \mathrm{d}z_1 \, f^{(n_2)}(z_2, \tau) \, \mathrm{d}z_2 \, \dots \, f^{(n_r)}(z_r, \tau) \, \mathrm{d}z_r \,, \qquad (2.8)$$

where the integration path is taken to be on the real line⁶. Using the parametrization of the torus in figure 2, the integration domain in eq. (2.8) corresponds to the A-cycle and justifies the term "A-cycle eMZVs". Accordingly, iterated integrals along the B-cycle connecting the points

⁵Earlier work on α' -expansions at n = 5, 6, 7 points include [36–39], and the representation of five-point integrals as hypergeometric functions has been exploited in the all-order methods of refs. [40, 41].

⁶Homotopy-invariant completions of the integrands in eq. (2.8) are known from ref. [16].



Figure 2: Parametrization of a torus as a lattice $\mathbb{C}/(\mathbb{Z}+\tau\mathbb{Z})$ with modular parameter τ in the upper half plane and complex coordinate $z \equiv z+1 \equiv z+\tau$. The homology cycle drawn in red is mapped to the unit interval (0, 1) and referred to as the A-cycle. Accordingly, the second homology cycle mapped to the path from 0 to τ is known as the B-cycle.

0 and τ in figure 2 give rise⁷ to *B*-cycle eMZVs

$$\omega_B(n_1, n_2, \dots, n_r | \tau) := \int_{\substack{0 \le z_1 \le z_2 \le \dots \le z_r \le \tau}} f^{(n_1)}(z_1, \tau) \, \mathrm{d}z_1 \, f^{(n_2)}(z_2, \tau) \, \mathrm{d}z_2 \, \dots \, f^{(n_r)}(z_r, \tau) \, \mathrm{d}z_r \,.$$
(2.9)

The doubly-periodic integration kernels $f^{(n)}$ in eqs. (2.8) and (2.9) are defined by their generating series [23,24],

$$\exp\left(2\pi i\alpha \frac{\mathrm{Im}(z)}{\mathrm{Im}(\tau)}\right)\frac{\theta'(0,\tau)\theta(z+\alpha,\tau)}{\theta(z,\tau)\theta(\alpha,\tau)} = \sum_{n=0}^{\infty} \alpha^{n-1} f^{(n)}(z,\tau)\,,\tag{2.10}$$

where $\theta(z,\tau)$ denotes the odd Jacobi theta function, and the simplest instances are $f^{(0)}(z,\tau) = 1$ as well as $f^{(1)}(z,\tau) = \partial_z \log \theta(z,\tau) + 2\pi i \frac{\text{Im}(z)}{\text{Im}(\tau)}$. We refer to the number r of entries of eMZVs and the quantity $n_1 + n_2 + \ldots + n_r$ as their *length* and *weight*, respectively. Furthermore, the number of non-zero entries $n_j \neq 0$ of eMZVs will be referred to as their *depth*.

B-cycle eMZVs can be obtained from *A*-cycle eMZVs by the modular *S*-transformation, which sends $\tau \to -\frac{1}{\tau}$,

$$\omega_A(n_1, n_2, \dots, n_r | -\frac{1}{\tau}) = \tau^{n_1 + n_2 + \dots + n_r - r} \,\omega_B(n_1, n_2, \dots, n_r | \tau) \,. \tag{2.11}$$

Since the restriction of the kernels $f^{(n)}$ to the real line admits a Fourier-expansion in $q = e^{2\pi i \tau}$ spelt out in subsection 3.3.3 of ref. [23], the same is true for A-cycle eMZVs in eq. (2.8), and one can prove that the coefficients are given by $\mathbb{Q}[(2\pi i)^{\pm 1}]$ -linear combinations of MZVs [22].

By contrast, *B*-cycle eMZVs have the more complicated behavior near the cusp $\tau \to i\infty$ (or $q \to 0$) [22,58],

$$\omega_B(n_1, n_2, \dots, n_r | \tau) = \sum_{l=1-n_1-\dots-n_r}^r \tau^l \sum_{k=0}^\infty b_{k,l}(n_1, n_2, \dots, n_r) q^k , \qquad n_1, n_r \neq 1 , \qquad (2.12)$$

where the coefficients $b_{k,l}(n_1, n_2, ..., n_r)$ are $\mathbb{Q}[(2\pi i)^{\pm 1}]$ -linear combinations of MZVs. In the resulting expansion for S-transformed A-cycle eMZVs

$$\omega_A(n_1, n_2, \dots, n_r | -\frac{1}{\tau}) = \sum_{l=1-r}^{n_1 + \dots + n_r} (2\pi i\tau)^l \sum_{k=0}^{\infty} c_{k,l}(n_1, n_2, \dots, n_r) q^k , \qquad n_1, n_r \neq 1 , \quad (2.13)$$

⁷We think of eq. (2.9) as an integral over the straight path $[0, \tau] \subset \mathbb{C}$. Again, these integrals are not homotopy invariant, and their relation with the homotopy invariant version known from ref. [16] is more subtle than in the A-case. The interested reader is referred to [57].

it is crucial for later purposes to note that the coefficients $c_{k,l}(n_1, n_2, \ldots, n_r)$ are $\mathbb{Q}[2\pi i]$ -linear (rather than $\mathbb{Q}[(2\pi i)^{\pm 1}]$ -linear) combinations of MZVs. As will be proven in appendix C, all the negative powers of π can been absorbed into the negative powers of $2\pi i \tau$ in eq. (2.13).

2.2.1 Elliptic iterated integrals

In the same way as MZVs descend from multiple polylogarithms at unit argument, A-cycle eMZVs defined in eq. (2.8) are special cases of elliptic iterated integrals subject to the recursive definition [23]

$$\Gamma\left(\begin{smallmatrix} n_{1} & n_{2} & \dots & n_{r} \\ a_{1} & a_{2} & \dots & a_{r} \end{smallmatrix}; z \mid \tau\right) := \int_{0}^{z} \mathrm{d}t \, f^{(n_{1})}(t - a_{1}, \tau) \, \Gamma\left(\begin{smallmatrix} n_{2} & \dots & n_{r} \\ a_{2} & \dots & a_{r} \end{smallmatrix}; t \mid \tau\right)$$
(2.14)

with initial condition $\Gamma(; z|\tau) = 1$, integration path along the real line and real upper limit z. Accordingly,

$$\omega_A(n_1, n_2, \dots, n_r \,|\, \tau) = \Gamma\left(\begin{smallmatrix} n_r & \dots & n_2 & n_1 \\ 0 & \dots & 0 & 0 \end{smallmatrix}; 1 \,|\, \tau\right) \,. \tag{2.15}$$

$$\Gamma\left(\begin{smallmatrix}n\\z\end{smallmatrix};z\right) = (-1)^n \,\Gamma\left(\begin{smallmatrix}n\\0\end{smallmatrix};z\right) \tag{2.16}$$

$$\Gamma\left(\begin{smallmatrix} 1 & 0 & 1 \\ z & 0 & 0 \end{smallmatrix}; z\right) = 2\,\Gamma\left(\begin{smallmatrix} 0 & 0 & 2 \\ 0 & 0 & 0 \end{smallmatrix}; z\right) + \Gamma\left(\begin{smallmatrix} 0 & 2 & 0 \\ 0 & 0 & 0 \end{smallmatrix}; z\right) - 2\,\Gamma\left(\begin{smallmatrix} 0 & 1 & 1 \\ 0 & 0 & 0 \end{smallmatrix}; z\right) + \zeta_2\,\Gamma\left(\begin{smallmatrix} 0 \\ 0 \end{smallmatrix}; z\right) \,. \tag{2.17}$$

2.2.2 Iterated Eisenstein integrals

Given that the differential equation in appendix C.2 allows to relate eMZVs to Eisenstein series, it is natural to represent them in terms of iterated integrals in τ (or $q = e^{2\pi i \tau}$), see ref. [24] for the detailed formalism of iterated Eisenstein integrals⁸,

$$\begin{aligned} \mathcal{E}(k_1, k_2, \dots, k_r; \tau) &= 2\pi i \int_{\tau}^{i\infty} \mathrm{d}\tau_r \ \frac{\mathrm{G}_{k_r}(\tau_r)}{(2\pi i)^{k_r}} \, \mathcal{E}(k_1, k_2, \dots, k_{r-1}; \tau_r) \\ &= -\int_{0}^{q} \mathrm{dlog} \, q_r \ \frac{\mathrm{G}_{k_r}(q_r)}{(2\pi i)^{k_r}} \ \mathcal{E}(k_1, k_2, \dots, k_{r-1}; q_r) \\ &= (-1)^r \int_{0 \le q_1 \le q_2 \le \dots \le q_r \le q} \mathrm{dlog} \, q_1 \cdots \mathrm{dlog} \, q_r \ \frac{\mathrm{G}_{k_1}(q_1)}{(2\pi i)^{k_1}} \cdots \frac{\mathrm{G}_{k_r}(q_r)}{(2\pi i)^{k_r}} \\ \mathcal{E}_0(k_1, k_2, \dots, k_r; \tau) &= 2\pi i \int_{\tau}^{i\infty} \mathrm{d}\tau_r \ \frac{\mathrm{G}_{k_r}^0(\tau_r)}{(2\pi i)^{k_r}} \, \mathcal{E}(k_1, k_2, \dots, k_{r-1}; \tau_r) \\ &= -\int_{0}^{q} \mathrm{dlog} \, q_r \ \frac{\mathrm{G}_{k_r}^0(q_r)}{(2\pi i)^{k_r}} \ \mathcal{E}_0(k_1, k_2, \dots, k_{r-1}; q_r) \end{aligned} \tag{2.19}$$

$$\gamma(k_1, k_2, \dots, k_r; \tau) = (2\pi i)^{k_1 + \dots + k_r - 2r} \mathcal{E}(k_1, k_2, \dots, k_r; \tau)$$

$$\gamma_0(k_1, k_2, \dots, k_r; \tau) = (2\pi i)^{k_1 + \dots + k_r - 2r} \mathcal{E}_0(k_1, k_2, \dots, k_r; \tau)$$

Please see appendix D.1 for further details of our conventions.

⁸In ref. [24], a slightly different convention for iterated Eisenstein integrals has been employed. Named γ and γ_0 , they differ from the objects \mathcal{E} and \mathcal{E}_0 defined in eqs. (2.18) and (2.19) by powers of $2\pi i$ and can be related via

$$= (-1)^r \int_{0 \le q_1 \le q_2 \le \dots \le q_r \le q} \operatorname{dlog} q_1 \cdots \operatorname{dlog} q_r \frac{\mathrm{G}^0_{k_1}(q_1)}{(2\pi i)^{k_1}} \cdots \frac{\mathrm{G}^0_{k_r}(q_r)}{(2\pi i)^{k_r}} \,.$$

The recursion starts with $\mathcal{E}(;\tau) = \mathcal{E}_0(;\tau) = 1$, and the non-constant parts of Eisenstein series are defined as

$$G_{2n}^{0}(\tau) := G_{2n}(\tau) - 2\zeta_{2n}, \qquad G_{0}(\tau) := G_{0}^{0}(\tau) := -1$$
(2.20)

with $n \in \mathbb{N}^+$. Our conventions for Eisenstein series G_k are listed in appendix D.1, and we will interchangeably refer to the argument of G_k , G_k^0 and their iterated integrals by τ or q. For both $\mathcal{E}(k_1, k_2, \ldots, k_r; \tau)$ and $\mathcal{E}_0(k_1, k_2, \ldots, k_r; \tau)$ in eqs. (2.18) and (2.19), we will refer to the number of non-zero entries $(k_j \neq 0)$ as the *depth* of the respective iterated Eisenstein integral (similar to the terminology for eMZVs).

Throughout this article, the endpoint divergences of the above integrals as $q_1 \rightarrow 0$ are understood to be shuffle-regularized through the tangential-basepoint prescription described in ref. [59] with the net effect $\int_0^q \frac{dq_1}{q_1} = \log q$. The iterated Eisenstein integrals $\mathcal{E}_0(k_1, \ldots, k_r)$ with $k_1 \neq 0$ do not need to be regularized and have the following Fourier-expansion (cf. eq. (4.62) of ref. [24]):

$$\mathcal{E}_{0}(k_{1}, 0^{p_{1}-1}, k_{2}, 0^{p_{2}-1}, \dots, k_{r}, 0^{p_{r}-1}; q) = (-2)^{r} \left(\prod_{j=1}^{r} \frac{1}{(k_{j}-1)!}\right)$$

$$\times \sum_{m_{i}, n_{i}=1}^{\infty} \frac{m_{1}^{k_{1}-1} m_{2}^{k_{2}-1} \dots m_{r}^{k_{r}-1} q^{m_{1}n_{1}+m_{2}n_{2}+\dots+m_{r}n_{r}}}{(m_{1}n_{1})^{p_{1}} (m_{1}n_{1}+m_{2}n_{2})^{p_{2}} \dots (m_{1}n_{1}+m_{2}n_{2}+\dots+m_{r}n_{r})^{p_{r}}},$$

$$(2.21)$$

where $k_j \neq 0$. The conversion of A-cycle eMZVs to iterated Eisenstein integrals therefore provides an easy way to find their functional dependence on q and, by the linear independence of \mathcal{E} with different labels [60, 29], exposes their relations [24].

The iterated Eisenstein integrals in eq. (2.18) are linear combinations of products of powers of τ and the objects

$$\mathcal{G}\begin{bmatrix}j_1 & j_2 & \dots & j_r\\k_1 & k_2 & \dots & k_r\end{bmatrix} := \int_{\tau}^{i\infty} \mathrm{d}\tau_r \ \tau_r^{j_r} \ \mathrm{G}_{k_r}(\tau_r) \, \mathcal{G}\begin{bmatrix}j_1 & \dots & j_{r-1}\\k_1 & \dots & k_{r-1}\end{bmatrix}, \qquad (2.22)$$

where k_i are even positive integers and j_i are non-negative integers. The results of Brown [59] on the integrals eq. (2.22) will be used to express the modular S-transformations $\mathcal{E}(k_1, k_2, \ldots, k_r; -\frac{1}{\tau})$ in terms of iterated Eisenstein integrals at argument τ , powers of τ and $\mathbb{Q}[(2\pi i)^{\pm 1}]$ -linear combinations of MZVs. For $k_i \neq 0$, one recovers $\mathcal{G}\begin{bmatrix} 0 & 0 & \dots & 0\\ k_1 & k_2 & \dots & k_r \\ k_r & k_r & \vdots \end{bmatrix} = \prod_{j=1}^r (2\pi i)^{k_j-1} \mathcal{E}(k_1, k_2, \ldots, k_r; \tau)$, and the general dictionary between the two types of iterated Eisenstein integrals eqs. (2.18) and (2.22) is described in section 3.3 below. The number r of integrations in eq. (2.22) will be referred to as the *depth* of Brown's iterated Eisenstein integrals, and it is compatible with the notion of depth in their representation via \mathcal{E} in eq. (2.18).

Given a suitable regularization scheme, all objects defined as iterated integrals naturally satisfy shuffle relations. This applies in particular to eMZVs, elliptic iterated integrals and iterated Eisenstein integrals. Shuffle relations can be neatly explained by reorganizing the higherdimensional integration domains and read for the example of iterated Eisenstein integrals:

$$\mathcal{E}(0,0;\tau)\mathcal{E}(4;\tau) = \mathcal{E}(0,0,4;\tau) + \mathcal{E}(0,4,0;\tau) + \mathcal{E}(4,0,0;\tau).$$
(2.23)

2.3 Modular graph functions

The definition of modular graph functions [6] is motivated by the low-energy expansion of the modular invariant integral

$$M_n^{\text{closed}}(s_{ij}|\tau) := \int \mathrm{d}\mu_n^{\text{closed}}(\tau) \exp\left(\sum_{i< j}^n s_{ij} G_{ij}(\tau)\right) \,, \tag{2.24}$$

which appears in one-loop amplitudes of the closed superstring [61, 62] and gives rise to the right-hand side of the correspondence in eq. (1.5). The Green function $G_{ij}(\tau) := G(z_i, z_j; \tau)$ on the torus is defined below, and the integration measure for n external states reads

$$\int d\mu_n^{\text{closed}}(\tau) := \frac{1}{\text{Im}(\tau)^{n-1}} \int_{T(\tau)} d^2 z_2 \int_{T(\tau)} d^2 z_3 \dots \int_{T(\tau)} d^2 z_n$$
(2.25)

with $z_1 = 0$. The z_j are to be integrated over a torus $T(\tau)$ of modular parameter τ , and the above measure is normalized such that $\int_{T(\tau)} d^2 z = \text{Im}(\tau)$. The Green function is only defined up to an additive function of τ , and we will employ the representative

$$G_{ij}(\tau) := -\log \left| \frac{\theta_1(z_{ij}, \tau)}{\eta(\tau)} \right|^2 - \frac{\pi}{2 \operatorname{Im}(\tau)} (z_{ij} - \bar{z}_{ij})^2, \qquad z_{ij} = z_i - z_j$$
(2.26)

which vanishes upon integration over the torus

$$\int_{T(\tau)} d^2 z_i \ G_{ij}(\tau) = 0.$$
 (2.27)

The low-energy expansion of eq. (2.24) can be conveniently represented graphically. After expanding the exponential in the integrand as a power series and exchanging integration and summation, one can associate a graph to every summand in the following way: each integration variable of eq. (2.25) is represented by a vertex, and each Green function G_{ij} between vertices i and j is visualized by an edge [1, 2]

$$G_{ij}(\tau) = {}_{i} \bullet - \bullet_{j} \,. \tag{2.28}$$

Then property eq. (2.27) implies the vanishing of one-particle reducible graphs⁹, so the simplest contributions to the low-energy expansion of eq. (2.24) stem from two-vertex graphs with multiple edges. The associated modular graph functions are given by

$$\mathbf{D}[\boldsymbol{\curvearrowleft}] := \int \mathrm{d}\mu_2^{\text{closed}} \ G_{12}^2, \qquad \mathbf{D}[\boldsymbol{\ominus}] := \int \mathrm{d}\mu_2^{\text{closed}} \ G_{12}^3, \qquad \mathbf{D}[\boldsymbol{\ominus}] := \int \mathrm{d}\mu_2^{\text{closed}} \ G_{12}^4, \quad (2.29)$$

and we will employ a graphical labeling for their generalizations to one-particle irreducible graphs with multiple vertices, e.g.

$$\mathbf{D}\left[\bigtriangleup\right] := \int d\mu_3^{\text{closed}} G_{12} G_{13} G_{23} , \qquad \mathbf{D}\left[\bigtriangleup\right] := \int d\mu_3^{\text{closed}} G_{12}^2 G_{13} G_{23} , \qquad (2.30)$$
$$\mathbf{D}\left[\bigsqcup\right] := \int d\mu_4^{\text{closed}} G_{12} G_{23} G_{34} G_{41} , \qquad \mathbf{D}\left[\boxdot\right] := \int d\mu_5^{\text{closed}} G_{13} G_{34} G_{42} G_{15} G_{52} G_{12} .$$

We suppress the dependence on τ in eqs. (2.29) and (2.30) as well as in later equations.

⁹One-particle reducible graphs are those which can be disconnected by removing an edge.

The number of edges in the graphical representation equals the *weight* of a modular graph function. A translation between graphs at higher weight and their names in refs. [2, 63] is provided in table 1 in appendix A. In terms of modular graph functions, the α' -expansion of the four-point integral eq. (2.24) reads

$$M_{4}^{\text{closed}}(s_{ij}|\tau) = 1 + 2\mathbf{D}\left[\boldsymbol{\bigcirc}\right](s_{12}^{2} + s_{12}s_{23} + s_{23}^{2}) + (\mathbf{D}\left[\boldsymbol{\bigcirc}\right] + 4\mathbf{D}\left[\boldsymbol{\bigtriangleup}\right])s_{12}s_{23}s_{13} + \frac{1}{6}\left(\mathbf{D}\left[\boldsymbol{\ominus}\right] + 9\mathbf{D}\left[\boldsymbol{\bigcirc}\right]^{2} + 6\mathbf{D}\left[\boldsymbol{\Box}\right]\right)(s_{12}^{2} + s_{12}s_{23} + s_{23}^{2})^{2}$$
(2.31)
$$+ \frac{1}{12}\left(\mathbf{D}\left[\boldsymbol{\ominus}\right] + 48\mathbf{D}\left[\boldsymbol{\bigtriangleup}\right]\mathbf{D}\left[\boldsymbol{\ominus}\right] + 12\mathbf{D}\left[\boldsymbol{\Box}\right] - 12\mathbf{D}\left[\boldsymbol{\bigtriangleup}\right] + 16\mathbf{D}\left[\boldsymbol{\bigtriangleup}\right] + 14\mathbf{D}\left[\boldsymbol{\ominus}\right]\mathbf{D}\left[\boldsymbol{\ominus}\right] - 24\mathbf{D}\left[\boldsymbol{\ominus}\right]\right) \times s_{12}s_{23}s_{13}(s_{12}^{2} + s_{12}s_{23} + s_{23}^{2}) + \mathcal{O}(\alpha'^{6}),$$

where we have used the relations $s_{12} = s_{34}$, $s_{14} = s_{23}$ and $s_{13} = s_{24} = -s_{12} - s_{23}$ among four-particle Mandelstam variables. Since $M_4^{\text{closed}}(s_{ij}|\tau)$ is the only integral contributing to the four-point amplitude, the one-loop contribution to $D^{2w}R^4$ operators in the effective action follows from integrating eq. (2.31) over the fundamental domain with respect to τ [1–3]. Closedstring one-loop amplitudes for $n \geq 5$ points, however, involve a variety of additional integrals besides $M_n^{\text{closed}}(s_{ij}|\tau)$ [62–64, 10].

The complexity of modular graph functions is correlated with the number of loops in its graphical representation. We will later on define a notion of depth for modular graph functions which relates to the depth of iterated Eisenstein integrals and which is conjecturally bounded from above by the loop order of the graph. One-loop graphs give rise to the simplest class of modular graph functions: These are non-holomorphic Eisenstein series,

$$\mathbf{D}[\boldsymbol{\frown}] = \mathbf{E}_2, \quad \mathbf{D}[\boldsymbol{\triangle}] = \mathbf{E}_3, \quad \mathbf{D}[\boldsymbol{\Box}] = \mathbf{E}_4, \quad \mathbf{D}[\boldsymbol{\frown}] = \mathbf{E}_5, \dots \quad (2.32)$$

which are defined by the lattice sums

$$E_{k}(\tau) = \left(\frac{\operatorname{Im}(\tau)}{\pi}\right)^{k} \sum_{(m,n)\neq(0,0)} \frac{1}{|m+\tau n|^{2k}}$$

$$= e_{k}(y) - 8y(2k-1)! \sum_{j=0}^{k-1} \binom{2k-2-j}{k-1} \frac{1}{j!} (4y)^{j-k} \operatorname{Re}[\mathcal{E}_{0}(2k, \underbrace{0, \dots, 0}_{2k-2-j}; q)]$$
(2.33)

with $y := \pi \operatorname{Im}(\tau)$, Bernoulli numbers B_{2k} and

$$\mathbf{e}_{k}(y) = (-1)^{k-1} \frac{B_{2k}}{(2k)!} (4y)^{k} + \frac{4(2k-3)!}{(k-2)! (k-1)!} \zeta_{2k-1} (4y)^{1-k} .$$
(2.34)

For generic modular graph functions, a lattice-sum representation generalizing the first line of eq. (2.33) can be straightforwardly deduced from the Fourier-expansion of the Green function eq. (2.26) with respect to $\frac{\text{Im} z}{\text{Im} \tau}$ [1],

$$G_{ij}(\tau) = \frac{\mathrm{Im}\,\tau}{\pi} \sum_{(m,n)\neq(0,0)} \frac{\mathrm{e}^{2\pi i (n\alpha_{ij} - m\beta_{ij})}}{|m + \tau n|^2} , \qquad z_{ij} = \alpha_{ij} + \tau\beta_{ij} , \qquad \alpha_{ij}, \beta_{ij} \in \mathbb{R} .$$
(2.35)

However, the q-expansions of modular graph functions beyond E_k have not been spelt out in

the literature before, and we will propose new results in terms of iterated Eisenstein integrals \mathcal{E}_0 with q-expansion eq. (2.21) in section 4.2.

Laurent polynomials in the zero modes 2.3.1

Modular graph functions associated with a one-particle irreducible graph \mathcal{G} admit a double expansion of the form

$$\mathbf{D}[\mathcal{G}] = \sum_{m,n=0}^{\infty} c_{m,n}^{\mathcal{G}}(y) q^m \bar{q}^n, \qquad (2.36)$$

where the coefficients $c_{m,n}^{\mathcal{G}}(y)$ are Laurent polynomials in $y = \pi \operatorname{Im}(\tau)$ of maximum degree equal to the number of edges (or weight) w of \mathcal{G} and minimum degree 1 - w [65]. A variety of results is available on the polynomial $c_{0,0}^{\mathcal{G}}(y) =: \mathbf{d}[\mathcal{G}]$ which describes the behavior of the corresponding modular graph function at the cusp $\tau \to i\infty$. In abuse of nomenclature, the polynomial $\mathbf{d}[\mathcal{G}]$ will be referred to as the zero mode. Apart from the zero modes $e_k(y)$ for the polygonal graphs in eq. (2.34), the results to be compared with an open-string setup below read [2,11]

$$\mathbf{d}\left[\bigtriangleup\right] = \frac{2y^4}{14175} + \frac{y\zeta_3}{45} + \frac{5\zeta_5}{12y} - \frac{\zeta_3^2}{4y^2} + \frac{9\zeta_7}{16y^3} \tag{2.37}$$

$$\mathbf{d}\left[\prod\right] = \frac{2y^5}{155925} + \frac{2y^2\zeta_3}{945} - \frac{\zeta_5}{180} + \frac{7\zeta_7}{16y^2} - \frac{\zeta_3\zeta_5}{2y^3} + \frac{43\zeta_9}{64y^4}$$
(2.38)

at weight four and five as well as

$$\mathbf{d}\left[\bigoplus\right] = \frac{38y^6}{91216125} + \frac{\zeta_7}{24y} - \frac{7\zeta_9}{16y^3} + \frac{15\zeta_5^2}{16y^4} - \frac{81\zeta_{11}}{128y^5} \tag{2.39}$$

$$\mathbf{d}\left[\mathbf{\textcircled{G}}\right] = \frac{808y^6}{638512875} + \frac{y^3\zeta_3}{4725} - \frac{y\zeta_5}{1890} + \frac{\zeta_7}{720y} + \frac{23\zeta_9}{64y^3} - \frac{\zeta_5^2 + 30\zeta_3\zeta_7}{64y^4} + \frac{167\zeta_{11}}{256y^5}$$
(2.40)

$$\mathbf{d}\left[\bigoplus\right] = \frac{43y^6}{58046625} + \frac{y\zeta_5}{630} + \frac{\zeta_7}{144y} + \frac{7\zeta_9}{64y^3} - \frac{17\zeta_5^2}{64y^4} + \frac{99\zeta_{11}}{256y^5}$$
(2.41)

$$\mathbf{d}\left[\textcircled{\textcircled{\textcircled{G}}}\right] = \frac{103y^6}{13030875} + \frac{y^3\zeta_3}{2025} + \frac{y\zeta_5}{54} - \frac{\zeta_3^2}{90} - \frac{\zeta_7}{360y} + \frac{5\zeta_3\zeta_5}{12y^2} + \frac{5\zeta_9 - 48\zeta_3^3}{288y^3} + \frac{14\zeta_3\zeta_7 + 25\zeta_5^2}{32y^4} - \frac{73\zeta_{11}}{128y^5} \tag{2.42}$$

at weight six. While the above examples exclusively involve zeta-values of $depth^{10}$ one, some of the modular graph functions at weight $w \ge 7$ were shown to involve single-valued MZVs at depth three, for instance¹¹ [65]

$$\mathbf{d}\left[\textcircled{\textcircled{6}}\right] = \frac{62y^7}{10945935} + \frac{2y^4\zeta_3}{243} + \frac{119y^2\zeta_5}{324} + \frac{11y\zeta_3^2}{27} + \frac{21\zeta_7}{16} + \frac{46\zeta_3\zeta_5}{3y} + \frac{7115\zeta_9}{288y^2} - \frac{25\zeta_3^3}{2y^2} - \frac{75\zeta_5^2}{8y^3} \quad (2.43)$$
$$+ \frac{1245\zeta_3\zeta_7}{16y^3} - \frac{9(\zeta_{3,5,3} - \zeta_3\zeta_{3,5})}{4y^4} - \frac{315\zeta_3^2\zeta_5}{8y^4} - \frac{9573\zeta_{11}}{128y^4} + \frac{2475\zeta_5\zeta_7}{32y^5} + \frac{1125\zeta_3\zeta_9}{32y^5} - \frac{1575\zeta_{13}}{32y^6}$$

can be rewritten as

$$\mathbf{d}\left[\textcircled{\textcircled{\mathbf{b}}}\right] = \frac{62y^7}{10945935} + \frac{y^4\zeta_3^{\mathrm{sv}}}{243} + \frac{119y^2\zeta_5^{\mathrm{sv}}}{648} + \frac{11y(\zeta_3^{\mathrm{sv}})^2}{108} + \frac{21\zeta_7^{\mathrm{sv}}}{32} + \frac{23\zeta_3^{\mathrm{sv}}\zeta_5^{\mathrm{sv}}}{6y} + \frac{7115\zeta_9^{\mathrm{sv}}}{576y^2} - \frac{25(\zeta_3^{\mathrm{sv}})^3}{16y^2} + \frac{11y(\zeta_3^{\mathrm{sv}})^2}{10y^2} + \frac{11y($$

¹⁰The depth r of MZVs $\zeta_{n_1,...,n_r}$ is not a grading, thus it is often possible that the same MZV has two different representations where the depth changes; for instance $\zeta_3 = \zeta_{1,2}$. Here, when we say that MZVs have a certain depth, we mean that they cannot be written as polynomials in MZVs of lower depth. ¹¹There is a typo in the coefficient of y^{-4} in the corresponding formula in ref. [65].

$$-\frac{75(\zeta_5^{\rm sv})^2}{32y^3} + \frac{1245\zeta_3^{\rm sv}\zeta_7^{\rm sv}}{64y^3} - \frac{9\zeta_{3,5,3}^{\rm sv}}{8y^4} - \frac{405(\zeta_3^{\rm sv})^2\zeta_5^{\rm sv}}{64y^4} - \frac{9573\zeta_{11}^{\rm sv}}{256y^4} + \frac{2475\zeta_5^{\rm sv}\zeta_7^{\rm sv}}{128y^5} + \frac{1125\zeta_3^{\rm sv}\zeta_9^{\rm sv}}{128y^5} - \frac{1575\zeta_{13}^{\rm sv}}{64y^6} . \tag{2.44}$$

It is conjectured that the coefficients of all Laurent polynomials in eq. (2.36) can be written in terms of single-valued MZVs [65]. Finally, the zero modes in modular graph functions associated with two-point or two-loop graphs are known in closed form [66, 13].

2.3.2 Relations among modular graph functions

Modular graph functions corresponding to different graphs are not independent objects: they satisfy various relations involving (conjecturally only) single-valued MZVs, starting with the relation proved by Don Zagier [67] (see also [3])

$$0 = \mathbf{D}\left[\boldsymbol{\leftrightarrow}\right] - \mathbf{D}\left[\boldsymbol{\bigtriangleup}\right] - \zeta_3. \tag{2.45}$$

At weight four and five, the techniques of [3, 4, 7] led to

$$\mathbf{D}\left[\textcircled{\begin{subarray}{c}}\right] = 24 \, \mathbf{D}\left[\textcircled{\begin{subarray}{c}}\right] - 18 \, \mathbf{D}\left[\textcircled{\begin{subarray}{c}}\right] + 3 \, \mathbf{D}\left[\textcircled{\begin{subarray}{c}}\right]^2 \tag{2.46}$$

$$40 \mathbf{D}\left[\bigtriangleup\right] = 300 \mathbf{D}\left[\Box\right] + 120 \mathbf{D}\left[\circlearrowright\right] \mathbf{D}\left[\bigtriangleup\right] - 276 \mathbf{D}\left[\circlearrowright\right] + 7\zeta_5 \qquad (2.47)$$

$$\mathbf{D}\left[\textcircled{\textcircled{}}\right] = 60 \,\mathbf{D}\left[\fbox{]}\right] + 10 \,\mathbf{D}\left[\textcircled{\textcircled{}}\right] \mathbf{D}\left[\textcircled{\textcircled{}}\right] - 48 \,\mathbf{D}\left[\fbox{]}\right] + 16\zeta_5 \tag{2.48}$$

$$10 \mathbf{D}[\Delta] = 20 \mathbf{D}[\Box] - 4 \mathbf{D}[\Delta] + 3\zeta_5$$
(2.49)

$$30 \mathbf{D} \Big[\bigoplus \Big] = 12 \mathbf{D} \Big[\bigcap \Big] + \zeta_5 , \qquad (2.50)$$

and the complete set of weight-six relations displayed in appendix F has been identified in ref. [11].

2.3.3 Laplace equations among modular graph functions

Various combinations and powers of modular graph functions are related through a web of eigenvalue equations for the Laplacian $\Delta := 4(\operatorname{Im} \tau)^2 \frac{\partial^2}{\partial \tau \partial \bar{\tau}}$. While the non-holomorphic Eisenstein series eq. (2.33) at one loop satisfy

$$(\Delta - k(k-1)) E_k = 0, \qquad (2.51)$$

the systematics of inhomogeneous Laplace eigenvalue equations at two loops has been described in ref. [3], leading for instance to

$$(\Delta - 2) \mathbf{D} \Big[\bigtriangleup \Big] = 9 \operatorname{E}_4 - \operatorname{E}_2^2 \tag{2.52}$$

$$(\Delta - 6) \mathbf{D} \left[\prod \right] = \frac{86}{5} \operatorname{E}_5 - 4 \operatorname{E}_2 \operatorname{E}_3 + \frac{\zeta_5}{10}$$
(2.53)

as well as

$$(\Delta - 2)(4\mathbf{D}[\textcircled{c}] + \mathbf{D}[\textcircled{c}]) = 52 \operatorname{E}_{6} - 4 \operatorname{E}_{3}^{2}$$
$$(\Delta - 12)(6\mathbf{D}[\textcircled{c}] - \mathbf{D}[\textcircled{c}]) = 108 \operatorname{E}_{6} - 36 \operatorname{E}_{3}^{2}$$
(2.54)

$$(\Delta - 12)(6 \mathbf{D}[\textcircled{\leftrightarrow}] + \mathbf{D}[\textcircled{\leftrightarrow}]) = 120 \operatorname{E}_6 + 12 \operatorname{E}_3^2 - 36 \operatorname{E}_2 \operatorname{E}_4.$$

Laplace equations for the tetrahedral topology \triangle at three loops¹² are known from ref. [12]; we will report on a new weight-six identity involving less symmetric topologies in section 4.2.5.

2.3.4 Cauchy–Riemann equations among modular graph functions

An essential tool in deriving relations between modular graph functions is the Cauchy–Riemann derivative

$$\nabla := 2i(\operatorname{Im}\tau)^2 \partial_\tau \tag{2.55}$$

with $\partial_{\tau} := \frac{\partial}{\partial \tau}$, which maps modular forms of weight (0, w) to those of weight (0, w-2). For instance, repeated application of the Cauchy–Riemann derivative (2.55) mediates between non-holomorphic and holomorphic Eisenstein series [7]

$$\Gamma(k)(\pi\nabla)^k \mathbf{E}_k = \Gamma(2k)(\mathrm{Im}(\tau))^{2k} \mathbf{G}_{2k} .$$
(2.56)

At higher loop-order, the Cauchy–Riemann equations

$$(\pi\nabla)^{3} \mathbf{D} \Big[\bigtriangleup \Big] = \frac{9}{10} (\pi\nabla)^{3} \mathbf{E}_{4} - 6 \operatorname{Im}(\tau)^{4} \mathbf{G}_{4} \, \pi\nabla \mathbf{E}_{2}$$
(2.57)

$$(\pi\nabla)^{3} \mathbf{D} \left[\prod \right] = \frac{43}{35} (\pi\nabla)^{3} \mathbf{E}_{5} - 2(\pi\nabla \mathbf{E}_{2})(\pi\nabla)^{2} \mathbf{E}_{3} - 4\operatorname{Im}(\tau)^{4} \mathbf{G}_{4} \,\pi\nabla \mathbf{E}_{3}$$
(2.58)

have been instrumental to prove the weight-four and weight-five relations in eqs. (2.46) to (2.50) [7]. The same method has been applied in [11] to derive the weight-six relations in eq. (F.1) as well as selected relations at weight seven.

Holomorphic Eisenstein series appear in both the Cauchy–Riemann derivatives of modular graph functions and the τ -derivative eq. (C.3) of eMZVs. In subsection 4.2.1 below, we will report on a correspondence between eqs. (2.56) to (2.58) and differential equations of associated combinations of eMZVs.

3 An open-string setup for graph functions

In this section, we will describe an open-string setup mimicking the graphical organization of the closed-string α' -expansion in subsection 2.3. Choosing auxiliary abelian open-string states, the permutation symmetry of the closed-string integration measure in eq. (2.25) can be implemented in an open-string setup. As a consequence, external abelian states allow to rewrite the low-energy expansion of open-string integrals without one-particle irreducible graphs. Having done so, the structure of the closed-string integral eq. (2.31) equals that of the four-point amplitude among abelian open-string states.

The open-string analogues of the modular graph functions will be referred to as "A-cycle graph functions" and expressed in terms of the A-cycle eMZVs introduced in subsection 2.2. Accordingly, the results of their modular S-transformation will be referred to as "B-cycle graph functions", and we will introduce techniques to express them in terms of the same iterated Eisenstein integrals as employed for A-cycle graph functions. These expressions for B-cycle graph functions will be the starting point for proposing an analogue of the single-valued projection

 $^{^{12}\}mathrm{See}$ [5,8] for earlier work on Laplace equations of specific three-loop examples.

from subsection 2.1 in the one-loop setup and furnish the left hand side of the correspondence in eq. (1.5).

3.1 Definition of A- and B-cycle graph functions

3.1.1 Review of open-string α' -expansions

The color-ordered one-loop amplitude of four non-abelian open-string states reads¹³

$$I_{4\text{pt}}(1,2,3,4|\tau) := \int_{0 \le z_2 \le z_3 \le z_4 \le 1} \mathrm{d}z_2 \, \mathrm{d}z_3 \, \mathrm{d}z_4 \, \exp\left(\frac{1}{2} \sum_{i< j}^4 s_{ij} G_{ij}(\tau)\right) \,, \tag{3.1}$$

with $z_1 = 0$. The integration domain corresponds to a single-trace contribution of the nonabelian gauge-group generators¹⁴. The open-string Green function G_{ij} can be obtained from the closed-string version in eq. (2.26) by restricting to real arguments. Comparing with the definition of elliptic iterated integrals in eq. (2.14) and the form of the integration kernel $f^{(1)}$, we find

$$G_{ij}(\tau) = -2\Gamma\left(\frac{1}{0}; z_{ij}|\tau\right) + k(\tau) = -2\Gamma\left(\frac{1}{z_j}; z_i|\tau\right) - 2\Gamma\left(\frac{1}{0}; z_j|\tau\right) + k(\tau).$$
(3.2)

The iterated elliptic integrals Γ in eq. (3.2) need regularization, see e.g. section 4.2.1 of ref. [25], which leads to the scheme-dependent quantity $k(\tau)$. The latter, however, does not depend on z_i, z_j and thus cancels out from eq. (3.1) after using momentum conservation $\sum_{i < j} s_{ij} = 0$. We will suppress the dependence on τ henceforth.

The representation eq. (3.2) of the Green function has been used to algorithmically perform the α' -expansion of eq. (3.1) in the framework of eMZVs, leading to [23]

$$I_{4\text{pt}}(1,2,3,4) = \frac{1}{6} - 2s_{13}\,\omega_A(0,1,0,0) + 2\,\omega_A(0,1,1,0,0)\left(s_{12}^2 + s_{23}^2\right) - 2\,\omega_A(0,1,0,1,0)\,s_{12}s_{23} \\ - \beta_5\left(s_{12}^3 + 2s_{12}^2s_{23} + 2s_{12}s_{23}^2 + s_{23}^3\right) - \beta_{2,3}\,s_{12}s_{23}(s_{12} + s_{23}) + \mathcal{O}(\alpha'^4) \quad (3.3)$$

with

$$\beta_5 = \frac{4}{3} \left[\omega_A(0,0,1,0,0,2) + \omega_A(0,1,1,0,1,0) - \omega_A(2,0,1,0,0,0) - \zeta_2 \,\omega_A(0,1,0,0) \right] \quad (3.4)$$

$$\beta_{2,3} = \frac{\zeta_3}{12} + \frac{8\,\zeta_2}{3}\,\omega_A(0,1,0,0) - \frac{5}{18}\,\omega_A(0,3,0,0)\,. \tag{3.5}$$

As can be seen from the non-vanishing contribution at linear order, a single Green function does not integrate to zero. This is true in general for the non-abelian situation: one cannot find a constant $c(\tau)$ such that both $G_{12}(\tau) + c(\tau)$ and the cyclically inequivalent $G_{13}(\tau) + c(\tau)$ integrate to zero simultaneously within eq. (3.1). Hence, in presence of non-abelian open-string states, there is no analogue of the property eq. (2.27) which eliminates one-particle reducible graphs in the expansion.

¹³Given that the normalization of α' is tailored to the closed-string setup in this work, the expressions for $I_{4pt}(1, 2, 3, 4)$ given in [23] is recovered from eq. (3.1) by rescaling $\alpha' \to 4\alpha'$. The definitions eqs. (2.3) and (2.26) of the Mandelstam invariants and the Green function on the torus are identical to those of [3,6,7,11] to match the conventions of the references for closed-string integrals and modular graph functions. The normalization of s_{ij} and G_{ij} chosen in [23,25] can be obtained from eqs. (2.3) and (2.26) by rescaling $s_{ij} \to -4s_{ij}$ and $G_{ij} \to -G_{ij}$, respectively.

¹⁴The contributions from cylinder- and Möbius-strip diagrams to planar one-loop amplitudes are obtained by integrating (3.1) over $\tau \in i\mathbb{R}_+$ and $\tau \in \frac{1}{2} + i\mathbb{R}_+$, respectively [26].

3.1.2 Open-string α' -expansion for abelian states

Switching from non-abelian to abelian open-string states amounts to democratically combining all different possible integration domains in eq. (3.1) and to independently integrating each z_j for j = 2, 3, 4 over the unit interval. Hence, we will be interested in symmetrized open-string integrals

$$M_n^{\text{open}}(s_{ij}) := \int d\mu_n^{\text{open}} \exp\left(\sum_{i< j}^n s_{ij} P_{ij}\right)$$
(3.6)

with $z_1 = 0$ and an integration measure analogous to eq. (2.25):

$$\int d\mu_n^{\text{open}} := \int_0^1 dz_2 \int_0^1 dz_3 \dots \int_0^1 dz_n \,. \tag{3.7}$$

Momentum conservation has been used to trade the Green function eq. (3.2) for¹⁵

$$P_{1j} := \omega_A(1,0) - \Gamma\left(\frac{1}{0}; z_j\right), \qquad P_{ij} := \omega_A(1,0) - \Gamma\left(\frac{1}{z_j}; z_i\right) - \Gamma\left(\frac{1}{0}; z_j\right), \tag{3.8}$$

with $z_1 = 0$ and $i, j \neq 1$ (we have suppressed the dependence on τ from the notation). Note that one can also swap the roles of z_i and z_j in the rightmost expression since $P_{ij} = P_{ji}$. In analogy to the situation for the quantity $k(\tau)$ in eq. (3.2), the addition of $\omega_A(1,0)$ in eq. (3.8) does not contribute to the open-string integral eq. (3.1) after taking momentum conservation into account. However, including $\omega_A(1,0)$ into the propagator eq. (3.8) ensures that an analogue of the crucial identity eq. (2.27) from the closed-string setup holds

$$\int_0^1 \mathrm{d}z_i \ P_{ij} = 0 \,, \tag{3.9}$$

as can be checked using the definition eq. (2.14) of elliptic iterated integrals. Then, the α' expansion of the four-point amplitude among abelian open-string states will be organized in
terms of one-particle irreducible graphs: each integration variable in eq. (3.7) is represented
by a vertex, and each propagator P_{ij} in eq. (3.8) between vertices *i* and *j* is visualized by an
undirected edge, as shown below.

$$P_{ij} = {}_{i} \bullet \underbrace{}_{j} \cdot \underbrace{}_{(3.10)}$$

In these conventions, the open-string analogue of eq. (2.31) reads

$$M_{4}^{\text{open}}(s_{ij}) = 1 + 2\mathbf{A}\left[\boldsymbol{\bigcirc}\right](s_{12}^{2} + s_{12}s_{23} + s_{23}^{2}) + (\mathbf{A}\left[\boldsymbol{\bigcirc}\right] + 4\mathbf{A}\left[\boldsymbol{\bigtriangleup}\right])s_{12}s_{23}s_{13} + \frac{1}{6}\left(\mathbf{A}\left[\boldsymbol{\bigcirc}\right] + 9\mathbf{A}\left[\boldsymbol{\bigcirc}\right]^{2} + 6\mathbf{A}\left[\boldsymbol{\Box}\right]\right)(s_{12}^{2} + s_{12}s_{23} + s_{23}^{2})^{2} + \frac{1}{12}\left(\mathbf{A}\left[\boldsymbol{\bigcirc}\right] + 48\mathbf{A}\left[\boldsymbol{\bigtriangleup}\right]\mathbf{A}\left[\boldsymbol{\bigcirc}\right] + 12\mathbf{A}\left[\boldsymbol{\Box}\right] - 12\mathbf{A}\left[\boldsymbol{\bigtriangleup}\right] + 16\mathbf{A}\left[\boldsymbol{\bigtriangleup}\right] + 14\mathbf{A}\left[\boldsymbol{\bigcirc}\right]\mathbf{A}\left[\boldsymbol{\bigcirc}\right] - 24\mathbf{A}\left[\boldsymbol{\bigcirc}\right]\right) \times s_{12}s_{23}s_{13}(s_{12}^{2} + s_{12}s_{23} + s_{23}^{2}) + \mathcal{O}(\alpha'^{6}),$$

$$(3.11)$$

¹⁵Note that the right hand side of eq. (3.8) does not match the definition of P_{ij} in ref. [25].

where the A-cycle graph function $\mathbf{A}[\mathcal{G}]$ associated with a graph \mathcal{G} is defined in analogy with the corresponding modular graph function $\mathbf{D}[\mathcal{G}]$

$$\mathbf{A}[\mathcal{G}] := \mathbf{D}[\mathcal{G}] \Big|_{G_{ij} \to P_{ij}}^{\mathrm{d}\mu_n^{\mathrm{closed}} \to \mathrm{d}\mu_n^{\mathrm{open}}}, \qquad (3.12)$$

for instance

$$\mathbf{A}\left[\longleftrightarrow\right] := \int \mathrm{d}\mu_2^{\text{open}} P_{12}^2, \qquad \mathbf{A}\left[\bigoplus\right] := \int \mathrm{d}\mu_5^{\text{open}} P_{13} P_{34} P_{42} P_{15} P_{52} P_{12}.$$

Again, the number of edges in the graphical representation equals the *weight* of an A-cycle graph function.

Finally, symmetrizing over the respective integration domains, the four-point integral in the abelian case coincides with the symmetrization of eq. (3.3),

$$M_4^{\text{open}}(s_{ij}) = \sum_{\rho \in S_3} I_{4\text{pt}}(1, \rho(2, 3, 4)).$$
(3.13)

In particular, up to the orders where $I_{4pt}(1, 2, 3, 4)$ is available, eq. (3.13) has been used as a consistency check for the explicit results for the A-cycle graph functions in eq. (3.11) to be obtained in the next section.

Although the *n*-point amplitude of the open superstring involves many integrals beyond eq. (3.6) [68,69,23,64,70], we still want to study A-cycle graph function with $n \ge 5$ vertices for the sake of their parallels with modular graph functions.

3.1.3 B-cycle graph functions

The open-string integral eq. (3.1) and the measure eq. (3.7) are expressed in a parametrization of the cylinder worldsheet, where one of the boundary components is the A-cycle. By a modular transformation, this setup is related to a parametrization of the boundary component through the path from 0 to τ , i.e. the B-cycle (cf. figure 2 in section 2.2). In order to compare openstring quantities with modular graph functions below, we will study the image of A-cycle graph functions under the S-transformation $\tau \to -\frac{1}{\tau}$ (cf. below eq. (2.9)),

$$\mathbf{B}[\mathcal{G}] := \mathbf{A}[\mathcal{G}] \Big|_{\tau \to -\frac{1}{2}}, \qquad (3.14)$$

which will be referred to as *B*-cycle graph functions, and can be expressed in terms of *B*-cycle eMZVs by eq. (2.11). Techniques for their systematic evaluation in terms of *A*-cycle quantities $\mathcal{E}_0(\ldots;\tau)$ with known *q*-expansion eq. (2.21) will be discussed in section 3.3. The main motivation to do this comes from the fact that the asymptotic expansion at the cusp of *B*-cycle eMZVs (2.12) looks more suitable to be compared with the asymptotic expansion of modular graph functions (2.36) than the simple Fourier expansion of their *A*-cycle counterparts.

3.2 Evaluating A-cycle graph functions

The representation of the propagator in eq. (3.8) guarantees that the low-energy expansion of open-string integrals eq. (3.6) is expressible in terms of elliptic iterated integrals. As will be argued below, there is no bottleneck in algorithmically computing A-cycle graph functions of arbitrary complexity by means of the techniques developed in refs. [23,24].

3.2.1 A-cycle graph functions at weight two

The simplest non-trivial A-cycle graph function at second order of eq. (3.11) can be computed using the definition eq. (2.14) of elliptic iterated integrals,

$$\mathbf{A}\left[\frown\right] = \int_{0}^{1} dz_{2} \left\{ \omega_{A}(1,0)^{2} - 2\omega_{A}(1,0)\Gamma\left(\frac{1}{0};z_{2}\right) + \Gamma\left(\frac{1}{0};z_{2}\right)^{2} \right\}$$
(3.15)
$$= 2\omega_{A}(1,1,0) - 2\omega_{A}(1,0)^{2} + \omega_{A}(1,0)^{2} = \omega_{A}(2,0,0) + \frac{5\zeta_{2}}{6}.$$

Here and below we have been using relations between eMZVs like $2 \omega_A(1, 1, 0) = \frac{5\zeta_2}{6} + \omega_A(1, 0)^2 + \omega_A(2, 0, 0)$, which can be found on the website [71] along with various generalizations up to and including length six. In eq. (3.15) as well as in all computations of A-cycle graph functions below, the term $\omega_A(1, 0)$ in the propagator eq. (3.8) avoids the appearance of divergent eMZVs.

3.2.2 A-cycle graph functions at weight three

The A-cycle graph functions at the third order of eq. (3.11) can be computed via

$$\begin{aligned} \mathbf{A} \Big[\bigoplus \Big] &= \int d\mu_2^{\text{open}} P_{12}^3 \\ &= \int_0^1 dz_2 \left\{ \omega_A(1,0)^3 - 3\,\omega_A(1,0)^2 \Gamma\left(\frac{1}{0};z_2\right) + 3\,\omega_A(1,0) \Gamma\left(\frac{1}{0};z_2\right)^2 - \Gamma\left(\frac{1}{0};z_2\right)^3 \right\} \\ &= -6\,\omega_A(1,1,1,0) + 6\,\omega_A(1,1,0)\,\omega_A(1,0) - 2\,\omega_A(1,0)^3 \\ &= \frac{\zeta_3}{2} + 8\zeta_2\,\omega_A(0,1,0,0) - \frac{1}{3}\,\omega_A(0,3,0,0) \end{aligned}$$
(3.16)
$$\mathbf{A} \Big[\bigtriangleup \Big] &= \int d\mu_3^{\text{open}} P_{12}P_{13}P_{23} \\ &= \omega_A(1,0)^3 - 2\int_0^1 dz_3\int_0^{z_3} dz_2 \,\Gamma\left(\frac{1}{0};z_2\right) \Gamma\left(\frac{1}{0};z_3\right) \left\{ \Gamma\left(\frac{1}{0};z_3\right) + \Gamma\left(\frac{1}{z_3};z_2\right) \right\} \end{aligned}$$

$$= \omega_A(1,0)^3 + 2 \int_0^{\cdot} dz_3 \Gamma\left(\begin{smallmatrix} 1 \\ 0 \\ 0 \end{bmatrix}; z_3) \Gamma\left(\begin{smallmatrix} 1 \\ 2 \\ 3 \\ 0 \\ 0 \end{bmatrix}; z_3\right)$$

= $2\zeta_2 \,\omega_A(0,1,0,0) - \frac{1}{3} \,\omega_A(0,3,0,0) \,.$ (3.17)

In eq. (3.16), the relevant eMZV relation is

$$\omega_A(0,1,1,1) = \frac{\zeta_3}{12} - \frac{\zeta_2}{4} \,\omega_A(0,1) + \frac{1}{6} \,\omega_A(0,1)^3 + \frac{1}{36} \,\omega_A(0,3)$$

$$+ \frac{1}{2} \,\omega_A(0,1) \,\omega_A(0,0,2) + 4\zeta_2 \,\omega_A(0,0,0,1) - \frac{1}{6} \,\omega_A(0,0,0,3) \,,$$
(3.18)

and the last step of eq. (3.17) involves the identity (2.17) for $\Gamma\left(\begin{smallmatrix}1&0&1\\z_3&0&0\end{smallmatrix};z_3\right)$ along with the eMZV relations from appendix I.2 of ref. [25]. Moreover, in eqs. (3.16) and (3.17) we have replaced the integration domains according to $\int_0^1 dz_2 \int_0^1 dz_3 \rightarrow 2 \int_0^1 dz_3 \int_0^{z_3} dz_2$, which is valid along with any monomial $P_{12}^m P_{13}^n P_{23}^q$ due to the symmetry $P_{ij} = P_{ji}$ of the propagator, i.e. for any three-vertex diagram.

3.2.3 Computing A-cycle graph functions at higher weight

A-cycle graph functions with higher numbers of vertices n can be algorithmically computed by iterating the manipulations in eq. (3.17). Among other things, the recursive techniques of [23]

to eliminate the appearance of the argument z in the second line of the elliptic iterated integral $\Gamma\left(\begin{smallmatrix}n_1 & n_2 & \dots & n_r \\ z & 0 & \dots & 0\end{smallmatrix}\right)$; z) – see e.g. eq. (2.17) – play a key role. As will be explained in the following, A-cycle graph functions with an arbitrary number of vertices or edges can always be expressed in terms of eMZVs.

In order to connect with the definition (2.14) of elliptic iterated integrals, the integration region $[0,1]^{n-1}$ of the measure eq. (3.7) has to be decomposed into simplicial cells defined by $0 \leq z_{\rho(2)} \leq z_{\rho(3)} \leq \ldots \leq z_{\rho(n)} \leq 1$ with $\rho \in S_{n-1}$. Using the symmetry $P_{ij} = P_{ji}$ of the propagator, this ordering is equivalent to its reversal $0 \leq z_{\rho(n)} \leq \ldots \leq z_{\rho(3)} \leq z_{\rho(2)} \leq 1$, that is, only $\frac{1}{2}(n-1)!$ inequivalent cells need to be considered. Different cells benefit from different representations of the propagators, e.g. in situations with $z_2 < z_3$, it is preferable to use the expression

$$P_{23} := \omega_A(1,0) - \Gamma\left(\frac{1}{z_3}; z_2\right) - \Gamma\left(\frac{1}{0}; z_3\right) \quad \text{rather than} \quad P_{23} := \omega_A(1,0) - \Gamma\left(\frac{1}{z_2}; z_3\right) - \Gamma\left(\frac{1}{0}; z_2\right)$$
(3.19)

as done in eq. (3.17).

Compact expressions for A-cycle graph functions are tied to expressing the eMZVs in terms of a basis over $\mathbb{Q}[(2\pi i)^{\pm 1}]$ -combinations of MZVs. For certain ranges of their length and weight, an exhaustive list of such relations among eMZVs is available for download [71], but already for A-cycle graph functions at weight four, some of the intermediate steps exceed the scope of this website. In deriving the subsequent results on A-cycle graph functions of weight $w \leq 6$, we have expressed the eMZVs in terms of iterated Eisenstein integrals eq. (2.19) to automatically attain the desired basis decomposition. Using this method, the divergent eMZV $\omega_A(1,0)$ could be shown to drop out in all cases considered, which is a strong consistency check for our calculational setup.

3.2.4 A-cycle graph functions at weight four and beyond

The strategy outlined in the previous section gives rise to the following expressions for the three A-cycle graph functions at weight four:

$$\mathbf{A}\left[\textcircled{\textcircled{\baselineskip}}\right] = 15\,\omega_A(0,0,2)^2 - 30\,\omega_A(0,0,0,0,4) + 3\,\omega_A(0,0,4) - 24\,\omega_A(0,0,0,2,2) - 48\zeta_2\,\omega_A(0,0,0,0,2) + 13\zeta_2\,\omega_A(0,0,2) + \frac{343\zeta_4}{24}$$
(3.20)
$$\mathbf{A}\left[\textcircled{\baselineskip}\right] = \frac{1}{2}\,\omega_A(0,0,2)^2 - \frac{1}{2}\,\omega_A(0,0,0,0,4) - \omega_A(0,0,0,2,2)$$

$$\mathbf{A}[\boldsymbol{\leftarrow}] = \frac{1}{2} \omega_A(0, 0, 2) - \frac{1}{2} \omega_A(0, 0, 0, 0, 4) - \omega_A(0, 0, 0, 2, 2) + \frac{1}{3} \zeta_2 \omega_A(0, 0, 2) - 14 \zeta_2 \omega_A(0, 0, 0, 0, 2) + \frac{301}{180} \zeta_4$$
(3.21)

$$\mathbf{A}\left[\Box\right] = \omega_A(0,0,0,0,4) - \frac{1}{6}\,\omega_A(0,0,4) + \frac{4}{3}\,\omega_A(0,0,2)\zeta_2 - 8\,\omega_A(0,0,0,0,2)\zeta_2 + \frac{311\zeta_4}{360}\,.$$
 (3.22)

At weight five there are six A-cycle graph functions, for example

$$\mathbf{A}\left[\Box\right] = \frac{1}{90}\omega_{A}(0,5) + \frac{2}{3}\omega_{A}(0,0,0,5) - \frac{1}{3}\omega_{A}(0,0,2,3) + 2\omega_{A}(0,3)\omega_{A}(0,0,0,0,2) - 6\omega_{A}(0,0,0,0,0,5) + 2\omega_{A}(0,0,0,0,1,4) - \frac{1}{3}\omega_{A}(0,3)\zeta_{2} + \frac{8}{9}\omega_{A}(0,0,3,0)\zeta_{2} + 24\omega_{A}(0,0,0,0,0,3)\zeta_{2} - 16\omega_{A}(0,0,0,0,1,2)\zeta_{2} + \frac{2}{3}\zeta_{2}\zeta_{3} + 7\omega_{A}(0,0,1,0)\zeta_{4} - 52\omega_{A}(0,0,0,1,0,0)\zeta_{4}$$

$$(3.23)$$

and expressions of comparable complexity for $\mathbf{A}[\textcircled{\baselinedow}], \mathbf{A}[\textcircled{\baselinedow}], \mathbf{A}[\textcircled{\baselinedow}]$ and $\mathbf{A}[\textcircled{\baselinedow}]$ are displayed in appendix E. Analogous results at weight six are available from the authors.

3.3 Evaluating *B*-cycle graph functions

In this section, we compute modular transformations of A-cycle eMZVs. For this purpose it will be convenient to represent A-cycle graph functions in terms of iterated Eisenstein integrals eq. (2.19)

$$\begin{aligned} \mathbf{A}[\boldsymbol{\leftarrow}] &= -6\,\mathcal{E}_{0}(4,0) + \frac{1}{2}\zeta_{2} \\ \mathbf{A}[\boldsymbol{\leftarrow}] &= \frac{3}{2}\zeta_{3} - 6\,\mathcal{E}_{0}(4,0,0) - 60\,\mathcal{E}_{0}(6,0,0) \\ \mathbf{A}[\boldsymbol{\bigtriangleup}] &= \frac{1}{4}\zeta_{3} - \frac{3}{2}\,\mathcal{E}_{0}(4,0,0) - 60\,\mathcal{E}_{0}(6,0,0) \\ \mathbf{A}[\boldsymbol{\bigtriangleup}] &= -36\,\mathcal{E}_{0}(4,4,0,0) - 756\,\mathcal{E}_{0}(8,0,0,0) - 70\,\mathcal{E}_{0}(6,0,0,0) - \frac{1}{10}\,\mathcal{E}_{0}(4,0,0,0) + \frac{3}{8}\zeta_{4} \\ \mathbf{A}[\boldsymbol{\Box}] &= -840\,\mathcal{E}_{0}(8,0,0,0) - 40\,\mathcal{E}_{0}(6,0,0,0) + \frac{\zeta_{4}}{8} \end{aligned}$$

and we will now present two methods to compute their S-transformation. Both of these methods leave certain additive constants built from MZVs undetermined. These constants can be either determined numerically or by a method of Enriquez [22], which allows to infer constant terms of B-cycle eMZVs from the Drinfeld associator, see appendix B for more details.

In subsections 3.3.1 to 3.3.3, the method of obtaining *B*-cycle eMZVs from *A*-cycle eMZVs as developed by Brown is explained. An alternative method using differential equations is provided in subsection 3.3.4.

3.3.1 Conversion to Brown's iterated Eisenstein integrals

In this subsection we want to briefly recall the theory of iterated integrals of Eisenstein series, developed by Brown in ref. [59], and explain how one can use it to get the q-expansion of B-cycle eMZVs. The key idea is to express the iterated Eisenstein integrals appearing in A-cycle graph functions in terms of the iterated integrals

$$\mathcal{G}\left[\begin{smallmatrix} j_{1} & j_{2} & \dots & j_{r} \\ k_{1} & k_{2} & \dots & k_{r} \end{smallmatrix}; \tau\right] = \int_{\tau}^{i\infty} \mathrm{d}\tau_{r} \ \tau_{r}^{j_{r}} \,\mathrm{G}_{k_{r}}(\tau_{r}) \int_{\tau_{r}}^{i\infty} \mathrm{d}\tau_{r-1} \ \tau_{r-1}^{j_{r-1}} \,\mathrm{G}_{k_{r-1}}(\tau_{r-1}) \ \dots \int_{\tau_{2}}^{i\infty} \mathrm{d}\tau_{1} \ \tau_{1}^{j_{1}} \,\mathrm{G}_{k_{1}}(\tau_{1}) \,, \quad (3.26)$$

which already appeared in eq. (2.22), and are regularized as explained in subsection 2.2.2. The modular properties of the functions \mathcal{G} are known from ref. [59] (for a certain range of the powers j_i 's) and will be discussed in the next subsection.

The translation between the expressions eq. (2.18) for iterated Eisenstein integrals \mathcal{E} and eq. (3.26) can be conveniently extracted from the respective generating series

$$\mathbb{E}_{\underline{k}}(Y_0, Y_1, \dots, Y_r; \tau) := \sum_{p_0, p_1, \dots, p_r \ge 0} \frac{1}{(2\pi i)^{2p_0}} \left[\prod_{i=1}^r (2\pi i)^{k_i - 2p_i - 1} \right]$$

$$\times \mathcal{E}(0^{p_0}, k_1, 0^{p_1}, \dots, k_r, 0^{p_r}; \tau) Y_0^{p_0} Y_1^{p_1} \cdots Y_r^{p_r}$$
(3.27)
$$\mathbb{G}_{\underline{k}}(T_1, T_2, \dots, T_r; \tau) := \sum_{p_1, \dots, p_r \ge 0} \left[\prod_{i=1}^r \frac{1}{p_i!} \left(\frac{T_i}{2\pi i} \right)^{p_i} \right] \mathcal{G} \begin{bmatrix} p_1 & p_2 & \dots & p_r \\ k_1 & k_2 & \dots & k_r \end{bmatrix}; \tau \end{bmatrix}$$

with formal variables Y_i and T_i . Here and in later places, we are using multi-index notation $\underline{k} := (k_1, k_2, \ldots, k_r)$, i.e. eqns. (3.27) define two generating series for any fixed *r*-tuple \underline{k} . As will be shown in appendix D.3, the series in eqns. (3.27) are related via

$$\mathbb{E}_{\underline{k}}(Y_0, Y_1, \dots, Y_r; \tau) = \exp\left(\frac{\tau Y_r}{2\pi i}\right) \mathbb{G}_{\underline{k}}(Y_0 - Y_1, Y_1 - Y_2, \dots, Y_{r-1} - Y_r; \tau)$$
(3.28)
$$\mathbb{G}_{\underline{k}}(T_1, T_2, \dots, T_r; \tau) = \exp\left(-\frac{\tau U}{2\pi i}\right) \mathbb{E}_{\underline{k}}(T_1 + T_2 + \dots + T_r + U, T_2 + \dots + T_r + U, \dots, T_r + U, U; \tau),$$

where the dependence of the right hand side on the formal variable U drops thanks to shuffle relations. By isolating the coefficients of suitable monomials in the formal variables, eq. (3.28) translates into the following relations at depth one and two,

$$\mathcal{E}(0^{p_0}, k_1, 0^{p_1}; \tau) = (2\pi i)^{p_0 + p_1 - k_1 + 1} \sum_{\substack{\alpha_1 + \beta_1 = p_1 \\ p_0 ! \alpha_1 ! \beta_1 ! }} \frac{(-1)^{\alpha_1} \tau^{\beta_1}}{p_0 ! \alpha_1 ! \beta_1 !} \mathcal{G} \begin{bmatrix} p_0 + \alpha_1 \\ k_1 \end{bmatrix}, \qquad (3.29)$$

$$\mathcal{E}(0^{p_0}, k_1, 0^{p_1}, k_2, 0^{p_2}; \tau) = (2\pi i)^{p_0 + p_1 + p_2 - k_1 - k_2 + 2} \sum_{\substack{\alpha_1 + \beta_1 = p_1 \\ \alpha_2 + \beta_2 = p_2 }} \frac{(-1)^{\alpha_1 + \alpha_2} \tau^{\beta_2}}{p_0 ! \alpha_1 ! \beta_1 ! \alpha_2 ! \beta_2 !} \mathcal{G} \begin{bmatrix} p_0 + \alpha_1 & \beta_1 + \alpha_2 \\ k_1 & k_2 \end{bmatrix}, \qquad (3.29)$$

and conversely

$$\mathcal{G}\begin{bmatrix}p_1\\k_1;\tau\end{bmatrix} = (2\pi i)^{k_1-p_1-1}p_1! \mathcal{E}(0^{p_1},k_1;\tau)$$

$$\mathcal{G}\begin{bmatrix}p_1\\k_1\\k_2;\tau\end{bmatrix} = (2\pi i)^{k_1+k_2-p_1-p_2-2} \sum_{a+b=p_2} \frac{(p_1+a)!p_2!}{a!} \mathcal{E}(0^{p_1+a},k_1,0^b,k_2;\tau).$$
(3.30)

3.3.2 Modular transformations of Brown's iterated Eisenstein integrals

The modular transformation of Brown's iterated Eisenstein integrals eq. (3.26) can be compactly encoded in another generating function

$$I^{E}(\tau,\infty) = 1 + \int_{\tau}^{i\infty} \Theta^{E}(X_{1},Y_{1},\tau_{1}) + \int_{\tau}^{i\infty} \Theta^{E}(X_{2},Y_{2},\tau_{2}) \int_{\tau_{1}}^{i\infty} \Theta^{E}(X_{1},Y_{1},\tau_{1}) + \dots, \qquad (3.31)$$

where Eisenstein series are combined with non-commutative formal variables¹⁶ g_k

$$\Theta^E(X,Y,\tau) = \mathrm{d}\tau \sum_{k\geq 4} \mathrm{G}_k(\tau) \left(X - \tau Y\right)^{k-2} \mathbf{g}_k.$$
(3.32)

As a special case of a lemma proved by Brown in ref. [59], there exists a series C_S^E in infinitely many non-commutative variables \mathbf{g}_k and infinitely many pairs of commutative variables (X_i, Y_i)

¹⁶Brown developed the theory for the full space of modular forms. Here, we specialize his construction to iterated integrals of Eisenstein series only, so we keep his original notation, adding the superscript E which stands for Eisenstein. Moreover, we chose a different normalization convention for Eisenstein series.

such that 17

$$I^{E}(\tau,\infty) = \mathcal{C}^{E}_{S}I^{E}\left(-\frac{1}{\tau},\infty\right)|_{S}, \qquad I^{E}\left(-\frac{1}{\tau},\infty\right) = \left(\mathcal{C}^{E}_{S}\right)^{-1}|_{S}I^{E}(\tau,\infty)|_{S}, \qquad (3.33)$$

where $|_S$ acts on a function $F(X_i, Y_i)$ of the commutative variables X_i, Y_i according to

$$F(X_i, Y_i)|_S = F(-Y_i, X_i).$$
(3.34)

The series C_S^E does not depend on τ , and its coefficients are called *multiple modular values* (of Eisenstein series). In all cases relevant to the computation of *B*-cycle graph functions at weight $w \leq 7$, these coefficients are $\mathbb{Q}[2\pi i]$ -linear combinations of MZVs of known transcendentality whose composition can be obtained either numerically, using the fact that (by eq. (3.33))

$$\mathcal{C}_{S}^{E} = I^{E}(i,\infty) \left(I^{E}(i,\infty)|_{S} \right)^{-1}, \qquad (3.35)$$

or by matching with the method of Enriquez reviewed in appendix B.

The desired modular transformations of iterated Eisenstein integrals can be extracted from the series in eq. (3.33): To isolate the coefficients of any non-commutative word $\mathbf{g}_{k_1}\mathbf{g}_{k_2}\cdots\mathbf{g}_{k_r}$ in the above generating series $I^E(\tau,\infty)$ and \mathcal{C}^E_S , we will write $I^E(\tau,\infty)(k_1,k_2,\ldots,k_r)$ and $\mathcal{C}^E_S(k_1,k_2,\ldots,k_r)$, respectively. In terms of Brown's iterated Eisenstein integrals eq. (3.26), we find

$$I^{E}(\tau,\infty)(k_{1},k_{2},\ldots,k_{r}) = \sum_{j_{1}=0}^{k_{1}-2} \sum_{j_{2}=0}^{k_{2}-2} \cdots \sum_{j_{r}=0}^{k_{r}-2} \left[\prod_{i=1}^{r} (-1)^{j_{i}} \binom{k_{i}-2}{j_{i}} \right]$$

$$\times \mathcal{G} \left[\begin{smallmatrix} j_{1} & j_{2} & \dots & j_{r} \\ k_{1} & k_{2} & \dots & k_{r} \end{smallmatrix}; \tau \right] X_{1}^{k_{1}-2-j_{1}} X_{2}^{k_{2}-2-j_{2}} \cdots X_{r}^{k_{r}-2-j_{r}} Y_{1}^{j_{1}} Y_{2}^{j_{2}} \cdots Y_{r}^{j_{r}}.$$

$$(3.36)$$

In the case of a single integration, one gets abelian cocycles $C_S^E(k)$, also called period polynomials, very well known after the work of Eichler, Shimura and Manin in the case of cusp forms, and worked out for Eisenstein series in refs. [72, 73]. In particular, it was proven that

$$\mathcal{C}_{S}^{E}(2k) = \frac{2\pi i}{2k-1} \Big(\zeta_{2k-1} (Y^{2k-2} - X^{2k-2}) - (2\pi i)^{2k-1} \sum_{i=1}^{k-1} \frac{B_{2i} B_{2k-2i}}{(2i)!(2k-2i)!} X^{2i-1} Y^{2k-2i-1} \Big) \,. \tag{3.37}$$

3.3.3 B-cycle eMZVs from Brown's iterated Eisenstein integrals

The computation of *B*-cycle eMZVs from eq. (3.33) follows a simple idea which has already been used in ref. [58] at depth one: once the underlying $\mathcal{E}(\underline{k}; -\frac{1}{\tau})$ are related to the coefficients eq. (3.26) of the series $I^E(\tau, \infty)$, one can use Brown's result. In particular, by inserting eq. (3.36) into the special cases of

$$I^{E}\left(-\frac{1}{\tau},\infty\right)(k_{1}) = I^{E}(\tau,\infty)(k_{1})|_{S} - \mathcal{C}^{E}_{S}(k_{1})|_{S}$$

$$I^{E}\left(-\frac{1}{\tau},\infty\right)(k_{1},k_{2}) = I^{E}(\tau,\infty)(k_{1},k_{2})|_{S} - \mathcal{C}^{E}_{S}(k_{1})|_{S}I^{E}(\tau,\infty)(k_{2})|_{S}$$

$$+ \mathcal{C}^{E}_{S}(k_{1})|_{S}\mathcal{C}^{E}_{S}(k_{2})|_{S} - \mathcal{C}^{E}_{S}(k_{1},k_{2})|_{S}$$

$$(3.38)$$

$$(3.38)$$

$$(3.39)$$

 $^{^{17}}$ In [59], the position of the factors on the right hand side is reversed, because of our opposite convention for iterated integrals.

of eq. (3.33), we arrive at

$$\mathcal{G}\begin{bmatrix} j_1\\k_1; -\frac{1}{\tau} \end{bmatrix} = (-1)^{j_1} \mathcal{G}\begin{bmatrix} k_1 - 2 - j_1\\k_1; \tau \end{bmatrix} - \binom{k_1 - 2}{j_1}^{-1} c_{k_1 - 2 - j_1}(k_1),$$
(3.40)

$$\mathcal{G}\begin{bmatrix} j_1 & j_2 \\ k_1 & k_2 \end{bmatrix}; -\frac{1}{\tau} = (-1)^{j_1+j_2} \mathcal{G}\begin{bmatrix} k_1-2-j_1 & k_2-2-j_2 \\ k_1 & k_2 \end{bmatrix}; \tau = (-1)^{j_2} \binom{k_1-2}{j_1} c_{k_1-2-j_1}(k_1) \mathcal{G}\begin{bmatrix} k_2-2-j_2 \\ k_2 \end{bmatrix}; \tau = (-1)^{j_2} \binom{k_1-2}{j_1} c_{k_1-2-j_1}(k_1) \mathcal{G}\begin{bmatrix} k_2-2-j_2 \\ k_2 \end{bmatrix}; \tau = (-1)^{j_2} \binom{k_1-2}{j_1} c_{k_1-2-j_1}(k_1) \mathcal{G}\begin{bmatrix} k_2-2-j_2 \\ k_2 \end{bmatrix}; \tau = (-1)^{j_2} \binom{k_1-2-j_1}{j_1} \binom{k_2-2-j_2}{j_2}; \tau = (-1)^{j_2} \binom{k_1-2-j_1}{j_1} \binom{k_2-2-j_2}{j_1}; \tau = (-1)^{j_2} \binom{k_1-2-j_1}{j_1} \binom{k_2-2-j_2}{j_2}; \tau = (-1)^{j_2} \binom{k_1-2-j_1}{j_1} \binom{k_2-2-j_2}{j_1}; \tau = (-1)^{j_2} \binom{k_1-2-j_1}{j_1} \binom{k_2-2-j_2}{j_2}; \tau = (-1)^{j_2} \binom{k_1-2-j_1}{j_1} \binom{k_1-2-j_1}{j_2}; \tau = (-1)^{j_2} \binom{k_1-2-j_1}{j_1} \binom{k_1-2-j_1}{j_1} \binom{k_1-2-j_1}{j_1} \binom{k_1-2-j_1}{j_2}; \tau = (-1)^{j_2} \binom{k_1-2-j_1}{j_1} \binom{k_1-2-j_$$

where the quantities $c_{\dots}(k_1, \ldots)$ are defined by the expansion

$$\mathcal{C}_{S}^{E}(k_{1},\ldots,k_{r}) = \sum_{j_{1}=0}^{k_{1}-2} \cdots \sum_{j_{r}=0}^{k_{r}-2} c_{j_{1},\ldots,j_{r}}(k_{1},\ldots,k_{r}) X_{1}^{k_{1}-2-j_{1}} \cdots X_{r}^{k_{r}-2-j_{r}} Y_{1}^{j_{1}} \cdots Y_{r}^{j_{r}}.$$
 (3.42)

One must be warned that not all $\mathcal{E}(\underline{k}; -\frac{1}{\tau})$ can be computed in this way: if \underline{k} contains too many zeros, eq. (3.28) gives rise to $\mathcal{G}\begin{bmatrix}j_1 & j_2 & \dots & j_r\\k_1 & k_2 & \dots & k_r\end{bmatrix}$ with $j_i \notin \{0, 1, \dots, k_i - 2\}$ which are excluded from the building block eq. (3.32) of Brown's series eq. (3.31). However, this method always applies to the special linear combinations of $\mathcal{E}(\underline{k}; -\frac{1}{\tau})$ given by eMZVs and therefore selected by a certain derivation algebra [74, 75, 24]: This is a consequence of Proposition 6.3 of ref. [29], and in fact, the linear combinations of $\mathcal{E}(\underline{k}; -\frac{1}{\tau})$ descending from eMZVs are contained in a proper subset of the iterated integrals eq. (2.22). Putting all of this together, one obtains a closed formula at depth one for $p_0 + p_1 \leq k_1 - 2$

$$\mathcal{E}(0^{p_0}, k_1, 0^{p_1}; -\frac{1}{\tau}) = (-1)^{p_1} (2\pi i)^{p_0 + p_1 + 1 - k_1} \sum_{\alpha + \beta = p_1} \frac{(k_1 - 2 - p_0 - \alpha)!}{\alpha! \beta! \tau^{\beta}}$$

$$\times \left((-2\pi i)^{p_0 + \alpha} \mathcal{E}(0^{k_1 - 2 - p_0 - \alpha}, k_1; \tau) - \frac{(p_0 + \alpha)!}{(k_1 - 2)!} c_{k_1 - 2 - p_0 - \alpha}(k_1) \right),$$
(3.43)

and higher-depth expressions such as^{18}

$$(2\pi i)^{6} \mathcal{E}(6,4;-\frac{1}{\tau}) = \frac{\zeta_{3,5}}{75} + \frac{\zeta_{3}\zeta_{5}}{15} - \frac{503\zeta_{8}}{10800} - \frac{2\zeta_{5}}{5} \mathcal{E}(0,0,4;\tau) + 48 \big(\mathcal{E}(0,0,0,0,6,0,0,4;\tau) + 5 \mathcal{E}(0,0,0,0,0,6,0,4;\tau) + 15 \mathcal{E}(0,0,0,0,0,0,6,4;\tau) \big), \quad (3.44)$$

as well as (setting $T = \pi \tau$)

$$\begin{aligned} \mathcal{E}(4,4,0,0;-\frac{1}{\tau}) &= \mathcal{E}(0^2,4,4;\tau) - \frac{\zeta_3 \,\mathcal{E}(4;\tau)}{6} + \frac{209\pi^4}{11664000} \\ &+ \frac{i}{T} \Big(\,\mathcal{E}(0^2,4,0,4;\tau) + 3 \,\mathcal{E}(0^3,4,4;\tau) - \frac{\zeta_3 \,\mathcal{E}(0,4;\tau)}{6} + \frac{\zeta_2 \zeta_3}{360} - \frac{5\zeta_5}{432} \Big) \\ &+ \frac{1}{2T^2} \Big(- \,\mathcal{E}(0^2,4,0^2,4;\tau) - 3 \,\mathcal{E}(0^3,4,0,4;\tau) - 6 \,\mathcal{E}(0^4,4,4;\tau) + \frac{\zeta_3 \,\mathcal{E}(0^2,4;\tau)}{6} - \frac{\zeta_3^2}{72} \Big) \,, \end{aligned}$$
(3.45)

$$\mathcal{E}(4,0,4,0,0;-\frac{1}{\tau}) + 3\mathcal{E}(4,4,0,0,0;-\frac{1}{\tau}) = -\frac{\pi^4}{108}\mathcal{E}(4;\tau) - 2\pi^2\mathcal{E}(0,4,4;\tau) - \frac{\zeta_2\zeta_3}{360} + \frac{5\zeta_5}{432}\mathcal{E}(0,1;\tau) - \frac{\zeta_2\zeta_3}{360} + \frac{5\zeta_5}{432}\mathcal{E}(0,1;\tau) - \frac{\zeta_2\zeta_3}{360} + \frac{\zeta_2}{36} +$$

¹⁸We do not have a closed formula like eq. (3.37) for multiple modular values at depth ≥ 2 , so for the purposes of this paper, we contented ourselves to guessing their representations as MZVs based on five hundred digits numerical approximations. In all cases up to weight six, these representations have been confirmed through the analytic method of appendix B.

$$+ \frac{i\pi^{2}}{T} \Big(\frac{\zeta_{3}}{3} \mathcal{E}(4;\tau) - \frac{\zeta_{2}}{18} \mathcal{E}(0,4;\tau) - 2 \mathcal{E}(0,4,0,4;\tau) - 6 \mathcal{E}(0,0,4,4;\tau) - \frac{167\zeta_{4}}{32400} \Big) + \frac{\pi^{2}}{T^{2}} \Big(-\frac{\zeta_{3}}{3} \mathcal{E}(0,4;\tau) + \frac{\zeta_{2}}{36} \mathcal{E}(0,0,4;\tau) - \frac{\zeta_{2}\zeta_{3}}{540} - \frac{5\zeta_{5}}{432} + \mathcal{E}(0,4,0,0,4;\tau) + 4 \mathcal{E}(0,0,4,0,4;\tau) + 9 \mathcal{E}(0,0,0,4,4;\tau) \Big) - \frac{i\pi^{2}}{T^{3}} \Big(\frac{\zeta_{3}}{6} \mathcal{E}(0,0,4;\tau) - \mathcal{E}(0,0,4,0,0,4;\tau) - 3 \mathcal{E}(0,0,0,4,0,4;\tau) - 6 \mathcal{E}(0,0,0,0,4,4;\tau) - \frac{\zeta_{3}^{2}}{72} \Big)$$
(3.46)

and the modular transformations given in appendix D.4. In all examples of A-cycle graph functions tested so far we indeed landed on iterated integrals of the kind eq. (3.26) with $j_i \leq k_i - 2$, whose S-transform can therefore be computed as explained above. Note that the relative factor of 3 on the left hand side of eq. (3.46) is crucial to obey this criterion.

In order to determine the q-expansion of B-cycle graph functions, the iterated Eisenstein integrals on the right hand side of eq. (3.44) and the above depth-two examples need to be cast into the form $\mathcal{E}_0(k,...)$ with $k \neq 0$ such that eq. (2.21) becomes applicable. This can always be achieved by first applying shuffle relations such as $\mathcal{E}(0,4;\tau) = \mathcal{E}(0;\tau)\mathcal{E}(4;\tau) - \mathcal{E}(4,0;\tau)$ and $\mathcal{E}(0,0,4;\tau) = \mathcal{E}(4,0,0;\tau) - \mathcal{E}(0)\mathcal{E}(4,0;\tau) + \mathcal{E}(0,0;\tau)\mathcal{E}(4;\tau)$ to attain the form $\mathcal{E}(k,...)$ with $k \neq 0$. Then, the conversion between $\mathcal{E}(...)$ and $\mathcal{E}_0(...)$ follows from the definitions eqs. (2.18) and (2.19) of the respective iterated Eisenstein integrals, along with

$$\mathcal{E}(\underbrace{0,0,\ldots,0}_{n};\tau) = \mathcal{E}_{0}(\underbrace{0,0,\ldots,0}_{n};\tau) = \frac{1}{n!}(2\pi i \tau)^{n}, \qquad (3.47)$$

for instance $\mathcal{E}(4,0;\tau) = \mathcal{E}_0(4,0;\tau) + \frac{\pi^2 \tau^2}{360}$ and $\mathcal{E}(4,0,0;\tau) = \mathcal{E}_0(4,0,0;\tau) + \frac{i\pi^3 \tau^3}{540}$. At depth larger than one, this might introduce further instances of $\mathcal{E}_0(0,\ldots)$ with zero in the first entry which call for additional shuffle manipulations. This can be illustrated through the following example at depth two

$$\mathcal{E}(4,4,0,0) = \mathcal{E}_{0}(4,4,0,0) - \frac{2\zeta_{4}}{(2\pi i)^{4}} \left[\mathcal{E}_{0}(4,0,0,0) + \mathcal{E}_{0}(0,4,0,0) \right] + \left(\frac{2\zeta_{4}}{(2\pi i)^{4}}\right)^{2} \mathcal{E}_{0}(0,0,0,0)$$
$$= \mathcal{E}_{0}(4,4,0,0) + \frac{1}{360} \mathcal{E}_{0}(4,0,0,0) - \frac{i\pi\tau}{360} \mathcal{E}_{0}(4,0,0) + \frac{\pi^{4}\tau^{4}}{777600} , \qquad (3.48)$$

where we have inserted $\mathcal{E}_0(0, 4, 0, 0) = \mathcal{E}_0(0) \mathcal{E}_0(4, 0, 0) - 3 \mathcal{E}_0(4, 0, 0, 0)$ in passing to the second line. A generating series for the most general case can be found in appendix D.2.

3.3.4 B-cycle eMZVs from differential equations

As an alternative and recursive method to determine modular transformations of iterated Eisenstein integrals eqs. (2.18) and (2.19), one can take advantage of the differential equation

$$\tau^2 2\pi i \partial_\tau \mathcal{E}(A, 2k| - \frac{1}{\tau}) = -(2\pi i)^{2-2k} \tau^{2k} \operatorname{G}_{2k}(\tau) \mathcal{E}(A| - \frac{1}{\tau})$$
(3.49)

$$\tau^2 2\pi i \partial_\tau \mathcal{E}_0(A, 2k| - \frac{1}{\tau}) = -(2\pi i)^{2-2k} \left[\tau^{2k} \, \mathcal{G}_{2k}^0(\tau) + 2\,\zeta_{2k}(\tau^{2k} - 1) \right] \mathcal{E}_0(A| - \frac{1}{\tau}) \tag{3.50}$$

for $k \neq 0$ as well as

$$\tau^2 2\pi i \partial_\tau \,\mathcal{E}(A,0|-\frac{1}{\tau}) = (2\pi i)^2 \,\mathcal{E}(A|-\frac{1}{\tau}) \,, \quad \tau^2 2\pi i \partial_\tau \,\mathcal{E}_0(A,0|-\frac{1}{\tau}) = (2\pi i)^2 \,\mathcal{E}_0(A|-\frac{1}{\tau}) \,, \quad (3.51)$$

resulting from their recursive definition. With this method, the expression for $\mathcal{E}(n, 0, 0, \dots, 0; -\frac{1}{\tau})$ in eq. (3.44) with j-1 successive zeros follows from integrating eqs. (3.49) and (3.51) j times, and the multiple modular values eq. (3.37) arise as the integration constants of the respective j steps. So the modular transformation is performed separately on each integration kernel in the iterated Eisenstein integrals. At higher depth, these integration constants can be obtained numerically or by matching with Enriquez's method reviewed in appendix B. In all cases we have checked the approach of this subsection matches the results obtained from Brown's theory.

3.3.5 Examples of *B*-cycle graph functions

In applying the modular transformation eq. (3.44) at depth one to the A-cycle graph functions in eq. (3.25), we have to take the offsets between the $\mathcal{E}(\ldots)$ and $\mathcal{E}_0(\ldots)$ into account. From the discussion around eq. (3.47), we have

$$\mathcal{E}_{0}(4,0;-\frac{1}{\tau}) = \mathcal{E}(4,0;-\frac{1}{\tau}) - \frac{\pi^{2}}{360\tau^{2}} = \mathcal{E}(4,0;\tau) + \frac{i}{\pi\tau} \mathcal{E}(4,0,0;\tau) - \frac{i\zeta_{3}}{6\pi\tau} + \frac{\pi^{2}}{216} - \frac{\pi^{2}}{360\tau^{2}}$$
$$= \mathcal{E}_{0}(4,0;\tau) + \frac{i}{T} \mathcal{E}_{0}(4,0,0;\tau) + \frac{T^{2}}{1080} + \frac{\zeta_{2}}{36} - \frac{i\zeta_{3}}{6T} - \frac{\zeta_{4}}{4T^{2}}$$
(3.52)

with $T := \pi \tau$, and by similar manipulations,

$$\mathcal{E}_0(4,0,0;-\frac{1}{\tau}) = \frac{\pi^2}{T^2} \mathcal{E}_0(4,0,0;\tau) + \frac{iT\zeta_2}{90} + \frac{\zeta_3}{6} - \frac{5i\zeta_4}{6T} - \frac{\zeta_2\zeta_3}{T^2} + \frac{7i\zeta_6}{4T^3}.$$
 (3.53)

Following the same strategy at higher weight, one obtains the following expressions for B-cycle graph functions:

$$\mathbf{B}\left[\frown\right] = -\frac{T^{2}}{180} + \frac{\zeta_{2}}{3} + \frac{i\zeta_{3}}{T} + \frac{3\zeta_{4}}{2T^{2}} - 6\mathcal{E}_{0}(4,0) - \frac{6i\mathcal{E}_{0}(4,0,0)}{T} \\
\mathbf{B}\left[\bigtriangleup\right] = \frac{iT^{3}}{3780} - \frac{iT\zeta_{2}}{60} + \frac{i\zeta_{4}}{T} + \frac{3\zeta_{2}\zeta_{3}}{2T^{2}} - \frac{3\zeta_{5}}{2T^{2}} - \frac{i\zeta_{6}}{8T^{3}} \\
- 60\mathcal{E}_{0}(6,0,0) - \frac{180i\mathcal{E}_{0}(6,0,0,0)}{T} + \frac{180\mathcal{E}_{0}(6,0,0,0)}{T^{2}} - \frac{9\mathcal{E}_{0}(4,0,0)\zeta_{2}}{T^{2}} \\
\mathbf{B}\left[\Box\right] = \frac{T^{4}}{75600} - \frac{T^{2}\zeta_{2}}{945} + \frac{13\zeta_{4}}{180} - \frac{4\zeta_{6}}{3T^{2}} + \frac{4i\zeta_{2}\zeta_{5}}{T^{3}} - \frac{5i\zeta_{7}}{2T^{3}} + \frac{95\zeta_{8}}{24T^{4}} \\
- 840\mathcal{E}_{0}(8,0,0,0) - \frac{5040i\mathcal{E}_{0}(8,0,0,0,0)}{T} + \frac{12600\mathcal{E}_{0}(8,0,0,0,0,0)}{T^{2}} \\
+ \frac{12600i\mathcal{E}_{0}(8,0,0,0,0,0,0,0)}{T^{3}} - \frac{240\mathcal{E}_{0}(6,0,0,0)\zeta_{2}}{T^{2}} - \frac{480i\mathcal{E}_{0}(6,0,0,0,0,0)\zeta_{2}}{T^{3}}.$$
(3.54)

We have rewritten the integrals \mathcal{E} following from the above modular transformations in terms of \mathcal{E}_0 to make the *q*-expansion of the *B*-cycle graph functions accessible from eq. (2.21). Moreover, this highlights the property of *B*-cycle eMZVs that coefficients of q^n are Laurent polynomials in τ . The change of variables from $\pi\tau$ to *T* absorbs all negative powers of π and yields \mathbb{Q} -linear combinations of MZVs as Laurent coefficient, as remarked in eq. (2.13). Hence, these Laurent polynomials can be thought of as the open-string antecedents of the zero modes $\mathbf{d}[\mathcal{G}]$ of modular graph functions discussed in section 2.3.1. Accordingly, we will denote the coefficient of q^0 in the *B*-cycle graph function $\mathbf{B}[\mathcal{G}]$ by $\mathbf{b}[\mathcal{G}]$, e.g. one has

$$\mathbf{b}\left[\frown\right] = -\frac{T^2}{180} + \frac{\zeta_2}{3} + \frac{i\zeta_3}{T} + \frac{3\zeta_4}{2T^2}, \qquad \mathbf{b}\left[\bigtriangleup\right] = \frac{iT^3}{3780} - \frac{iT\zeta_2}{60} + \frac{i\zeta_4}{T} + \frac{3\zeta_2\zeta_3}{2T^2} - \frac{3\zeta_5}{2T^2} - \frac{i\zeta_6}{8T^3}, \quad (3.55)$$

and a method to determine such $\mathbf{b}[\mathcal{G}]$ from the Drinfeld associator is presented in appendix B. This method goes back to Enriquez [22], where a generating series for the constant terms of Acycle and B-cycle eMZVs is given, and a procedure to extract the constant terms of individual A-cycle eMZVs is explained in section 2.3 of ref. [24].

At depth two, the modular transformation eq. (3.45) of $\mathcal{E}(4,4,0,0)$ leads to

$$\mathbf{B}\left[\bigtriangleup\right] - \frac{9}{10} \mathbf{B}\left[\Box\right] = -\frac{T^4}{324000} - \frac{17T^2\zeta_2}{18900} - \frac{iT\zeta_3}{180} + \frac{253\zeta_4}{1800} + \frac{5i\zeta_5}{12T} - \frac{49\zeta_6}{80T^2} \\
+ \frac{\zeta_3^2}{4T^2} - \frac{3i\zeta_3\zeta_4}{2T^3} + \frac{17i\zeta_5\zeta_2}{5T^3} + \frac{277\zeta_8}{48T^4} + \left(\frac{iT}{30} - \frac{3\zeta_3}{T^2} + \frac{9i\zeta_4}{T^3}\right)\mathcal{E}_0(4,0,0) + \frac{9\mathcal{E}_0(4,0,0)^2}{T^2} \\
- \frac{36i}{T}\left[\mathcal{E}_0(4,0,4,0,0) + 3\mathcal{E}_0(4,4,0,0,0) + \frac{1}{360}\mathcal{E}_0(4,0,0,0,0)\right] \quad (3.56) \\
- 36\left[\mathcal{E}_0(4,4,0,0) + \frac{1}{360}\mathcal{E}_0(4,0,0,0)\right] - \frac{204\zeta_2\mathcal{E}_0(6,0,0,0)}{T^2} - \frac{408i\zeta_2\mathcal{E}_0(6,0,0,0,0)}{T^3},$$

where the specific linear combination of *B*-cycle graph functions will be motivated in section 4.2. Note that the combination $\mathcal{E}_0(4,4,0,0) + \frac{1}{360} \mathcal{E}_0(4,0,0,0)$ in the last line can be recombined to $\mathcal{E}(4,4,0,0)$ according to eq. (3.48), and a similar statement applies to the length-five combination in the third line of eq. (3.56).

By the modular transformation eq. (D.19), the A-cycle graph functions eqs. (3.23) and (3.24) at weight five are mapped to the B-cycle graph function

For reasons to be explained in subsection 4.3 below, we have suppressed terms of the form $\pi^{2k} \mathcal{E}_0(...)T^m$ with $k \geq 1$ and $m \in \mathbb{Z}$ and refer to their omission by mod ζ_2 . The modular transformations in appendix D.4 lead to similar expressions for *B*-cycle graph functions at weight six which are available from the authors upon request. We have also determined numerically a Laurent polynomial at weight seven

$$\mathbf{b}\left[\textcircled{\textcircled{6}}\right] = \frac{31iT^7}{700539840} - \frac{5251iT^5\zeta_2}{233513280} + \frac{T^4\zeta_3}{3888} + \frac{7405iT^3\zeta_4}{598752} - \frac{119T^2\zeta_5}{2592} - \frac{31T^2\zeta_2\zeta_3}{864} \\ - \frac{11iT\zeta_3^2}{216} - \frac{15527iT\zeta_6}{10368} + \frac{21\zeta_7}{32} + \frac{67\zeta_2\zeta_5}{27} + \frac{167\zeta_3\zeta_4}{48} + \frac{23i\zeta_3\zeta_5}{3T} + \frac{80017i\zeta_8}{1296T} - \frac{3i\zeta_2\zeta_3^2}{T}$$

I	$25\zeta_3^3$	$7115\zeta_{s}$	$_9 21\zeta_2\zeta_7$	$35\zeta_4\zeta_5$ _ ($6613\zeta_{3}\zeta_{6}$	$75i\zeta_5^2$	$1245i\zeta_3\zeta_7$	$48i\zeta_{3,5}\zeta_2$	2
Т	$4T^{2}$	$144T^{2}$	$\frac{1}{2}$ T^2	$-\frac{1}{6T^2}$	$288T^{2}$ +	$4T^{3}$	$8T^{3}$	T^3	_
	$443i\zeta_2$	$\zeta_3\zeta_5$	$275i\zeta_{3}^{2}\zeta_{4}$	$941869i\zeta_{10}$	$9573\zeta_{11}$	$18\zeta_{3,5,3}$	$405\zeta_3^2\zeta_5$	$195\zeta_2\zeta_3$	3 3
+	T^3	3 —	$-8T^{3}$ +	$5760T^3$	$-\frac{16T^4}{16T^4}$	T^4	$-4T^{4}$	$+ \frac{1}{2T^4}$	_
	27745	ζ5ζ6 Ξ	$3795\zeta_4\zeta_7$	$17731\zeta_{3}\zeta_{8}$	$15875\zeta_2\zeta_9$	$2475i\zeta$	$\zeta_5 \zeta_7 + 1128$	$5i\zeta_3\zeta_9$	
+	487	14	$16T^4$ +	$-16T^4$ +	$12T^{4}$	$+ - 4T^{\xi}$	5 +	$\overline{T^5}$	
	$90i\zeta_{3,5}$	$\zeta_4 = 45$	$50i\zeta_{3,7}\zeta_{2}$	$165i\zeta_3\zeta_4\zeta_5$	$3375i\zeta_2\zeta_5^2$	$3335i\zeta$	$\zeta_3^2 \zeta_6 = 396$	$0i\zeta_2\zeta_3\zeta_7$	
_	T^5		$7T^5$ +	$2T^{5}$ –	$7T^{5}$	$-4T^{\xi}$	5 – –	$7T^{5}$	
	930919	$945i\zeta_{12}$	$1575\zeta_{13}$	$13275\zeta_2\zeta_{11}$	$7425\zeta_4\zeta_4$	$\frac{1}{9}$ 1294	$65\zeta_6\zeta_7$ 2	$233525\zeta_5\zeta_8$	
_	1105	$56T^{5}$	$+ T^{6}$	$4T^{6}$	$-\frac{1}{8T^{6}}$	1	$6T^{6}$	$48T^{6}$	
	16005	$3\zeta_3\zeta_{10}$	15301285i	$i\zeta_{14}$				(2 5 8
_	64	T^{6} +	$768T^{7}$					(5.50

comprising the depth-three MZV $\zeta_{3,5,3}$ along with T^{-4} which will be argued to harmonize with the Laurent polynomial eq. (2.43) of the corresponding modular graph function.

4 Open versus closed strings

In this section, we are going to establish and discuss the relation and connection between openstring graph functions and modular graph functions. The reason and origin for our investigations is a stunning similarity of the relations satisfied by open-string graph functions and their corresponding modular graph functions: in subsection 4.1 we are going to spell out commonalities and differences in order to establish a clear starting point. Given this similarity, it is an obvious question, whether modular graph functions can be eventually calculated from their open-string analogues. Anticipating the main result of this article, the answer is indeed positive: we can obtain modular graph functions from A-cycle graph functions performing the operations noted at the arrows in figure 3. The two different paths which can be taken in order to obtain modular



Figure 3: Two paths for the calculation of modular graph functions

graph functions from A-cycle graph functions are as follows:

- the first path starts from A-cycle graph functions and employs the similarity between the τ -derivative on the A-cycle graph functions and the Cauchy–Riemann derivative acting on modular graph functions. Using the appropriate derivatives multiple times on both sides of the correspondence allows to successively infer the elements of the modular graph functions from their A-cycle graph analogues. This method is described in subsection 4.2.
- for the second path one converts A-cycle graph functions into B-cycle ones. Following this

step, the projection esv is applied, which we conjecture to be an elliptic analogue of the single-valued projection sv mentioned in eq. (1.2). While the conversion from A- to B-cycle graph functions using a modular transformation has been described in subsection 3.3, the map esv will be described and discussed in subsection 4.3.

Both methods yield the same results, which are simultaneously in agreement with all expressions for modular graph functions calculated before [1-3].

4.1 Comparing relations among *A*-cycle graph functions with relations among modular graph functions

Given the common graphical representation of A-cycle graph functions and modular graph functions it is tempting to investigate, whether the known relations for modular graph functions reviewed in subsection 2.3.2 have an echo for A-cycle graph functions.

The simplest relation among modular graph functions, $\mathbf{D}[\bigcirc] - \mathbf{D}[\bigtriangleup] = \zeta_3$ at weight three, translates into

$$\mathbf{A}\left[\boldsymbol{\leftrightarrow}\right] - \mathbf{A}\left[\boldsymbol{\bigtriangleup}\right] = \frac{1}{2}\,\zeta_3 + 6\,\zeta_2\,\omega_A(0,1,0,0)\,,\tag{4.1}$$

and the weight-four relation eq. (2.46) leads to

$$\mathbf{A}\left[\textcircled{\clubsuit}\right] - 24\,\mathbf{A}\left[\textcircled{\bigtriangleup}\right] + 18\,\mathbf{A}\left[\fbox{]}\right] - 3\,\mathbf{A}\left[\textcircled{\clubsuit}\right]^2 = 144\,\zeta_2\,\omega_A(0,0,0,0,2) - 24\,\zeta_2\,\omega_A(0,0,2) - \frac{31}{2}\zeta_4\,,\tag{4.2}$$

where the right-hand sides have been obtained by simply plugging in our results from subsection 3.2.

The right-hand sides of the corresponding equations for modular graph functions, eqs. (2.45) and (2.46) read ζ_3 and 0, respectively. This is rather suggestive: a relation between A-cycle graph functions might imply a valid relation between modular graph functions by formally replacing

$$\omega_A(\underline{m})\,\zeta_{\underline{n}} \to \omega_A(\underline{m})\,\zeta_n^{\rm sv}\,,\tag{4.3}$$

where the single-valued projection of MZVs has been discussed in eqs. (1.2) and (1.3), and we remind the reader of the multi-index notation $\underline{n} = (n_1, n_2, \ldots, n_r)$. The ad-hoc prescription eq. (4.3) has the desired effect of replacing $\frac{1}{2}\zeta_3$ with ζ_3 on the right hand side of eq. (4.1)¹⁹. Similarly, ζ_4 is mapped to zero on the right hand side of eq. (4.2), and all instances of $\zeta_2 \omega_A(\underline{m})$ are suppressed. For brevity of notation below, let X be the rational vector space generated by products of classical and elliptic MZVs vanishing after applying eq. (4.3). That is

$$X = \langle \zeta_2 \,\omega_A(\underline{m}), \zeta_4 \,\omega_A(\underline{m}), \dots, (2 \,\zeta_{3,5} + 5 \,\zeta_3 \,\zeta_5) \omega_A(\underline{m}), \dots \rangle_{\mathbb{Q}} \,. \tag{4.4}$$

In the equations below we will write "mod X", which means that we are not writing terms from the space X. At weight five, the expressions in eqs. (3.23), (3.24) and appendix E for A-cycle

¹⁹When eMZVs are expressed in terms of iterated Eisenstein integrals \mathcal{E} , the prescription in eq. (4.3) might seem to be tension with the intuition from the single-valued projection of MZVs. For instance, $\zeta_2 \omega_A(0, 1, 0, 0|\tau)$ can be represented as $\frac{1}{8}\zeta_3 - \frac{3}{4}\mathcal{E}_0(4, 0, 0; q)$. Demanding consistency after application of eq. (4.3) to both expressions would yield a constraint for a replacement of $\mathcal{E}_0(4, 0, 0; q)$. As will become clear in subsection 4.3, the formulation of eq. (4.3) in terms of ω_A is very natural after converting the A-cycle eMZVs ω_A to B-cycle eMZVs ω_B by a modular transformation.

graph functions lead to the relations

$$40 \mathbf{A} \left[\bigtriangleup \right] = 300 \mathbf{A} \left[\bigtriangleup \right] + 120 \mathbf{A} \left[\bigtriangleup \right] \mathbf{A} \left[\bigtriangleup \right] - 276 \mathbf{A} \left[\bigtriangleup \right] + \frac{7}{2} \zeta_5 \mod X$$
(4.5)

$$\mathbf{A}[\textcircled{\textcircled{}}] = 60 \,\mathbf{A}[\fbox{]} + 10 \,\mathbf{A}[\textcircled{\textcircled{}}] \,\mathbf{A}[\textcircled{\textcircled{}}] - 48 \,\mathbf{A}[\swarrow] + 8\zeta_5 \mod X \tag{4.6}$$

$$10 \mathbf{A} \left[\bigtriangleup \right] = 20 \mathbf{A} \left[\bigsqcup \right] - 4 \mathbf{A} \left[\bigtriangleup \right] + \frac{3}{2} \zeta_5 \mod X$$

$$(4.7)$$

$$30 \mathbf{A} \left[\bigoplus \right] = 12 \mathbf{A} \left[\bigoplus \right] + \frac{\zeta_5}{2} \mod X, \qquad (4.8)$$

which – upon employing eq. (4.3) – yield relations (2.47) to (2.50) among modular graph functions. The validity of the connection between A-cycle graph functions and modular graph functions described above has been also checked for relations between A-cycle graph functions of weight six, see appendix F – applying eq. (4.3) reproduces the relations eq. (F.1) among modular graph functions from open-string input²⁰.

Note that the prescription in eq. (4.3) is ill-defined as it does depend on the particular representations of eMZVs. In particular, there exist many relations among eMZVs and classical MZVs: For instance, the combination $2 \omega_A(0,2,2) + \omega_A(0,0,4) = 3\zeta_4$ should in principle be annihilated by applying eq. (4.3), but the definition in eq. (4.3) leaves both terms $\omega_A(0,2,2)$ and $\omega_A(0,0,4)$ inert. However, this does not affect the statement of the following conjecture: Given a polynomial \mathcal{P} in A-cycle graph functions and MZVs such that

$$\mathcal{P}(\mathbf{A}[\mathcal{G}], \zeta_n) = 0 \mod X, \tag{4.9}$$

with some graphs \mathcal{G} , one can replace $\mathbf{A}[\mathcal{G}] \to \mathbf{D}[\mathcal{G}]$ and $\zeta_{\underline{n}} \to \zeta_{\underline{n}}^{\text{sv}}$ in that polynomial to obtain a relation between modular graph functions

$$\mathcal{P}(\mathbf{D}[\mathcal{G}], \zeta_n^{\mathrm{sv}}) = 0.$$
(4.10)

While this alone is a beautiful result, we would like to turn it into a formalism to actually *compute* modular graph functions. The next two subsections are dedicated to the description of the two possible methods outlined in figure 3.

4.2 Modular graph functions from *A*-cycle graph functions

Given that relations among A-cycle graph functions can be mapped to those of modular graph functions, a natural follow-up question concerns a mapping between the respective functions of τ themselves: eMZVs on the open-string side and, as will be shown below, real parts of iterated Eisenstein integrals on the closed-string side. For this purpose, we compare first-order differential operators, namely $\partial_{\tau} := \frac{\partial}{\partial \tau}$ acting on A-cycle graph functions and the Cauchy– Riemann derivative ∇ defined in eq. (2.55) acting on modular graph functions.

²⁰More precisely, we have been calculating only 12 out of the 13 A-cycle graph functions at weight six, since $\mathbf{A}\left[\overleftarrow{\mathbf{\nabla}}\right]$ is beyond the reach of our current computer implementation. Instead, we have inferred a conjectural expression for $\mathbf{A}\left[\overleftarrow{\mathbf{\nabla}}\right] \mod X$ from one of the relations in eq. (F.1). Hence, only seven out of the eight relations in appendix F could be used as a check.

4.2.1 τ -derivatives versus Cauchy–Riemann equations

From the representation of A-cycle graph functions in terms of eMZVs, their τ -derivatives can be conveniently computed using eq. (C.3). For instance, the expressions eqs. (3.15) and (3.17) straightforwardly imply that

$$2\pi i \partial_{\tau} \mathbf{A} [\boldsymbol{\smile}] = -2 \,\omega_A(0,3)$$
$$(2\pi i \partial_{\tau})^2 \mathbf{A} [\boldsymbol{\smile}] = 6 \,\mathrm{G}_4^0 \tag{4.11}$$

as well as

$$2\pi i \partial_{\tau} \mathbf{A} \Big[\triangle \Big] = 3(\omega_A(0,0,4) - \frac{1}{6} \omega_A(4)) - \zeta_2(6 \omega_A(0,0,2) - \omega_A(2))$$

$$(2\pi i \partial_{\tau})^2 \mathbf{A} \Big[\triangle \Big] = -12 \omega_A(0,5) + 12 \zeta_2 \omega_A(0,3)$$

$$(2\pi i \partial_{\tau})^3 \mathbf{A} \Big[\triangle \Big] = 60 \operatorname{G}_6^0 - 36 \zeta_2 \operatorname{G}_4^0,$$

(4.12)

see eq. (2.20) for our conventions for G_k^0 . In the previous subsection, relations between A-cycle graph functions were found to only resemble those of modular graph functions after dropping terms from the space X defined in eq. (4.4). Hence, we shall consider the simpler differential equations obeyed by a hatted version of A-cycle graph functions, in which the terms projected to zero by eq. (4.3) are omitted:

$$\hat{\mathbf{A}}[\mathcal{G}] = \mathbf{A}[\mathcal{G}] \mod X.$$
(4.13)

The simplest examples of $\hat{\mathbf{A}}[\mathcal{G}]$ can be expressed as:

$$\hat{\mathbf{A}}[\frown] = \omega_A(0,0,2), \qquad \hat{\mathbf{A}}[\triangle] = -\omega_A(0,0,0,3) + \frac{1}{6}\omega_A(0,3)
\hat{\mathbf{A}}[\square] = \omega_A(0,0,0,0,4) - \frac{1}{6}\omega_A(0,0,4)
\hat{\mathbf{A}}[\triangle] = -\omega_A(0,0,0,0,0,5) + \frac{1}{6}\omega_A(0,0,0,5) - \frac{7}{360}\omega_A(0,5)
\hat{\mathbf{A}}[\triangle] = \frac{1}{2}\omega_A(0,0,2)^2 - \frac{1}{2}\omega_A(0,0,0,0,4) - \omega_A(0,0,0,2,2).$$
(4.14)

Writing the analogue of eq. (4.12) for $\hat{\mathbf{A}}[\mathcal{G}]$, the Eisenstein series \mathbf{G}_4^0 in the last line is no longer existent. Considering other simple graphs, one finds for instance

$$(2\pi i\partial_{\tau})^{2} \hat{\mathbf{A}} \Biggl[\longleftrightarrow \Biggr] = 6 \operatorname{G}_{4}^{0}, \qquad (2\pi i\partial_{\tau})^{3} \hat{\mathbf{A}} \Biggl[\bigtriangleup \Biggr] = 60 \operatorname{G}_{6}^{0}$$

$$(2\pi i\partial_{\tau})^{4} \hat{\mathbf{A}} \Biggl[\bigsqcup] = 840 \operatorname{G}_{8}^{0}, \qquad (2\pi i\partial_{\tau})^{5} \hat{\mathbf{A}} \Biggl[\bigtriangleup \Biggr] = 15120 \operatorname{G}_{10}^{0},$$

$$(4.15)$$

which intriguingly resemble the following instances of eq. (2.56):

$$(\pi\nabla)^{2} \mathbf{D}[\boldsymbol{\frown}] = 6 \,(\mathrm{Im}(\tau))^{4} \,\mathrm{G}_{4} \,, \qquad (\pi\nabla)^{3} \,\mathbf{D}[\boldsymbol{\bigtriangleup}] = 60 \,(\mathrm{Im}(\tau))^{6} \,\mathrm{G}_{6} \qquad (4.16)$$
$$(\pi\nabla)^{4} \,\mathbf{D}[\boldsymbol{\Box}] = 840 \,(\mathrm{Im}(\tau))^{8} \,\mathrm{G}_{8} \,, \qquad (\pi\nabla)^{5} \,\mathbf{D}[\boldsymbol{\bigtriangleup}] = 15120 \,(\mathrm{Im}(\tau))^{10} \,\mathrm{G}_{10} \,.$$

A similar correspondence can be established for graphs with more than one loop: For instance, the expression for $\hat{\mathbf{A}}[\Delta]$ in eq. (4.14) yields

$$(2\pi i\partial_{\tau})^{3} \hat{\mathbf{A}}\left[\bigtriangleup\right] = 12 \operatorname{G}_{4}^{0} \omega_{A}(0,3) - 108 \omega_{A}(0,7) - 72\zeta_{4} \omega_{A}(0,3)$$

$$= -6 \operatorname{G}_{4}^{0} 2\pi i\partial_{\tau} \hat{\mathbf{A}}\left[\smile\right] + \frac{9}{10} (2\pi i\partial_{\tau})^{3} \hat{\mathbf{A}}\left[\bigsqcup\right] \mod X,$$

$$(4.17)$$

which resembles the differential equation (2.57) among modular graph functions

$$(\pi\nabla)^{3} \mathbf{D}\left[\bigtriangleup\right] = -6(\operatorname{Im}(\tau))^{4} \operatorname{G}_{4} \pi\nabla \mathbf{D}\left[\circlearrowright\right] + \frac{9}{10}(\pi\nabla)^{3} \mathbf{D}\left[\bigsqcup\right].$$
(4.18)

In passing to the second line of eq. (4.17), we have identified $2\pi i \partial_{\tau} \hat{\mathbf{A}} \bigg[\longleftrightarrow \bigg] = -2 \omega_A(0,3)$ as well as $(2\pi i \partial_{\tau})^3 \hat{\mathbf{A}} \bigg[\bigsqcup] = -120 \omega_A(0,7)$ and dropped $-72\zeta_4 \omega_A(0,3)$ as it is contained in the space X defined in eq. (4.4). In a similar way, discarding²¹ terms from X in the third τ -derivative of $\mathbf{A} \bigg[\bigsqcup]$ gives rise to an open-string counterpart of eq. (2.58).

We infer the following general conjecture from the above examples: Suppose that A-cycle graph functions associated with some graphs \mathcal{G} satisfy the differential equation

$$\mathcal{Q}(2\pi i \partial_{\tau}, \mathbf{G}_{2k}^{0}, \mathbf{A}[\mathcal{G}]) = 0 \mod X, \qquad (4.19)$$

with some polynomial \mathcal{Q} in $\mathrm{G}_{2k}^0(2\pi i\partial_{\tau})^n \mathbf{A}[\mathcal{G}]$ where $k, n \geq 0$. Then, one can coherently replace $\mathbf{A}[\mathcal{G}] \to \mathbf{D}[\mathcal{G}]$ as well as $2\pi i\partial_{\tau} \to \pi \nabla$ and $\mathrm{G}_{2k}^0 \to (\mathrm{Im}(\tau))^{2k} \mathrm{G}_{2k}$ in that polynomial and obtain a Cauchy–Riemann equation among modular graph functions

$$\mathcal{Q}(\pi\nabla, (\operatorname{Im}(\tau))^{2k} \operatorname{G}_{2k}, \mathbf{D}[\mathcal{G}]) = 0.$$
(4.20)

This procedure has been used at weight w = 5, 6 to derive conjectural Cauchy–Riemann differential equations for modular graph functions from A-cycle graph functions and thus constitutes an alternative way compared to the graphical manipulations of refs. [7,11]. Our method has been checked to either reproduce the Cauchy–Riemann equations in the above reference or to yield expressions for modular graph functions that satisfy the Laplace equations in subsection 2.3.3 as discussed in the following section.

4.2.2 Integrating Cauchy–Riemann equations

We shall now describe techniques to convert Cauchy–Riemann equations derived via eqs. (4.19) and (4.20) into explicit representations of modular graph functions. The idea is to solve the differential equations in terms of iterated Eisenstein integrals eq. (2.19) along with integer powers of $\text{Im}(\tau)$ and to fix the integration constants via modular invariance and reality of $\mathbf{D}[\mathcal{G}]$. However, these constraints do not fix the last integration constant which amounts to adding MZVs of the appropriate weight to the modular graph function under investigations. This shortcoming can be fixed either by numerical evaluation or by employing the alternative method described in subsection 4.3.

In case of one-loop graphs, eq. (2.56) can be integrated to yield the representation eq. (2.33) of non-holomorphic Eisenstein series E_k up to integration constants and antiholomorphic iterated

²¹Of course, we will as well discard terms like $G_{2k}^0 \zeta_2 \omega_A(\underline{n})$ containing a factor from X.

Eisenstein integrals. The case k = 2 in eq. (2.56) reads

$$(\pi \nabla)^2 \mathbf{E}_2 = 6(\mathrm{Im}(\tau))^4 \mathbf{G}_4$$
(4.21)

which – upon integration in τ – yields

$$\pi \nabla E_2 = \frac{2y^3}{45} + c_1 \zeta_3 + 24y^2 \mathcal{E}_0(4) + 12y \mathcal{E}_0(4,0) + 3\mathcal{E}_0(4,0,0) + c_2 \overline{\mathcal{E}_0(4,0,0)}$$
(4.22)

with rational constants c_1, c_2 and $y = \pi \operatorname{Im}(\tau)$. Then, a further integration gives rise to

$$E_2 = \frac{y^2}{45} - c_1 \frac{\zeta_3}{y} - 6 \mathcal{E}_0(4,0) + c_3 \overline{\mathcal{E}_0(4,0)} - \frac{3}{y} \mathcal{E}_0(4,0,0) - \frac{c_2}{y} \overline{\mathcal{E}_0(4,0,0)}$$
(4.23)

with another rational constant c_3 . While performing the above integrations, we have used that Cauchy–Riemann derivatives act via

$$\pi \nabla(y^n) = n \, y^{n+1} \,, \qquad \pi \nabla(\mathcal{E}_0(k_1, k_2, \dots, k_r)) = \frac{4y^2}{(2\pi i)^{k_r}} \, \mathbf{G}_{k_r}^0 \, \mathcal{E}_0(k_1, k_2, \dots, k_{r-1}) \,, \qquad (4.24)$$

and the integration constants $c_i \in \mathbb{Q}$ have been introduced following two selection rules:

- (i) Let $\mathbf{D}[\mathcal{G}_w]$ denote a modular graph function of weight w, then the admissible integration constants in $(\pi \nabla)^n \mathbf{D}[\mathcal{G}_w]$ without any accompanying $\overline{\mathcal{E}_0(\underline{k})}$ are rational combinations of single-valued MZVs of weight w+n.
- (ii) Whenever $(\pi \nabla)^n \mathbf{D}[\mathcal{G}_w]$ contains a term $\zeta_{\underline{m}}^{\mathrm{sv}} \mathcal{E}_0(\underline{k})$, then rational multiples of its complex conjugate $\zeta_{\underline{m}}^{\mathrm{sv}} \overline{\mathcal{E}_0(\underline{k})}$ have to be included in the integration constant.

Note that, as a consequence of (i), there is no rational multiple of ζ_2 in eq. (4.23).

The rational constants $c_i \in \mathbb{Q}$ in eq. (4.23) can be fixed by imposing reality $\mathbf{D}[\mathcal{G}_w] = \overline{\mathbf{D}[\mathcal{G}_w]}$ and modular invariance: Reality requires the coefficients of $\mathcal{E}_0(4,0)$ and $\overline{\mathcal{E}_0(4,0)}$ as well as $\mathcal{E}_0(4,0,0)$ and $\overline{\mathcal{E}_0(4,0,0)}$ to match, yielding $c_2 = 3$ and $c_3 = -6$. Then, the modular transformation eqs. (3.52) and (3.53) of $\mathcal{E}_0(4,0), \mathcal{E}_0(4,0,0)$ and their complex conjugates introduce ζ_3 in a way such that eq. (4.23) can only be modular invariant for $c_1 = -1$. Hence, we arrive at

$$E_2 = \frac{y^2}{45} + \frac{\zeta_3}{y} - 12 \operatorname{Re}[\mathcal{E}_0(4,0)] - \frac{6}{y} \operatorname{Re}[\mathcal{E}_0(4,0,0)], \qquad (4.25)$$

which agrees with eq. (2.33). However, the criterion based on modular invariance still leaves the freedom to add single-valued MZVs to $\mathbf{D}[\mathcal{G}_w]$ which do not exist in the case at hand with w = 2. When applying the above integration procedure to obtain the expressions

$$E_{3} = \frac{2y^{3}}{945} + \frac{3\zeta_{5}}{4y^{2}} - 120 \operatorname{Re}[\mathcal{E}_{0}(6,0,0)] - \frac{180}{y} \operatorname{Re}[\mathcal{E}_{0}(6,0,0,0)] - \frac{90}{y^{2}} \operatorname{Re}[\mathcal{E}_{0}(6,0,0,0,0)] - (4.26)$$

$$E_{4} = \frac{y^{4}}{4725} + \frac{5\zeta_{7}}{8y^{3}} - 1680 \operatorname{Re}[\mathcal{E}_{0}(8,0,0,0)] - \frac{5040}{y} \operatorname{Re}[\mathcal{E}_{0}(8,0,0,0,0,0)] - \frac{6300}{y^{2}} \operatorname{Re}[\mathcal{E}_{0}(8,0,0,0,0,0,0)] - \frac{3150}{y^{3}} \operatorname{Re}[\mathcal{E}_{0}(8,0,0,0,0,0,0)] - (4.27)$$

at weight w = 3, 4, the absence of ζ_3 in E_3 must be checked either by numerical evaluation or by the methods of section 4.3. Note that the task of integrating Cauchy–Riemann equations is completely analogous to computing modular transformations of iterated Eisenstein integrals from their differential equations, see section 3.3.4. In particular, the differential operator $\sim \tau^2 \partial_{\tau}$ for recursive computations of *B*-cycle graph functions in eqs. (3.49) to (3.51) can be mapped to the Cauchy–Riemann derivative eq. (2.55) by replacing $\tau^2 \partial_{\tau} \rightarrow (\text{Im } \tau)^2 \partial_{\tau}$. This is another reason to expect strong parallels between *B*-cycle graph functions and modular graph functions.

4.2.3 Simplifying Cauchy–Riemann equations for multi-loop graphs

When applying the integration procedure of the previous subsection to modular graph functions corresponding to graphs with more than one loop, it is useful to disentangle iterated Eisenstein integrals with different types of entries. For instance, the simplest irreducible two-loop modular graph function $\mathbf{D}[\bigtriangleup]$ will comprise two kinds of iterated Eisenstein integrals involving either two instances of G_4^0 or a single integration kernel G_8^0 . Any appearance of G_8^0 in modular graph functions at weight four can be captured via E_4 , so it is convenient to study the linear combination

$$\mathbf{E}_{2,2} := \mathbf{D}\left[\bigtriangleup\right] - \frac{9}{10} \mathbf{E}_4 \tag{4.28}$$

for which the Cauchy–Riemann equation (4.18) simplifies to

$$(\pi \nabla)^3 \mathbf{E}_{2,2} = -6 \operatorname{Im}(\tau)^4 \mathbf{G}_4 \, \pi \nabla \mathbf{E}_2 \, . \tag{4.29}$$

Then, starting from the representation (4.25) of E_2 , integration of eq. (4.29) yields depth-two iterated Eisenstein integrals with two entries of G_4^0 . This observation motivates us to define the depth of a modular graph function to be the minimum depth of the iterated Eisenstein integrals required to represent it, see section 2.2.2. Hence, the object $E_{2,2}$ in eq. (4.28) is our simplest example of a modular graph function of depth two.

Similarly, Cauchy–Riemann equations at higher weight (which can be extracted from refs. [7, 11] and which we obtained from employing the correspondence in eqs. (4.19) and (4.20)) simplify when considering the following combinations:

$$\mathbf{E}_{2,3} = \mathbf{D}\left[\Box\right] - \frac{43}{35}\mathbf{E}_5 \tag{4.30}$$

$$\mathbf{E}_{3,3} = 3\mathbf{D}\left[\textcircled{\odot}\right] + \mathbf{D}\left[\textcircled{\odot}\right] - \frac{15}{14}\mathbf{E}_{6} \tag{4.31}$$

$$\mathbf{E}_{3,3}' = \mathbf{D}\left[\bigoplus\right] + \frac{17}{60} \mathbf{D}\left[\bigoplus\right] - \frac{39}{140} \mathbf{E}_{6} \tag{4.32}$$

$$E_{2,4} = 9 \mathbf{D} \left[\bigoplus \right] + 3 \mathbf{D} \left[\bigoplus \right] + \mathbf{D} \left[\bigoplus \right] - 13 E_6$$
(4.33)

$$E_{2,2,2} = -\mathbf{D}\left[\bigoplus\right] + \frac{232}{45} \mathbf{D}\left[\bigoplus\right] + \frac{292}{15} \mathbf{D}\left[\bigoplus\right] + \frac{2}{5} \mathbf{D}\left[\bigoplus\right] + 2E_3^2 + E_2E_4 - \frac{466}{45}E_6.$$
(4.34)

The above combinations can be thought of as higher-depth generalizations of non-holomorphic Eisenstein series. The benefit of the subtractions of E_k in eq. (4.30) to eq. (4.34) becomes apparent²² in

$$(\pi\nabla)^3 E_{2,3} = -2(\pi\nabla E_2)(\pi\nabla)^2 E_3 - 4\operatorname{Im}(\tau)^4 G_4 \,\pi\nabla E_3$$
(4.35)

$$(\pi\nabla)^5 E_{3,3} = 180 \operatorname{Im}(\tau)^6 G_6(\pi\nabla)^2 E_3$$
(4.36)

²²Note that these subtractions also simplify the respective Laplace equations, e.g. we have $(\Delta - 2) E_{2,2} = -E_2^2$ instead of eq. (2.52).

$$(\pi \nabla)^4 \mathcal{E}'_{3,3} = -12 \operatorname{Im}(\tau)^6 \mathcal{G}_6(\pi \nabla) \mathcal{E}_3$$
(4.37)

$$(\pi\nabla)^3 E_{2,4} = -27 \operatorname{Im}(\tau)^4 G_4(\pi\nabla) E_4 + R_{2,4}$$
(4.38)

$$\pi \nabla \mathbf{R}_{2,4} = -81 \,\mathrm{Im}(\tau)^4 \,\mathrm{G}_4(\pi \nabla)^2 \mathbf{E}_4 - 27(\pi \nabla) \mathbf{E}_2(\pi \nabla)^3 \mathbf{E}_4 \tag{4.39}$$

from which we can anticipate all of $E_{2,3}$, $E_{3,3}$, $E'_{3,3}$ and $E_{2,4}$ to be of depth two. Finally, modular graph functions at weight six contain one independent depth-three representative satisfying

$$(\pi\nabla)^3 \operatorname{E}_{2,2,2} = -12 \operatorname{Im}(\tau)^4 \operatorname{G}_4 \pi \nabla \operatorname{E}_{2,2} + (\pi\nabla \operatorname{E}_2)^3.$$
(4.40)

For all terms $\nabla^n \mathbf{E}_{\underline{k}}$ on the right hand side of the above Cauchy–Riemann equations, a representation in terms of iterated Eisenstein integrals \mathcal{E}_0 can be found in appendix G.1. We will now proceed to solving eq. (4.29) and eqs. (4.35) to (4.40) using the method in subsection 4.2.2.

4.2.4 Explicit solutions to Cauchy–Riemann equations at higher depth

For the simplest modular graph function of depth two, $E_{2,2}$, the differential equation eq. (4.29) can be integrated to yield

$$E_{2,2} = -\frac{y^4}{20250} + \frac{y\zeta_3}{45} + \frac{5\zeta_5}{12y} - \frac{\zeta_3^2}{4y^2} - \left(\frac{2y}{15} - \frac{3\zeta_3}{y^2}\right) \operatorname{Re}[\mathcal{E}_0(4,0,0)] - \frac{9}{2y^2} \left(\operatorname{Re}[\mathcal{E}_0(4,0,0)]^2 + \operatorname{Im}[\mathcal{E}_0(4,0,0)]^2\right) - 72 \operatorname{Re}[\mathcal{E}_0(4,4,0,0)] - \frac{1}{5} \operatorname{Re}[\mathcal{E}_0(4,0,0,0)] - \frac{36 \operatorname{Re}[\mathcal{E}_0(4,0,4,0,0)]}{y} - \frac{108 \operatorname{Re}[\mathcal{E}_0(4,4,0,0,0)]}{y} - \frac{\operatorname{Re}[\mathcal{E}_0(4,0,0,0,0)]}{10y}$$
(4.41)
 $- \frac{9 \operatorname{Re}[\mathcal{E}_0(4,0,0,4,0,0)]}{y^2} - \frac{27 \operatorname{Re}[\mathcal{E}_0(4,0,4,0,0,0)]}{y^2} - \frac{54 \operatorname{Re}[\mathcal{E}_0(4,4,0,0,0,0)]}{y^2}$,

see eq. (G.1) for a convenient representation of the factor ∇E_2 therein. Unlike the expression for E_k , eq. (4.41) contains products of holomorphic and antiholomorphic iterated Eisenstein integrals, for example in

$$\operatorname{Re}[\mathcal{E}_{0}(4,0,0)]^{2} = \frac{1}{4}\mathcal{E}_{0}(4,0,0)^{2} + \frac{1}{2}\mathcal{E}_{0}(4,0,0)\overline{\mathcal{E}_{0}(4,0,0)} + \frac{1}{4}\overline{\mathcal{E}_{0}(4,0,0)}^{2}$$
(4.42)

and $\text{Im}[\mathcal{E}_0(4,0,0)]^2$. The latter can be eliminated from eq. (4.41) by taking the real part of $\mathcal{E}_0(4,0,0)^2$ and taking the shuffle relation

$$\mathcal{E}_0(4,0,0)^2 = 2 \,\mathcal{E}_0(4,0,0,4,0,0) + 6 \,\mathcal{E}_0(4,0,4,0,0,0) + 12 \,\mathcal{E}_0(4,4,0,0,0,0) , \qquad (4.43)$$

into account. This manipulation turns out to cancel all iterated Eisenstein integrals of length six from eq. (4.41):

$$E_{2,2} = -\frac{y^4}{20250} + \frac{y\zeta_3}{45} + \frac{5\zeta_5}{12y} - \frac{\zeta_3^2}{4y^2} - \left(\frac{2y}{15} - \frac{3\zeta_3}{y^2}\right) \operatorname{Re}[\mathcal{E}_0(4,0,0)] - \frac{9\operatorname{Re}[\mathcal{E}_0(4,0,0)]^2}{y^2} - 72\operatorname{Re}[\mathcal{E}_0(4,4,0,0)] - \frac{1}{5}\operatorname{Re}[\mathcal{E}_0(4,0,0,0)] - \frac{36\operatorname{Re}[\mathcal{E}_0(4,0,4,0,0)]}{y} - \frac{108\operatorname{Re}[\mathcal{E}_0(4,4,0,0,0)]}{y} - \frac{\operatorname{Re}[\mathcal{E}_0(4,0,0,0,0)]}{10y}.$$
(4.44)

The coefficients of $\frac{\zeta_5}{y}$ and $\frac{\zeta_3^2}{y^2}$ in eqs. (4.41) and (4.44) appear as integration constants in intermediate steps and can by fixed by imposing modular invariance²³ of eq. (4.44). We have checked the resulting expression for $\mathbf{D}[\bigtriangleup]$ to satisfy the Laplace eigenvalue equation (2.52), and its coefficient of $q^1 \overline{q}^0$ has been verified to agree with the results of ref. [3].

Similarly, the Cauchy–Riemann equation (4.35) for the depth-two modular graph function eq. (4.30) at weight five can be integrated to yield

$$\begin{split} \mathbf{E}_{2,3} &= -\frac{4y^5}{297675} + \frac{2y^2\zeta_3}{945} - \frac{\zeta_5}{180} - \frac{\zeta_3\zeta_5}{2y^3} + \frac{7\zeta_7}{16y^2} - \left(\frac{4y^2}{315} - \frac{3\zeta_5}{y^3}\right) \operatorname{Re}[\mathcal{E}_0(4,0,0)] \\ &- \left(\frac{8y}{3} - \frac{60\zeta_3}{y^2}\right) \operatorname{Re}[\mathcal{E}_0(6,0,0,0)] - \left(8 - \frac{60\zeta_3}{y^3}\right) \operatorname{Re}[\mathcal{E}_0(6,0,0,0,0)] \\ &- \frac{360\operatorname{Re}[\mathcal{E}_0(4,0,0)]\operatorname{Re}[\mathcal{E}_0(6,0,0,0)]}{y^2} - \frac{360\operatorname{Re}[\mathcal{E}_0(4,0,0)]\operatorname{Re}[\mathcal{E}_0(6,0,0,0,0)]}{y^3} \\ &- 1440\operatorname{Re}[\mathcal{E}_0(4,6,0,0,0)] + \frac{\operatorname{Re}[\mathcal{E}_0(4,0,0,0,0)]}{21} - \frac{4320\operatorname{Re}[\mathcal{E}_0(4,6,0,0,0)]}{y} + \frac{\operatorname{Re}[\mathcal{E}_0(4,0,0,0,0)]}{14y} \operatorname{Re}[\mathcal{E}_0(6,0,4,0,0)] \\ &- \frac{720\operatorname{Re}[\mathcal{E}_0(4,0,6,0,0,0)]}{y} - \frac{4320\operatorname{Re}[\mathcal{E}_0(4,6,0,0,0)]}{y} + \frac{\operatorname{Re}[\mathcal{E}_0(4,0,0,0,0)]}{14y} - \frac{4320\operatorname{Re}[\mathcal{E}_0(6,0,4,0,0,0)]}{y} \\ &- \frac{720\operatorname{Re}[\mathcal{E}_0(6,0,0,4,0,0)]}{y} - \frac{2160\operatorname{Re}[\mathcal{E}_0(6,0,4,0,0,0)]}{y^2} - \frac{3600\operatorname{Re}[\mathcal{E}_0(6,4,0,0,0,0,0)]}{y^2} \\ &- \frac{10\operatorname{Re}[\mathcal{E}_0(4,0,0,0,0,0,0)]}{28y^2} - \frac{360\operatorname{Re}[\mathcal{E}_0(6,0,0,4,0,0,0)]}{y^2} - \frac{1080\operatorname{Re}[\mathcal{E}_0(6,0,0,4,0,0,0)]}{y^2} \\ &- \frac{2160\operatorname{Re}[\mathcal{E}_0(6,0,4,0,0,0,0)]}{y^2} - \frac{3600\operatorname{Re}[\mathcal{E}_0(6,4,0,0,0,0,0)]}{y^2} - \frac{5\operatorname{Re}[\mathcal{E}_0(6,0,0,0,0,0,0)]}{y^2} , \end{split}$$

see eqs. (G.2) and (G.3) for explicit expressions of ∇E_3 and $\nabla^2 E_3$. Following the strategy of simplifying $E_{2,2}$, we have eliminated the appearance of $\text{Im}[\mathcal{E}_0(4,0,0)] \text{Im}[\mathcal{E}_0(6,0,0,0)]$ and $\text{Im}[\mathcal{E}_0(4,0,0)] \text{Im}[\mathcal{E}_0(6,0,0,0,0)]$ in intermediate steps by taking the real part of appropriate shuffle relations. These manipulations also remove all iterated Eisenstein integrals of length 8 from our final expression eq. (4.45). Hence, elimination of any $\text{Im}[\mathcal{E}_0(\ldots)]$ via shuffle relations will be our guiding principle for all subsequent cases which turns out to reduce the maximum length of the iterated Eisenstein integrals appearing in a given E_k .

The coefficient of ζ_5 in $E_{2,3}$ is not fixed by modular invariance and can be inferred by comparison with the results in the literature, numerical evaluation or by the method discussed in subsection 4.3. The expression for $\mathbf{D}[\square]$ resulting from eq. (4.45) has been checked to satisfy the Laplace equation (2.53), and its coefficient of $q^1 \overline{q}^0$ agrees with the results of [3].

There are three independent modular graph functions at weight six and depth two: $E_{3,3}$, $E'_{3,3}$ as well as $E_{2,4}$ defined in eqs. (4.31) to (4.33) are a convenient choice of basis. Integrating the Cauchy–Riemann equation (4.36) for $E_{3,3}$ gives rise to

$$\begin{split} \mathbf{E}_{3,3} &= \frac{2y^6}{6251175} + \frac{y\zeta_5}{210} + \frac{\zeta_7}{16y} - \frac{7\zeta_9}{64y^3} + \frac{9\zeta_5^2}{64y^4} - \left(\frac{4y}{7} + \frac{135\zeta_5}{4y^4}\right) \operatorname{Re}[\mathcal{E}_0(6,0,0,0,0)] \\ &+ \frac{2025 \operatorname{Re}[\mathcal{E}_0(6,0,0,0,0)]^2}{y^4} + 21600 \operatorname{Re}[\mathcal{E}_0(6,6,0,0,0,0)] - \frac{20}{7} \operatorname{Re}[\mathcal{E}_0(6,0,0,0,0,0)] \\ &+ \frac{21600 \operatorname{Re}[\mathcal{E}_0(6,0,6,0,0,0,0)]}{y} + \frac{108000 \operatorname{Re}[\mathcal{E}_0(6,6,0,0,0,0,0)]}{y} \end{split}$$

 23 The modular transformations in eqs. (3.53), (3.45) and (3.46) are sufficient to check this.

$$-\frac{45 \operatorname{Re}[\mathcal{E}_{0}(6,0,0,0,0,0)]}{7y} + \frac{16200 \operatorname{Re}[\mathcal{E}_{0}(6,0,0,6,0,0,0)]}{y^{2}}$$

$$+\frac{81000 \operatorname{Re}[\mathcal{E}_{0}(6,0,6,0,0,0,0)]}{y^{2}} + \frac{243000 \operatorname{Re}[\mathcal{E}_{0}(6,6,0,0,0,0,0)]}{y^{2}}$$

$$-\frac{15 \operatorname{Re}[\mathcal{E}_{0}(6,0,0,0,0,0,0,0)]}{2y^{2}} + \frac{8100 \operatorname{Re}[\mathcal{E}_{0}(6,0,0,0,6,0,0,0)]}{y^{3}}$$

$$+\frac{40500 \operatorname{Re}[\mathcal{E}_{0}(6,0,0,6,0,0,0,0,0)]}{y^{3}} + \frac{121500 \operatorname{Re}[\mathcal{E}_{0}(6,0,6,0,0,0,0,0,0)]}{y^{3}}$$

$$+\frac{283500 \operatorname{Re}[\mathcal{E}_{0}(6,6,0,0,0,0,0,0,0)]}{y^{3}} - \frac{15 \operatorname{Re}[\mathcal{E}_{0}(6,0,0,0,0,0,0,0)]}{4y^{3}} ,$$

$$(4.46)$$

and similar expressions for $E'_{3,3}$ and $E_{2,4}$ based on eqs. (4.37) to (4.39) are provided in appendix G.2. The resulting expressions for $\mathbf{D}[\bigoplus], \mathbf{D}[\bigoplus]$ and $\mathbf{D}[\bigoplus]$ have been checked to satisfy the Laplace eigenvalue equations (2.54).

Finally, there is a single irreducible modular graph function of depth three at weight six: $E_{2,2,2}$ defined in eq. (4.34). Integrating its Cauchy–Riemann equation (4.40) (with $\nabla E_{2,2}$ spelt out in eq. (G.7)) yields

$$\begin{split} \mathrm{E}_{2,2,2} &= \frac{4y^6}{9568125} - \frac{2y^3\zeta_3}{10125} + \frac{y\zeta_5}{54} + \frac{\zeta_3^2}{90} + \frac{661\zeta_7}{1800y} - \frac{5\zeta_3\zeta_5}{12y^2} + \frac{\zeta_3^3}{6y^3} \\ &+ \left(\frac{4y^3}{3375} - \frac{2\zeta_3}{15} + \frac{5\zeta_5}{2y^2} - \frac{3\zeta_3^2}{y^3}\right) \mathrm{Re}[\mathcal{E}_0(4,0,0)] + \left(\frac{2}{5} + \frac{18\zeta_3}{y^3}\right) \mathrm{Re}[\mathcal{E}_0(4,0,0)]^2 - \frac{36 \,\mathrm{Re}[\mathcal{E}_0(4,0,0)]^3}{y^3} \\ &- 36 \left(\frac{2y}{45} - \frac{\zeta_3}{y^2} + \frac{6 \,\mathrm{Re}[\mathcal{E}_0(4,0,0)]}{y^2}\right) \mathrm{Re}\left[\mathcal{E}_0(4,0,4,0,0)\right] + 3 \,\mathcal{E}_0(4,4,0,0,0) + \frac{\mathcal{E}_0(4,0,0,0,0)}{360}\right] \\ &- 864 \,\mathrm{Re}[\mathcal{E}_0(4,4,0,4,0,0)] - 2592 \,\mathrm{Re}[\mathcal{E}_0(4,4,4,0,0,0)] - \frac{12}{5} \,\mathrm{Re}[\mathcal{E}_0(4,0,0,4,0,0)] \\ &- \frac{36}{5} \,\mathrm{Re}[\mathcal{E}_0(4,0,4,0,0,0)] - \frac{84}{5} \,\mathrm{Re}[\mathcal{E}_0(4,4,0,0,0,0)] - \frac{1}{150} \,\mathrm{Re}[\mathcal{E}_0(4,0,0,0,0,0)] \\ &- \frac{1296 \,\mathrm{Re}[\mathcal{E}_0(4,0,4,0,4,0,0)]}{y} - \frac{3888 \,\mathrm{Re}[\mathcal{E}_0(4,0,4,0,0,0)]}{y} - \frac{7776 \,\mathrm{Re}[\mathcal{E}_0(4,4,4,0,0,0,0)]}{5y} \\ &- \frac{432 \,\mathrm{Re}[\mathcal{E}_0(4,0,4,0,4,0,0)]}{y} - \frac{1296 \,\mathrm{Re}[\mathcal{E}_0(4,0,4,0,0,0,0)]}{y} - \frac{6 \,\mathrm{Re}[\mathcal{E}_0(4,0,0,4,0,0,0)]}{5y} \\ &- \frac{18 \,\mathrm{Re}[\mathcal{E}_0(4,0,0,4,0,0,0)]}{y} - \frac{42 \,\mathrm{Re}[\mathcal{E}_0(4,0,0,0,0,0,0)]}{300y}, \end{split}$$

which, together with $E_{3,3}$, $E'_{3,3}$ and $E_{2,4}$, completes the basis of weight-six modular graph functions under the relations in appendix F. For all the above expressions for modular graph functions, modular invariance has been confirmed numerically.

All the above examples confirm our conjecture that the number of loops in a graph is an upper bound for the depth of the associated modular graph function. Said upper bound is saturated for the independent modular graph functions $\mathbf{D}[\bigtriangleup], \mathbf{D}[\boxdot], \mathbf{D}[\textcircled], \mathbf{D}[$

4.2.5 Laplace equation at weight six

From their representations in terms of iterated Eisenstein integrals, we infer the following Laplace equation among modular graph functions which has not yet been spelt out in the literature:

$$(\Delta - 2) \left(\mathbf{D} \left[\bigoplus \right] - 2 \, \mathrm{E}_3^2 - \mathrm{E}_2 \, \mathrm{E}_4 \right) - \frac{14}{9} \, \mathbf{D} \left[\bigoplus \right] + \frac{16}{3} \, \mathbf{D} \left[\bigoplus \right] - 4 \, \mathbf{D} \left[\bigoplus \right] + \frac{284}{9} \, \mathrm{E}_6 + \frac{2}{3} \, \mathrm{E}_2^3 + 16 \, \mathrm{E}_3^2 + \frac{12}{5} \, \mathrm{E}_2 \, \mathrm{E}_4 - 4 \, \mathrm{E}_2 \, \mathrm{E}_{2,2} = 0 \,.$$

$$(4.48)$$

The combination $\mathbf{D}\left[\bigoplus\right] - 2 \operatorname{E}_3^2 - \operatorname{E}_2 \operatorname{E}_4$ along with the Laplacian is designed to absorb contributions $\sim \partial_\tau \operatorname{E}_p \overline{\partial_\tau \operatorname{E}_q}$ in eq. (4.48) with p+q=6. Moreover, the combination $\mathbf{D}\left[\bigoplus\right] - 2 \operatorname{E}_3^2 - \operatorname{E}_2 \operatorname{E}_4$ is selected by the formalism of ref. [11] to linearize the relations between modular graph functions²⁴, as can be verified from the second equation from below in eq. (F.1).

4.2.6 Representations of modular graph functions in terms of \mathcal{E} rather than \mathcal{E}_0 ?

While all expressions for modular graph functions or their constituents have been expressed in terms of iterated Eisenstein integrals \mathcal{E}_0 defined in eq. (2.19), we conclude this subsection with expressions for modular graph functions in terms of iterated Eisenstein integrals \mathcal{E} defined in eq. (2.18), where the constant terms $2\zeta_k$ of the integrands G_k are not subtracted. At depth one, these \mathcal{E} appear to be the more suitable language for modular graph functions than the \mathcal{E}_0 since the polynomial term $E_k \sim y^k$ in eq. (2.34) is absorbed in this way:

$$\mathbf{E}_{k} = \frac{4 \left(2k-3\right)! \zeta_{2k-1} \left(4y\right)^{1-k}}{(k-2)! \left(k-1\right)!} - 8y(2k-1)! \sum_{j=0}^{k-1} \binom{2k-2-j}{k-1} \frac{(4y)^{j-k}}{j!} \operatorname{Re}\left[\mathcal{E}(2k, \underbrace{0, \dots, 0}_{2k-2-j}; q)\right].$$
(4.49)

However, the analogous rearrangements at depth two convert eq. (4.44) into

$$E_{2,2} = \frac{\zeta_3 |T|^2}{60y} + \frac{5\zeta_5}{12y} - \frac{\zeta_3^2}{4y^2} + \frac{3\zeta_3}{y^2} \operatorname{Re}[\mathcal{E}(4,0,0)] - \frac{9\operatorname{Re}[\mathcal{E}(4,0,0)]^2}{y^2} - 72\operatorname{Re}[\mathcal{E}(4,4,0,0)] - \frac{36\operatorname{Re}[\mathcal{E}(4,0,4,0,0)]}{y} - \frac{108\operatorname{Re}[\mathcal{E}(4,4,0,0,0)]}{y} - \frac{108\operatorname{Re}[\mathcal{E}(4,4,0,0,0)]}{y} - \frac{36\operatorname{Re}[\mathcal{E}(4,0,0,0)]}{y} - \frac{36\operatorname{Re}[\mathcal{E}(4,0,0,0)$$

and introduce an explicit appearance of $\operatorname{Re} \tau$ via $|T|^2 = \pi^2 ((\operatorname{Re} \tau)^2 + (\operatorname{Im} \tau)^2)$. Similar observations have been made for $\operatorname{E}_{2,3}$ and examples at higher weight, so it is not clear if representations in terms of \mathcal{E} are preferable at generic depth.

4.3 Modular graph functions from *B*-cycle graph functions

In this section, we suggest a mapping between *B*-cycle graph functions and the corresponding modular graph functions which is based on their representations via iterated Eisenstein integrals (see subsection 3.3.5 and subsection 4.2.4, respectively).

 $^{^{24}}$ The general formalism ref. [11] assigns a so-called "primitive" version to each modular graph function which is observed to linearize all relations know up to date. We are grateful to Eric D'Hoker and Justin Kaidi for bringing the connection between primitive modular graph functions and the Laplace equation (4.48) to our attention.

4.3.1 Depth one

For illustrative purposes, we repeat the expressions

$$\begin{aligned} \mathbf{D}\left[\longleftrightarrow\right] &= \frac{y^2}{45} + \frac{\zeta_3}{y} - 12 \operatorname{Re}[\mathcal{E}_0(4,0)] - \frac{6}{y} \operatorname{Re}[\mathcal{E}_0(4,0,0)] \\ \mathbf{D}\left[\bigtriangleup\right] &= \frac{2y^3}{945} + \frac{3\zeta_5}{4y^2} - 120 \operatorname{Re}[\mathcal{E}_0(6,0,0)] - \frac{180}{y} \operatorname{Re}[\mathcal{E}_0(6,0,0,0)] - \frac{90}{y^2} \operatorname{Re}[\mathcal{E}_0(6,0,0,0)] \\ \mathbf{D}\left[\Box\right] &= \frac{y^4}{4725} + \frac{5\zeta_7}{8y^3} - 1680 \operatorname{Re}[\mathcal{E}_0(8,0,0,0)] - \frac{5040}{y} \operatorname{Re}[\mathcal{E}_0(8,0,0,0,0)] \\ &- \frac{6300}{y^2} \operatorname{Re}[\mathcal{E}_0(8,0,0,0,0,0)] - \frac{3150}{y^3} \operatorname{Re}[\mathcal{E}_0(8,0,0,0,0,0)], \end{aligned}$$

for the simplest modular graph functions which agree with the all-weight formula eq. (2.33) for non-holomorphic Eisenstein series. These closed-string expressions will be brought into correspondence with the analogous *B*-cycle graph functions eq. (3.54) modulo ζ_2 on the open-string side,

$$\mathbf{B}[\frown] = -\frac{T^2}{180} + \frac{i\zeta_3}{T} - 6\mathcal{E}_0(4,0) - \frac{6i\mathcal{E}_0(4,0,0)}{T} \mod \zeta_2
\mathbf{B}[\triangle] = \frac{iT^3}{3780} - \frac{3\zeta_5}{2T^2} - 60\mathcal{E}_0(6,0,0) - \frac{180i\mathcal{E}_0(6,0,0,0)}{T} + \frac{180\mathcal{E}_0(6,0,0,0,0)}{T^2} \mod \zeta_2
\mathbf{B}[\Box] = \frac{T^4}{75600} - \frac{5i\zeta_7}{2T^3} - 840\mathcal{E}_0(8,0,0,0) - \frac{5040i\mathcal{E}_0(8,0,0,0,0,0)}{T} + \frac{12600i\mathcal{E}_0(8,0,0,0,0,0,0,0)}{T^3} \mod \zeta_2.$$
(4.52)

As in eq. (3.57), the notion of "mod ζ_2 " refers to a representation of all the τ -dependence via Tand $\mathcal{E}_0(\underline{k})$, where all terms of the form $\zeta_2^n T^m \mathcal{E}_0(\underline{k})$ with $n \ge 1$ and $m \in \mathbb{Z}$ are suppressed. Both, iterated Eisenstein integrals and Laurent polynomials in y or T exhibit striking similarities in their coefficients: Every single term in eq. (4.52) will find a correspondent in eq. (4.51) once we replace

$$T \to 2iy, \quad \mathcal{E}_0(2k, 0, \ldots) \to 2 \operatorname{Re}[\mathcal{E}_0(2k, 0, \ldots)], \quad \zeta_{2k+1} \to 2\zeta_{2k+1}, \quad \zeta_{2k} \to 0.$$
 (4.53)

The τ dependence of the $\mathcal{E}_0(\underline{k})$ through their *q*-series eq. (2.21) is understood to be unaffected by the prescription $T \to 2iy$. The same correspondence has been verified between the depth-one modular graph functions $\mathbf{D}[\widehat{\frown}], \mathbf{D}[\widehat{\frown}]$ and their *B*-cycle counterparts $\mathbf{B}[\widehat{\frown}], \mathbf{B}[\widehat{\frown}].$

4.3.2 General form

Both the doubling of odd zeta-values in eq. (4.53) and the suppression of ζ_2 in matching *B*-cycle graph functions with non-holomorphic Eisenstein series are reminiscent of the single-valued projection of MZVs. From the above examples associated with one-loop graphs, it is tempting to study the following generalization of eq. (4.53)²⁵

esv:
$$\begin{cases} (i) & T \to 2iy \\ (ii) & \mathcal{E}_0(k_1, k_2, \dots, k_r) \to 2 \operatorname{Re}[\mathcal{E}_0(k_1, k_2, \dots, k_r)], \quad k_1 \neq 0 \\ (iii) & \zeta_{\underline{n}} \to \zeta_{\underline{n}}^{\mathrm{sv}} \end{cases}$$
(4.54)

²⁵As pointed out in appendix D.2, one can always write \mathcal{E}_0 's in terms of powers of T and \mathcal{E}_0 's with $k_1 \neq 0$.

to arbitrary MZVs and iterated Eisenstein integrals. Note that part (i) or $\tau \to 2i \operatorname{Im}(\tau)$ is in fact a special case of part (ii) since $\mathcal{E}_0(0) = 2\pi i \tau$. As before, part (i) is understood to not act on the *q*-series eq. (2.21) of $\mathcal{E}_0(k,...)$ with $k \neq 0$. Moreover, part (iii) motivates our earlier choices to occasionally display *B*-cycle graph function modulo terms sent to zero by the esv-map such as $\zeta_2 T^m \mathcal{E}_0(\underline{k})$. As the key result of this section, we conjecture that, once a *B*-cycle graph function is *suitably* expressed in terms of *T* and $\mathcal{E}_0(\underline{k})$, the esv-map in eq. (4.54) yields the corresponding modular graph function,

$$\operatorname{esv} \mathbf{B}[\mathcal{G}] = \mathbf{D}[\mathcal{G}]. \tag{4.55}$$

The notion of *suitably* expressing *B*-cycle graph function in terms of *T* and $\mathcal{E}_0(\underline{k})$ will be made more precise in the next subsection 4.3.3 using examples at depth two and three. We must introduce this notion, because the esv-map is a map on iterated integrals only if we consider them as symbols and forget about the algebraic relations among them. The reason is, that these relations would not be respected by part (*ii*) of eq. (4.54). Since open- and closed-string amplitudes comprise generating functions of the respective graph functions, eq. (4.55) immediately implies the main result of this work – the connection eq. (1.6) between the four-point open- and closed-string integrals eqs. (3.6) and (2.24).

Let us already note here that the esv-map eq. (4.54) is consistent with the truncation eq. (4.13) of A-cycle graph functions selected by the replacement in eq. (4.3): Using the result eq. (2.13), it follows that adding any term $\zeta_{\underline{n}} \omega(\underline{m})$ contained in the space X defined in section 4.1 to an A-cycle graph function will result in terms proportional to $\zeta_{\underline{n}}T^m \mathcal{E}_0(\ldots)$ in the corresponding B-cycle graph function²⁶, which are in turn annihilated by part (*iii*) of eq. (4.54). In other words, all terms contained in X will yield zero upon taking their modular transformation and applying the rules eq. (4.54) afterwards. Accordingly, the observation of subsection 4.1 that A-cycle graph functions – after omission of terms from the space X and replacing $\zeta_{\underline{n}} \to \zeta_{\underline{n}}^{\text{sv}}$ – satisfy the relations of modular graph functions, is made plausible by eq. (4.55).

4.3.3 Higher depth

We shall now discuss the representations of *B*-cycle graph functions in which the esv-map eq. (4.55) to modular graph functions is applicable. At depth two, it is instructive to compare the expression eq. (4.44) for $E_{2,2} = \mathbf{D}\left[\bigtriangleup\right] - \frac{9}{10}\mathbf{D}\left[\Box\right]$ with the analogous *B*-cycle graph function

$$\mathbf{B}\left[\bigtriangleup\right] - \frac{9}{10} \mathbf{B}\left[\Box\right] = -\frac{T^4}{324000} - \frac{iT\zeta_3}{180} + \frac{5i\zeta_5}{12T} + \frac{\zeta_3^2}{4T^2} + \left(\frac{iT}{30} - \frac{3\zeta_3}{T^2}\right) \mathcal{E}_0(4,0,0) + \frac{9\,\mathcal{E}_0(4,0,0)^2}{T^2} - 36\,\mathcal{E}_0(4,4,0,0) - \frac{1}{10}\,\mathcal{E}_0(4,0,0,0) - \frac{36i\,\mathcal{E}_0(4,0,4,0,0)}{T} - \frac{108i\,\mathcal{E}_0(4,4,0,0,0)}{T} - \frac{i\,\mathcal{E}_0(4,0,0,0,0)}{10T} \mod \zeta_2,$$
(4.56)

see eq. (3.56) for the terms $\sim \zeta_2$ suppressed by the esv-map. In the present form of eq. (4.56), the esv-map in eq. (4.54) correctly reproduces the corresponding modular graph function in eq. (4.44). However, as already anticipated in the previous section, a major shortcoming is that the esv-map in eq. (4.54) is not compatible with shuffle multiplication: Rewriting eq. (4.56) via

²⁶For instance, even though $\zeta_2 \omega_A(0,0,1,0|\tau) = -\frac{\zeta_3}{8} + \frac{3}{4} \mathcal{E}_0(4,0,0|\tau)$ appears to introduce a term proportional to ζ_3 which is preserved by the single-valued projection, the modular image $\zeta_2 \omega_A(0,0,1,0|-\frac{1}{\tau}) = \frac{i\pi^6}{720T^3} - \frac{i\pi^4}{144T} + \frac{i\pi^2 T}{720} - \frac{\pi^2 \zeta_3}{8T^2} + \frac{3\pi^2 \mathcal{E}_0(4,0,0;\tau)}{4T^2}$ and thereby the contribution to a *B*-cycle graph function is sent to zero by the esv-map eq. (4.54) term by term.

 $\mathcal{E}_0(4,0,0)^2 = 2 \mathcal{E}_0(4,0,0,4,0,0) + 6 \mathcal{E}_0(4,0,4,0,0,0) + 12 \mathcal{E}_0(4,4,0,0,0,0)$ results in a different image under the esv-map. When performing this shuffle multiplication, one could at best hope to make contact with the more cumbersome representation of $E_{2,2}$ in eq. (4.41) with spurious iterated Eisenstein integrals of length six, but it is not clear how to extend the esv-map such as to generate $\operatorname{Im}[\mathcal{E}_0(\underline{k})]$.

In the depth-two case at hand, one can still argue that the representation in eq. (4.56) is optimized with respect to the length of the iterated Eisenstein integrals and therefore particularly canonical: There is currently no $\mathcal{E}_0(\ldots)$ at length six, provided that the shuffle multiplication of $\mathcal{E}_0(4,0,0)^2$ is not performed.

At weight five, the expression eq. (4.45) for the two-loop modular graph function $E_{2,3}$ can be reached by applying eq. (4.54) to the representation eq. (3.57) of the corresponding *B*-cycle graph function. For the first non-trivial product $\mathcal{E}_0(4,0,0) \mathcal{E}_0(6,0,0,0,0)$ of iterated Eisenstein integrals in eq. (3.57), the absence of $\mathcal{E}_0(\underline{k})$ at length eight in the remaining equation suggests to not perform this shuffle multiplication. However, the other product $\mathcal{E}_0(4,0,0) \mathcal{E}_0(6,0,0,0)$ in eq. (3.57) does not admit a comparable argument to leave it inert: Said *B*-cycle graph function inevitably contains $\mathcal{E}_0(\underline{k})$ at length seven, independent on the treatment of $\mathcal{E}_0(4,0,0) \mathcal{E}_0(6,0,0,0)$.

We expect that each B-cycle graph function admits a scheme of performing selected shuffle multiplications such that the esv-map eq. (4.54) converts it to the corresponding modular graph function via eq. (4.55). It would be desirable to identify a general criterion on the representations of B-cycle graph function that are tailored to the esv-map. Inappropriate ways of performing shuffle multiplications before applying the esv-map will usually result in a breakdown of modular invariance.

We have checked that the independent modular graph functions $E_{3,3}$, $E'_{3,3}$, $E_{2,4}$ and $E_{2,2,2}$ at weight six can be obtained through the esv-map from suitable representations of the corresponding *B*-cycle graph functions. This adds a depth-three example to support the general conjecture eq. (4.55).

4.3.4 Expressing esv rules in terms of \mathcal{E}_0 versus \mathcal{E}

One very obvious question is whether one can find a formulation of the esv-map in eq. (4.54) which applies to iterated Eisenstein integrals \mathcal{E} rather than \mathcal{E}_0 . It has been noted in eq. (4.49) that the leading term in the Laurent polynomial of non-holomorphic Eisenstein series cancels when \mathcal{E}_0 are collectively traded for \mathcal{E} . Indeed, inserting $\mathcal{E}_0(4,0;\tau) = \mathcal{E}(4,0;\tau) - \frac{\pi^2 \tau^2}{360}$ and $\mathcal{E}_0(4,0,0;\tau) = \mathcal{E}(4,0,0;\tau) - \frac{i\pi^3 \tau^3}{540}$ into the *B*-cycle graph function eq. (4.52) gives rise to the analogous cancellation of the term T^2 ,

$$\mathbf{B}\left[\boldsymbol{\curvearrowleft}\right] = \frac{i\zeta_3}{T} - 6\,\mathcal{E}(4,0) - \frac{6i\,\mathcal{E}(4,0,0)}{T} \mod \zeta_2\,. \tag{4.57}$$

Given that this effect persists in one-loop graph functions of higher weight, it is conceivable that replacing the esv-rule (ii) by $\mathcal{E}(\underline{k}) \to 2 \operatorname{Re}[\mathcal{E}(\underline{k})]$ correctly reproduces all E_n from *B*-cycle graph functions in terms of iterated Eisenstein integrals \mathcal{E} .

At depth two, however, there is a discouraging example: When replacing \mathcal{E}_0 by combinations of \mathcal{E} in eq. (4.56), one arrives at a shorter expression

$$\mathbf{B}\left[\bigtriangleup\right] - \frac{9}{10} \mathbf{B}\left[\Box\right] = \frac{5i\zeta_5}{12T} + \frac{\zeta_3^2}{4T^2} - \frac{3\zeta_3 \mathcal{E}(4,0,0)}{T^2} + \frac{9\mathcal{E}(4,0,0)^2}{T^2} - 36\mathcal{E}(4,4,0,0) - \frac{36i}{T}(\mathcal{E}(4,0,4,0,0) + 3\mathcal{E}(4,4,0,0,0)) \mod \zeta_2,$$
(4.58)

which should be compared with the representation eq. (4.50) of the modular graph function $E_{2,2}$. It turns out that there is no *B*-cycle analogue of the term $E_{2,2} = \frac{\zeta_3|T|^2}{60y} + \ldots$ in eq. (4.50), which was already mentioned as a drawback of representations in terms of \mathcal{E} . We hope that this particular term in the expression for $E_{2,2}$ will shed light on a reformulation of the esv-map which respects shuffle multiplication.

4.3.5 Zero modes

We recall that (by abuse of nomenclature) the zero mode of a modular graph function (see eq. (2.36))

$$\mathbf{D}[\mathcal{G}] = \sum_{m,n=0}^{\infty} c_{m,n}^{\mathcal{G}}(y) q^m \bar{q}^n, \qquad (4.59)$$

is defined to be $\mathbf{d}[\mathcal{G}] := c_{0,0}^{\mathcal{G}}(y)$. The analogue of the zero mode for a *B*-cycle graph function

$$\mathbf{B}[\mathcal{G}] = \sum_{m=0}^{\infty} b_m^{\mathcal{G}}(T) q^m \tag{4.60}$$

will be denoted as $\mathbf{b}[\mathcal{G}] := b_0^{\mathcal{G}}(T)$ and is also referred to as a *constant term* in appendix B. The map esv is well defined on zero modes, as it does not present the problem of being dependent on the way we write $\mathbf{B}[\mathcal{G}]$ in terms of iterated Eisenstein integrals $\mathcal{E}_0(\underline{k};q) = \mathcal{O}(q)$. Hence, the conjectural formula

$$\operatorname{esv} \mathbf{b}[\mathcal{G}] = \mathbf{d}[\mathcal{G}] \tag{4.61}$$

is well defined, and it has been verified on all examples up to weight six. Moreover, in order to confirm part (*iii*) of the esv-map for an MZV of depth 3 and weight 11, where the sv-map acts in a non-trivial way, eq. (4.61) has been checked to hold for the weight seven-examples $\mathbf{b}\left[\bigtriangleup\right]$ and $\mathbf{d}\left[\bigsqcup\right]$ spelt out in eqs. (3.58) and (2.44).

Hence, we propose eq. (4.61) as a conjectural method to compute the zero modes $\mathbf{d}[\mathcal{G}]$ of modular graph functions. Expressions for the constant terms $\mathbf{b}[\mathcal{G}]$ of *B*-cycle graph functions can be calculated using the methods described in appendix B. There is no conceptual bottleneck to addressing graphs of arbitrary complexity in this way, though the amount of data in intermediate steps of the calculations in appendix B imposes practical limitations²⁷ for weights larger than six. As pointed out in subsection 3.3, the analytic computation of $\mathbf{b}[\mathcal{G}]$ bypasses the necessity to numerically determine the multiple modular values arising in the modular transformations of iterated Eisenstein integrals described in subsection 3.3.3 as well as the integration constants in the method outlined in subsection 3.3.4.

5 Non-planar A-cycle graph function

As will be demonstrated in this section, the graphical organization of open-string α' -expansions is not tied to planar one-loop amplitudes. Even for non-abelian open-string states, the α' expansions of the non-planar open-string amplitudes can be conveniently expressed via mild generalizations of A-cycle graph functions which we will call *non-planar A-cycle graph functions*. As shown in ref. [25], non-planar α' -expansions are composed of A-cycle eMZVs, and one-particle

²⁷The expression for $\mathbf{b}\begin{bmatrix} \boldsymbol{\dot{\omega}} \end{bmatrix}$ in eq. (3.58) has been obtained by modular transformation of the *A*-cycle graph function $\mathbf{A}\begin{bmatrix} \boldsymbol{\dot{\omega}} \end{bmatrix}$, involving numerical evaluations of multiple modular values, see section 3.3.3.

reducible graphs will be shown to again decouple once one employs a suitable choice of the Green function. Most surprisingly, planar and non-planar A-cycle graph functions turn out to be indistinguishable under the esv-map eq. (4.54), i.e. under the correspondence eq. (1.5) between open-string graph functions and modular graph functions for the closed string. As will be detailed in section 5.3, this gives rise to expect that planar open-string amplitudes carry the complete information on the closed string, without any need for non-planar input.

5.1 Non-planar open-string integrals

Non-planar one-loop amplitudes of both abelian and non-abelian open-string states comprise the integrals [76]

$$M_{12|34}^{\text{open}}(s_{ij}|\tau) := \int_0^1 \mathrm{d}z_2 \int_0^1 \mathrm{d}z_3 \int_0^1 \mathrm{d}z_4 \, \exp\left(s_{12}P_{12} + s_{34}P_{34} + \sum_{i=1}^2 \sum_{j=3}^4 s_{ij}Q_{ij}\right) \tag{5.1}$$

$$M_{123|4}^{\text{open}}(s_{ij}|\tau) := \int_0^1 \mathrm{d}z_2 \int_0^1 \mathrm{d}z_3 \int_0^1 \mathrm{d}z_4 \, \exp\left(\sum_{i< j}^3 s_{ij} P_{ij} + \sum_{j=1}^3 s_{j4} Q_{j4}\right) \,, \tag{5.2}$$

which remain to be integrated over the modular parameter $\tau \in i\mathbb{R}$ of the respective cylinder worldsheet. The subscripts 12|34 and 123|4 refer to the distribution of the open-string states over the two boundaries of the cylinder. When performing the τ -integral for $M_{12|34}^{\text{open}}(s_{ij}|\tau)$ an additional factor of $q^{\alpha' k_1 \cdot k_2/2}$ needs to be taken into account²⁸. Given that none of the worldsheet boundaries in eqs. (5.1) and (5.2) comprises more than three punctures, the integrals are universal to both abelian and non-abelian open-string states. However, generic non-planar integrals at $n \geq 5$ points will necessitate a distinction between abelian and non-abelian states.

In contrast to the integrals eqs. (3.6) and (3.7) from the planar abelian sector, eqs. (5.1) and (5.2) contain a second species of propagators: Q_{ij} . This propagator connects punctures on different boundaries of the cylinder, and its representation following from P_{ij} in eq. (3.8) reads

$$Q_{1j} := \omega \begin{pmatrix} 1, & 0 \\ \tau/2, & 0 \end{pmatrix} - \Gamma \begin{pmatrix} 1 \\ \tau/2 \\ ; & z_j \end{pmatrix}, \qquad Q_{ij} := \omega \begin{pmatrix} 1, & 0 \\ \tau/2, & 0 \end{pmatrix} - \Gamma \begin{pmatrix} 1 \\ z_j + \tau/2 \\ ; & z_i \end{pmatrix} - \Gamma \begin{pmatrix} 1 \\ \tau/2 \\ ; & z_j \end{pmatrix}.$$
(5.3)

The planar and non-planar propagators P_{ij} and Q_{ij} given by eqs. (3.8) and (5.3) are related to the Green functions employed in section 4 of ref. [25] through a shift by $\omega_A(1,0)$. Momentum conservation $\sum_{i < j} s_{ij} = 0$ again guarantees that the Green functions of the reference and the present expressions for P_{ij} and Q_{ij} yield the same integrand in eqs. (5.1) and (5.2). As a key benefit of the representations eqs. (3.8) and (5.3) of P_{ij} and Q_{ij} , they satisfy eq. (3.9) and

$$\int_0^1 \mathrm{d}z_i \ Q_{ij} = 0 \,, \tag{5.4}$$

which furnish a suitable starting point to again organize the α' -expansion of eqs. (5.1) and (5.2) in terms of one-particle irreducible graphs. As elaborated on in ref. [25], the non-planar propagator introduces twisted eMZVs in intermediate steps,

$$\omega \begin{pmatrix} n_1, n_2, \dots, n_r \\ b_1, b_2, \dots, b_r \end{pmatrix} := \Gamma \begin{pmatrix} n_r & n_{r-1} & \dots & n_1 \\ b_r & b_{r-1} & \dots & b_1 \end{pmatrix}; 1 \end{pmatrix},$$
(5.5)

²⁸The integral $I_{12|34}$ expanded in ref. [25] is related to eq. (5.1) via $I_{12|34} = q^{\alpha' k_1 \cdot k_2/2} M_{12|34}^{\text{open}}$.



Figure 4: In the framework of non-planar A-cycle graph functions, the two types of propagators Q_{ij} and P_{ij} are represented by edges (drawn as solid lines) which do and do not cross the dashed line, respectively. Said dashed line tracks the distribution of the punctures in non-planar open-string amplitudes over two cylinder boundaries. The situations for the non-planar integrals $M_{12|34}^{\text{open}}$ and $M_{123|4}^{\text{open}}$ are depicted in the left and right panel, respectively.

which drop out from the final expressions such as²⁹

$$M_{12|34}^{\text{open}}(s_{ij}|\tau) = 1 + s_{12}^2 \left(\frac{7\zeta_2}{6} + 2\omega_A(0,0,2)\right) - 2s_{13}s_{23}\left(\frac{\zeta_2}{3} + \omega_A(0,0,2)\right)$$
(5.6)

$$-4\zeta_2 \,\omega_A(0,1,0,0)\,s_{12}^3 + s_{12}s_{13}s_{23}\Big(\frac{\zeta_3}{2} - \frac{5}{3}\,\omega_A(0,3,0,0) - 4\zeta_2\,\omega_A(0,1,0,0)\Big) + \mathcal{O}(\alpha'^4)$$

$$M_{123|4}^{\text{open}}(s_{ij}|\tau) = 1 + (s_{12}^2 + s_{12}s_{23} + s_{23}^2) \left(\frac{7\zeta_2}{6} + 2\omega_A(0,0,2)\right) + s_{12}s_{23}s_{13} \left(\frac{\zeta_3}{2} + 4\zeta_2\omega_A(0,1,0,0) - \frac{5}{3}\omega_A(0,3,0,0)\right) + \mathcal{O}(\alpha'^4),$$
(5.7)

also see [77] for the linear orders in s_{ij} . Moreover, all-order expressions for the $\tau \to i\infty$ limit of eq. (5.1) and eq. (5.2) have been given in [77] and [78], respectively.

5.2 Examples of non-planar A-cycle graph functions

The definition of planar A-cycle graph functions in section 3.1 has a natural extension to the non-planar setup. Monomials in P_{ij} and Q_{ij} from the expansion of the integrand in eqs. (5.1) and (5.2) are translated into graphs according to the following rules: Each integration variable in the open-string measure eq. (3.7) is again represented by a vertex, and the two kinds of propagators P_{ij} and Q_{ij} in eqs. (3.8) and (5.3) between vertices *i* and *j* are visualized by two types of undirected edges.

A convenient way to track the two types of edges stems from the distribution of the punctures in $M_{12|34}^{\text{open}}(s_{ij}|\tau)$ and $M_{123|4}^{\text{open}}(s_{ij}|\tau)$ into two sets according to the vertical-bar notation. These two sets correspond to the location of the punctures on two different boundaries of a cylindrical worldsheet, and the separation of the boundaries can be incorporated into the graphs through the dashed line in figure 4. Then, propagators Q_{ij} and P_{ij} correspond to edges that cross and do not cross the dashed line, respectively. The generalization of eq. (2.28) then reads

$$P_{ij} = {}_{i} \bullet - \bullet_{j} , \qquad Q_{ij} = {}_{i} \bullet - \bullet_{j} .$$

$$(5.8)$$

At weight two and three, for instance, one will have to evaluate the following non-planar A-cycle

²⁹The $M_{123|4}^{\text{open}}$ is defined as twice the integral $I_{123|4}$ expanded in ref. [25] in order to illustrate the parallels to the α' -expansion of $M_{12|34}^{\text{open}}$, see in particular section 5.3. Also note the relative minus sign in the definition of the Mandelstam variables in this work and ref. [25] which affects the third order in α' in the subsequent expansions.

graph functions

$$\mathbf{A}\left[\Leftrightarrow\right] := \int \mathrm{d}\mu_2^{\mathrm{open}} Q_{12}^2 , \quad \mathbf{A}\left[\Leftrightarrow\right] := \int \mathrm{d}\mu_2^{\mathrm{open}} Q_{12}^3 , \quad \mathbf{A}\left[\bigtriangleup\right] := \int \mathrm{d}\mu_3^{\mathrm{open}} P_{12} Q_{13} Q_{23} \quad (5.9)$$

in addition to their planar counterparts $\mathbf{A}[\frown]$, $\mathbf{A}[\ominus]$ and $\mathbf{A}[\triangle]$. The essential steps of their computation can be assembled from ref. [25], see in particular appendix I.1 of the reference, with the following results in terms of *untwisted* A-cycle eMZVs:

$$\mathbf{A}\left[\Leftrightarrow\right] = \omega_A(0,0,2) + \frac{\zeta_2}{3} \tag{5.10}$$

$$\mathbf{A}\left[\bigoplus \right] = \frac{\zeta_3}{2} - \frac{1}{3} \,\omega_A(0, 3, 0, 0) - 4\zeta_2 \,\omega_A(0, 1, 0, 0) \tag{5.11}$$

$$\mathbf{A}\left[\underline{\boldsymbol{\Delta}}\right] = -\frac{1}{3}\,\omega_A(0,3,0,0)\,. \tag{5.12}$$

From weight four on, certain graph topologies admit several non-planar A-cycle graph functions which correspond to different distributions of punctures over two boundaries or different numbers of Q_{ij} propagators. For instance, there are two and three non-planar analogues to $\mathbf{A}[\bigtriangleup]$ and $\mathbf{A}[\Box]$, respectively,

$$\mathbf{A}\left[\textcircled{\oplus}\right] := \int \mathrm{d}\mu_2^{\mathrm{open}} Q_{12}^4 \tag{5.13}$$

$$\mathbf{A}[\bigtriangleup] := \int d\mu_3^{\text{open}} P_{12}^2 Q_{13} Q_{23}, \quad \mathbf{A}[\bigtriangleup] := \int d\mu_3^{\text{open}} Q_{12}^2 P_{13} Q_{23}$$
(5.14)

$$\mathbf{A}\left[\begin{array}{c} \blacksquare \end{array}\right] := \int d\mu_4^{\text{open}} P_{12}Q_{23}P_{34}Q_{41}, \quad \mathbf{A}\left[\begin{array}{c} \blacksquare \end{array}\right] := \int d\mu_4^{\text{open}} P_{12}P_{23}Q_{34}Q_{41} \\ \mathbf{A}\left[\begin{array}{c} \blacksquare \end{array}\right] := \int d\mu_4^{\text{open}} Q_{12}Q_{23}Q_{34}Q_{41}. \tag{5.15}$$

5.3 Comparing α' -expansions of planar and non-planar integrals

By means of momentum conservation, the α' -expansion eqs. (5.6) and (5.7) of the non-planar integrals eqs. (5.1) and (5.2) can be recovered from the following planar and non-planar A-cycle graph functions:

$$M_{12|34}^{\text{open}}(s_{ij}|\tau) = 1 + s_{12}^{2}(\mathbf{A}[\textcircled{c}] + \mathbf{A}[\textcircled{c}]) - 2s_{13}s_{23}\mathbf{A}[\textcircled{c}] + \frac{s_{12}^{3}}{3}(\mathbf{A}[\textcircled{c}] - \mathbf{A}[\textcircled{c}]) + s_{12}s_{13}s_{23}(\mathbf{A}[\textcircled{c}] + 4\mathbf{A}[\textcircled{c}]) + \mathcal{O}(\alpha'^{4}), \quad (5.16)$$

$$M_{123|4}^{\text{open}}(s_{ij}|\tau) = 1 + (s_{12}^2 + s_{12}s_{23} + s_{23}^2)(\mathbf{A}[\frown] + \mathbf{A}[\frown]) + s_{12}s_{13}s_{23}\left(\frac{1}{2}\mathbf{A}[\frown] + \frac{1}{2}\mathbf{A}[\ominus] + \mathbf{A}[\bigtriangleup] + 3\mathbf{A}[\bigtriangleup]\right) + \mathcal{O}(\alpha'^4).$$
(5.17)

Clearly, when identifying the two boundaries and formally sending $\mathbf{A}[\textcircled] \to \mathbf{A}[\textcircled]$, $\mathbf{A}[\textcircled] \to \mathbf{A}[\textcircled]$, $\mathbf{A}[\textcircled] \to \mathbf{A}[\textcircled]$, and $\mathbf{A}[\textcircled] \to \mathbf{A}[\textcircled]$, both eq. (5.16) and eq. (5.17) reduce to the integral eq. (3.11) of the abelian planar amplitude. Nevertheless, the expressions eqs. (5.16) and (5.17) for non-planar integrals also apply to non-abelian open-string amplitudes.

It is tempting to compare the eMZV representation of non-planar A-cycle graph functions

with their planar counterparts. The above examples in eqs. (5.10) to (5.12),

$$\mathbf{A}[\boldsymbol{\hookrightarrow}] - \mathbf{A}[\boldsymbol{\Leftrightarrow}] = \frac{1}{2}\zeta_{2}$$
$$\mathbf{A}[\boldsymbol{\ominus}] - \mathbf{A}[\boldsymbol{\ominus}] = 12\zeta_{2}\,\omega_{A}(0,0,1,0)$$
$$\mathbf{A}[\boldsymbol{\bigtriangleup}] - \mathbf{A}[\boldsymbol{\bigtriangleup}] = 2\zeta_{2}\,\omega_{A}(0,0,1,0)$$
(5.18)

give rise to the following observation: to the orders considered, planar and non-planar A-cycle graph functions associated with the same graph differ by terms in the space X defined in eq. (4.4)and thus lead to identical results after applying eq. (4.3). This is furthermore supported by the weight-four example

$$\mathbf{A}\left[\textcircled{\textcircled{\leftrightarrow}}\right] = 216 \,\mathcal{E}_0(4,0,4,0) - 432 \,\mathcal{E}_0(4,4,0,0) - 3024 \,\mathcal{E}_0(8,0,0,0) \, \mathrm{mod} \,\,\zeta_2 \tag{5.19}$$

which again matches the expression for $\mathbf{A}[\bigoplus]$ in eq. (3.20) modulo terms in X and validates the observation beyond one-loop graphs. We therefore expect the matching of planar and non-planar A-cycle graph functions modulo terms in X to persist at higher weights,

$$\mathbf{A}[\mathcal{G}] = \mathbf{A}[\cdot \mathcal{G} \cdot] \mod X , \qquad (5.20)$$

where " \mathcal{G} " represents an arbitrary non-planar generalization³⁰ of the graph \mathcal{G} .

As explained in the paragraph below eq. (4.55), the modular S-transformation maps terms contained in the space X to terms sent to zero by esv in eq. (4.54). Hence, given the definition of non-planar B-cycle graph functions

$$\mathbf{B}[\cdot \mathcal{G} \cdot] := \mathbf{A}[\cdot \mathcal{G} \cdot] \big|_{\tau \to -\frac{1}{2}}, \qquad (5.21)$$

in direct analogy with the planar ones eq. (3.14), our conjecture eq. (5.20) implies that modular graph functions can also be constructed from non-planar open-string graph functions

$$\operatorname{esv} \mathbf{B}[\cdot \mathcal{G} \cdot] = \operatorname{esv} \mathbf{B}[\mathcal{G}] = \mathbf{D}[\mathcal{G}] .$$
(5.22)

At the level of their generating functions, this leads to the following conclusion: When the closedstring four-point amplitude is obtained from open-string input through the esv-projection as in eq. (1.6), then the non-planar sectors do not carry any additional information beyond the planar sector for abelian open-string states:

esv
$$M_4^{\text{open}}(s_{ij}|-\frac{1}{\tau}) = \text{esv } M_{12|34}^{\text{open}}(s_{ij}|-\frac{1}{\tau}) = \text{esv } M_{123|4}^{\text{open}}(s_{ij}|-\frac{1}{\tau})$$
. (5.23)

In other words – the esv-map identifies non-planar open-string integrals with planar abelian ones! It would be interesting to understand this in the light of monodromy relations among one-loop open-string amplitudes [79, 77].

6 Conclusions

In this work, we have identified new connections between building blocks of open- and closedstring one-loop amplitudes at the level of their α' -expansions. In view of the relation between

³⁰For instance, \mathcal{G} can be either \bigtriangleup or \bigtriangleup when \mathcal{G} is taken to be \bigtriangleup .

the respective tree-level amplitudes through the single-valued projection of MZVs, we have proposed an elliptic version of a single-valued map called "esv". The latter acts on the eMZVs in symmetrized one-loop open-string integrals and yields the corresponding integrals of the closed string. This connection between open and closed strings through the esv-map has been explicitly verified at the leading seven orders in α' and suggests to envision the following scenario in the long run: Closed-string α' -expansions at generic multiplicity and loop order might be entirely derivable from open-string data using suitable operations.

Our construction is based on a graphical organization of the α' -expansions of planar and non-planar open-string amplitudes: Convenient arrangements of the genus-one Green function cancel the contributions from one-particle reducible graphs which has already been used to simplify closed-string α' -expansions. For each one-particle irreducible graph, we have defined a meromorphic A-cycle graph function comprising eMZVs, its modular S-transformation called B-cycle graph function as well as non-planar generalizations. Representing these open-string constituents in terms of iterated Eisenstein integrals leads to a straightforward identification through the esv-map with the modular graph functions governing the closed-string α' -expansion.

Expressing modular graph functions in terms of iterated Eisenstein integrals gives rise to new results on their Fourier expansions beyond the simplest cases of non-holomorphic Eisenstein series. Furthermore, our iterated-Eisenstein-integral representations automatically manifest all relations between modular graph functions and their Laplace equations at the weights under consideration. We expect that this language is suitable to represent the general systematics of indecomposable modular graph functions and their network of Laplace eigenvalue equations.

Having applied methods from the open-string to modular graph functions on the closedstring side, it would be interesting to try the converse: The representation of modular graph functions in terms of nested lattice sums, which is immediately accessible from their definition through the genus-one Green function, should have an echo for eMZVs. In particular, tentative lattice-sum representations of A-cycle and B-cycle eMZVs are likely to offer new perspectives on their algebraic and differential relations and new insights on the esv-map.

Moreover, it would be desirable to connect the present proposal for the esv-map with the framework developed by Brown in refs. [59, 28, 29]. This would make clear whether our observations hold true for any graph or should be corrected at higher depth.

While the present results are restricted to scattering amplitudes of four external states, a natural follow-up question concerns the generalization to *n*-point one-loop amplitudes. Since the coefficients of the Kronecker–Eisenstein series capture the all-multiplicity integrands [23], the *n*-point α' -expansion for both open and closed strings is expressible in terms of iterated Eisenstein integrals and therefore accessible to the proposed esv-map. However, it remains to identify the correspondence between cyclic orderings in higher-point open-string amplitudes and the additional functions of the punctures in closed-string integrands at five and more points [62–64, 10]. The recent double-copy representation [70] of open-string integrands is expected to play a key role in this endeavor.

Relations between open- and closed-string amplitudes at higher genus should be encoded in a similar organization scheme of the integrals over the punctures. A strategic path at genus two would be to express the moduli-space integrand for the Zhang–Kawazumi invariant [80,81] and its recent generalizations to higher orders in α' [82] in terms of open-string quantities. For this purpose, higher-genus generalizations of eMZVs along with the appropriate analogues of iterated Eisenstein integrals seem to be a suitable framework.

Finally, both the α' -expansion of closed-string one-loop amplitudes and the iterated-integral description of modular graph functions have important implications for the non-perturbative S-

duality of type-IIB superstrings [83]: this duality symmetry connects amplitudes of different loop orders and incorporates their non-perturbative completion. It would be desirable to express the underlying modular invariant functions of the axion-dilaton field – non-holomorphic Eisenstein series at half-odd integer arguments [84,85] and beyond [86,87] – via esv-projected open-string quantities. This would set the stage for taking maximal advantage of S-duality to infer exact and non-perturbative results on the low-energy regime of type-II superstrings at unprecedented orders in α' .

Acknowledgments

First of all, we are very grateful to Nils Matthes for collaboration in early stages of the project. We would like to thank Francis Brown, Axel Kleinschmidt, Eric D'Hoker, Nils Matthes and Pierre Vanhove for comments on the draft and various helpful discussions. In addition, we are grateful to Eric D'Hoker and Justin Kaidi for several email exchanges. We would like to thank the Kolleg Mathematik und Physik Berlin for support in various ways and the Hausdorff Research Institute for Mathematics for hospitality while finalizing this article.

JB and OS would like to thank the Munich Institute for Astro- and Particle Physics for hospitality and providing a stimulating atmosphere during a workshop "Mathematics and Physics of Scattering Amplitudes" in August and September 2017, where a substantial part of this project was realized. This research was supported in part by the National Science Foundation under Grant No. NSF PHY17-48958, and we are grateful to the KITP Santa Barbara for providing a vibrant research environment during the workshop "Scattering Amplitudes and Beyond".

The research of OS was supported in part by Perimeter Institute for Theoretical Physics. Research at Perimeter Institute is supported by the Government of Canada through the Department of Innovation, Science and Economic Development Canada and by the Province of Ontario through the Ministry of Research, Innovation and Science.

The research of FZ was supported by the Max Planck Institute for Mathematics, by a French public grant as part of the Investissement d'avenir project, reference ANR-11-LABX-0056-LMH, LabEx LMH, and by the People Programme (Marie Curie Actions) of the European Union's Seventh Framework Programme (FP7/2007-2013) under REA grant agreement n. PCOFUND-GA-2013-609102, through the PRESTIGE programme coordinated by Campus France.

Appendix

Α	Translating between graphs and notations for modular gra	ıph
	functions	

Graph	D-notation	C-notation	Graph	D-notation	C-notation
\bigcirc	$D_2 = \mathbf{E}_2$	$C_{2,0}$	\bigcirc	D_6	$C_{1,1,1,1,1,1}$
\ominus	D_3	C _{1,1,1}	\mathbf{i}	D_{411}	$C_{2,1,1,1,1}$
Δ	$D_{111} = \mathcal{E}_3$	$C_{3,0}$	\bigtriangleup	D ₃₂₁	-
\bigcirc	D_4	$C_{1,1,1,1}$	\bigtriangleup	D_{222}	-
$\underline{\qquad}$	D_{211}	$C_{2,1,1}$		D ₃₁₁₁	$C_{3,1,1,1}$
	$D_{1111} = \mathcal{E}_4$	$C_{4,0}$		D ₂₂₁₁	-
\bigcirc	D_5	$C_{1,1,1,1,1}$		D'_{2111}	-
	D_{221}	-		D_{1111}''	$C_{2,2,1,1}$
\mathbf{i}	D_{311}	$C_{2,1,1,1}$	$\mathbf{\boxtimes}$	D_{1111}^{\times}	-
	D_{2111}	$C_{3,1,1}$		D_{21111}	$C_{4,1,1}$
	D'_{1111}	$C_{2,2,1}$		D'_{11111}	$C_{3,2,1}$
	$D_{11111} = \mathbf{E}_5$	$C_{5,0}$	\bigcirc	$D_{11,11,11}$	$C_{2,2,2}$
\mathbf{i}	D_{511}	$C_{2,1,1,1,1,1}$	\bigcirc	$D_{111111} = \mathbf{E}_6$	$C_{6,0}$

 Table 1: Different notations for modular graph functions in various publications

B Constant term of *B*-cycle eMZVs

As explained in detail in subsection 2.3 in ref. [24], the constant term of an A-cycle eMZV can be calculated using a method developed in refs. [74, 22]. In short, the construction relies on comparing the properly regulated generating series of A-cycle eMZVs, the elliptic KZB associator $A(\tau)$

$$e^{\pi i[y,x]}A(\tau) \equiv \sum_{r\geq 0} (-1)^r \sum_{n_1,n_2,\dots,n_r\geq 0} \omega_A(n_1,n_2,\dots,n_r|\tau) \operatorname{ad}_x^{n_r}(y) \dots \operatorname{ad}_x^{n_2}(y) \operatorname{ad}_x^{n_1}(y) , \quad (B.1)$$

to its asymptotic expansion as $\tau \to i\infty$ [74]

$$A(\tau) = \Phi(\tilde{y}, t) e^{2\pi i \tilde{y}} \Phi(\tilde{y}, t)^{-1} + \mathcal{O}(e^{2\pi i \tau}) .$$
 (B.2)

Taking the limit $\tau \to i\infty$ in eq. (B.1) amounts to replacing the full eMZV $\omega_A(n_1, \ldots, n_r)$ with its constant part $\omega_{A,0}(n_1, \ldots, n_r)$, which is the quantity of interest here.

Comparison between eqs. (B.1) and (B.2) is done for coefficients of words built from the letters x and y, which in turn denote generators of a complete and free algebra $\mathbb{C}\langle\langle x, y \rangle\rangle$, whose multiplication is concatenation and $\mathrm{ad}_x(y) \equiv [x, y]$. Equation (B.2) takes its concise and short form only after defining additional auxiliary letters

$$t \equiv [y, x]$$
, $\tilde{y} \equiv -\frac{\operatorname{ad}_x}{e^{2\pi i \operatorname{ad}_x} - 1}(y)$. (B.3)

Finally, Φ in eq. (B.2) denotes the Drinfeld associator [88–90]

$$\Phi(e_0, e_1) \equiv \sum_{\hat{W} \in \langle e_0, e_1 \rangle} \zeta^{\scriptscriptstyle \sqcup}(W) \cdot \hat{W} .$$
(B.4)

The sum over $\hat{W} \in \langle e_0, e_1 \rangle$ runs over all non-commutative words built from letters e_0 and e_1 . The operation \hat{a} acts on a word W by replacing letters e_0 and e_1 by 0 and 1, respectively. The notion $\zeta^{\mu}(W)$ refers to shuffle-regularized MZVs [91] which are uniquely determined from eq. (1.1), the shuffle product and the definition $\zeta^{\mu}(0) = \zeta^{\mu}(1) = 0$. Accordingly, the first couple of terms of $\Phi(e_0, e_1)$ are given by

$$\Phi(e_0, e_1) = 1 - \zeta_2[e_0, e_1] - \zeta_3[e_0 + e_1, [e_0, e_1]] + \dots$$
(B.5)

Numerous results for constant terms have been calculated and noted in ref. [24].

For *B*-cycle eMZVs, an analogous construction does exist. Considering the expansion of *B*-cycle eMZVs, this time it is not a constant term, but rather a polynomial $\omega_{B,0}(n_1, \ldots, n_r)$ in τ which comes with the term q^0 in the expansion

$$\omega_B(n_1, \dots, n_r | \tau) = \omega_{B,0}(n_1, \dots, n_r | \tau) + \sum_{k=1}^{\infty} \omega_{B,k}(n_1, \dots, n_r | \tau) q^k,$$
(B.6)

see eq. (2.12). The *B*-cycle analogue of the *A*-cycle associator in eq. (B.1) reads [22]

$$e^{\pi i[y,x]}B(\tau) \equiv \sum_{r\geq 0} (-1)^r \sum_{n_1,n_2,\dots,n_r\geq 0} \omega_B(n_1,n_2,\dots,n_r|\tau) \mathrm{ad}_x^{n_r}(y) \dots \mathrm{ad}_x^{n_2}(y) \mathrm{ad}_x^{n_1}(y) , \qquad (B.7)$$

While taking the limit $\tau \to i\infty$ in the above equation will again replace ω_B by $\omega_{B,0}$, obtaining the *B*-cycle analogue of eq. (B.2) takes a little more effort. In ref. [22], it was shown that the comparison ought to be done between the $(\tau \to i\infty)$ -limit of eq. (B.7) and

$$B(\tau) = \exp\left(-\frac{2\pi i}{\tau}e_{+}\right)e^{i\pi t}\Phi(-\tilde{y}-t,t)e^{2\pi i x}e^{2\pi i\tilde{y}\tau}\Phi^{-1}(\tilde{y},t) + \mathcal{O}(e^{2\pi i(1-\epsilon)\tau}), \quad (B.8)$$

where the introduction of an arbitrary $\epsilon > 0$ is needed to account for the suppressed terms of the form $\tau^l q^k$ with $k, l \ge 1$. The new ingredient in comparison to eq. (B.2) is the derivation e_+ , which acts on algebra generators x and y via

$$e_+(x) = 0$$
 and $e_+(y) = x$.

The term $\omega_{B,0}(n_1,\ldots,n_r|\tau)$ in the expansion eq. (B.6) of *B*-cycle eMZVs can then be obtained by equating eqs. (B.7) and (B.8) and isolating the coefficients of a given word in $\operatorname{ad}_x^{n_i}(y)$. For instance, applying this procedure to the simplest B-cycle graph function

$$\mathbf{B}\left[\bigtriangleup\right] = \frac{1}{2} \frac{\omega_B(0,0,2)^2}{\tau^2} - \frac{1}{2} \frac{\omega_B(0,0,0,0,4)}{\tau} - \frac{\omega_B(0,0,0,2,2)}{\tau} + \frac{7}{3} \zeta_2 \frac{\omega_B(0,0,2)}{\tau} - 14 \zeta_2 \frac{\omega_B(0,0,0,0,2)}{\tau^3} + \frac{301\zeta_4}{180}$$
(B.9)

involving eMZVs of depth two, we arrive at the constant term

$$\mathbf{b}\left[\bigtriangleup\right] = \frac{T^4}{113400} - \frac{T^2\zeta_2}{540} - \frac{iT\zeta_3}{180} + \frac{37\zeta_4}{180} + \frac{5i\zeta_5}{12T} - \frac{29\zeta_6}{16T^2} + \frac{\zeta_3^2}{4T^2} - \frac{9i\zeta_7}{4T^3} + \frac{7i\zeta_2\zeta_5}{T^3} - \frac{3i\zeta_3\zeta_4}{2T^3} + \frac{28\zeta_8}{3T^4} ,$$
(B.10)

and the same can be repeated at higher weight.

C Expanding S-transformed A-cycle eMZVs

This appendix is dedicated to a proof of our observation on the expansion eq. (2.13) of Stransformed A-cycle eMZVs: the coefficients of a given $(2\pi i\tau)^l q^k$ (with $l \in \mathbb{Z}$ and $k \ge 0$) in the expansion of $\omega_A(n_1, n_2, \ldots, n_r | -\frac{1}{\tau})$ around the cusp are claimed to be $\mathbb{Q}[2\pi i]$ -linear combinations of MZVs. This claim is essential in subsection 4.3, where we show that terms from the space X defined in eq. (4.4) are annihilated by the esv-map after a modular transformation.

Using eqs. (2.11) and (2.12), we can write

$$\omega_A(n_1, n_2, \dots, n_r | -\frac{1}{\tau}) = \sum_{l=1-r}^{n_1 + \dots + n_r} \tau^l \sum_{k=0}^{\infty} \tilde{b}_{k,l}(n_1, n_2, \dots, n_r) q^k , \qquad (C.1)$$

where the coefficients $\tilde{b}_{k,l}$ are $\mathbb{Q}[(2\pi i)^{\pm 1}]$ -linear combinations of MZVs and related to the coefficients in eq. (2.12) via $\tilde{b}_{k,l} := b_{k,l+r-(n_1+\cdots n_r)}$. What we need to prove is that $c_{k,l} := \tilde{b}_{k,l}/(2\pi i)^l$ are $\mathbb{Q}[2\pi i]$ -linear combinations of MZVs. The proof is divided into two parts. In appendix C.1, the setup of appendix B will be used to prove that the $c_{k,l}$ at k = 0 are $\mathbb{Q}[2\pi i]$ -linear combinations of MZVs. The proof is divided into two parts. In appendix C.1, the setup of appendix B will be used to prove that the $c_{k,l}$ at k = 0 are $\mathbb{Q}[2\pi i]$ -linear combinations of MZVs. Then, the analogous statement for $c_{k,l}$ at k > 0 will be deduced from the differential equation of eMZVs in appendix C.2, which together with the previous step implies our claim for all k.

C.1 The Laurent polynomial

For the coefficients $b_{k,l}(n_1, n_2, \ldots, n_r)$ of the *B*-cycle eMZV $\omega_B(n_1, n_2, \ldots, n_r | \tau)$ in eq. (2.12), it will now be shown that $b_{0,l}/(2\pi i)^{n_1+\cdots+n_r-r}$ is a $\mathbb{Q}[2\pi i]$ -linear combination of MZVs. After multiplication by $\tau^{n_1+\cdots+n_r-r}$, this implies the above claim at k = 0.

The *B*-cycle eMZV $\omega_B(n_1, n_2, \ldots, n_r | \tau)$ can be obtained as the coefficient of the word $\operatorname{ad}_x^{n_r}(y) \ldots \operatorname{ad}_x^{n_1}(y)$ in the *B*-elliptic associator eq. (B.7). In the degeneration eq. (B.8) of the associator where all the coefficients $b_{k>0,l}$ of the *B*-cycle eMZVs eq. (2.12) are suppressed, each instance of the letter x is accompanied by at least one power of $2\pi i$, and each letter of y introduces at most one negative power $\frac{1}{2\pi i}$. This follows from the following properties of eq. (B.8):

- The letter t in the Drinfeld associator and its inverse can be written as $t = \left[\frac{y}{2\pi i}, 2\pi i x\right]$, and the exponential $e^{i\pi t}$ adds further powers of $2\pi i$ to each $\frac{y}{2\pi i}$ and $2\pi i x$.
- The exponential $e^{2\pi i x}$ introduces precisely one factor of $2\pi i$ for each appearance of x.

• The expansion

$$\tilde{y} = -\frac{1}{2\pi i} \sum_{n\geq 0} \frac{B_n}{n!} (2\pi i)^n \mathrm{ad}_x^n(y)$$
 (C.2)

of the letter \tilde{y} in eq. (B.3) yields the word $\frac{y}{2\pi i}$ when n = 0, and words containing one instance of $\frac{y}{2\pi i}$ and n - 1 instances of $2\pi i x$ when n > 0 (so that the negative $2\pi i$ cancel out for n > 0).

Moreover, the factor of exp $\left(-\frac{2\pi i}{\tau}e_{+}\right)$ in eq. (B.8) does not alter the argument, because when it acts on $\frac{y}{2\pi i}$ it gives back $\frac{x}{\tau}$. Hence, the fact that the $\tau \to i\infty$ asymptotics of $\omega_B(n_1, n_2, \ldots, n_r | \tau)$ enter eq. (B.8) along with $n_1 + n_2 + \ldots + n_r$ letters x and r letters y implies that the coefficient $b_{0,l}$ can be written as $(2\pi i)^{n_1+n_2+\ldots+n_r-r}$ times $\mathbb{Q}[2\pi i]$ -linear combination of MZVs.

C.2 The *q*-expansion

The second part of the proof is based on Enriquez's differential equation satisfied by the generating series of A-cycle eMZVs [22]. For any fixed A-cycle eMZV, it implies that

$$2\pi i \frac{\partial}{\partial \tau} \omega_A(n_1, \dots, n_r | \tau) = n_1 \operatorname{G}_{n_1+1} \omega_A(n_2, \dots, n_r | \tau) - n_r \operatorname{G}_{n_r+1} \omega_A(n_1, \dots, n_{r-1} | \tau) + \sum_{i=2}^r \left\{ (-1)^{n_i} (n_{i-1} + n_i) \operatorname{G}_{n_{i-1}+n_i+1} \omega_A(n_1, \dots, n_{i-2}, 0, n_{i+1}, \dots, n_r | \tau) - \sum_{k=0}^{n_{i-1}+1} (n_{i-1} - k) \binom{n_i + k - 1}{k} \operatorname{G}_{n_{i-1}-k+1} \omega_A(n_1, \dots, n_{i-2}, k + n_i, n_{i+1}, \dots, n_r | \tau) + \sum_{k=0}^{n_i+1} (n_i - k) \binom{n_{i-1} + k - 1}{k} \operatorname{G}_{n_i-k+1} \omega_A(n_1, \dots, n_{i-2}, k + n_{i-1}, n_{i+1}, \dots, n_r | \tau) \right\}.$$

One can straightforwardly deduce the differential equation for the modular images (which up to powers of τ coincide with *B*-cycle MZVs, see eq. (2.11))

$$\frac{\partial}{\partial \tau} \omega_A(n_1, \dots, n_r | -\frac{1}{\tau}) = n_1 \left((2\pi i \tau)^{n_1 - 1} \hat{\mathbf{G}}_{n_1 + 1}(\tau) - \frac{\delta_{n_1, 1}}{\tau} \right) \omega_A(n_2, \dots, n_r | -\frac{1}{\tau})$$

$$- n_r \left((2\pi i \tau)^{n_r - 1} \hat{\mathbf{G}}_{n_r + 1}(\tau) - \frac{\delta_{n_r, 1}}{\tau} \right) \omega_A(n_1, \dots, n_{r-1} | -\frac{1}{\tau})$$

$$+ \sum_{i=2}^r \left\{ (-1)^{n_i} (n_{i-1} + n_i) (2\pi i \tau)^{n_{i-1} + n_i - 1} \hat{\mathbf{G}}_{n_{i-1} + n_i + 1}(\tau) \omega_A(n_1, \dots, n_{i-2}, 0, n_{i+1}, \dots, n_r | -\frac{1}{\tau}) \right.$$

$$- \sum_{k=0}^{n_{i-1} + 1} (n_{i-1} - k) \binom{n_i + k - 1}{k} (2\pi i \tau)^{n_{i-1} - k - 1} \hat{\mathbf{G}}_{n_{i-1} - k + 1}(\tau) \omega_A(n_1, \dots, n_{i-2}, k + n_i, n_{i+1}, \dots, n_r | -\frac{1}{\tau})$$

$$+ \sum_{k=0}^{n_i + 1} (n_i - k) \binom{n_{i-1} + k - 1}{k} (2\pi i \tau)^{n_i - k - 1} \hat{\mathbf{G}}_{n_i - k + 1}(\tau) \omega_A(n_1, \dots, n_{i-2}, k + n_i, n_{i+1}, \dots, n_r | -\frac{1}{\tau})$$

where the Kronecker-delta terms in the first and second line follow from the exceptional modular transformation $G_2(-\frac{1}{\tau}) = \tau^2 G_2(\tau) - 2\pi i \tau$. We are using the normalization conventions

$$\hat{\mathbf{G}}_{2k}(\tau) = \frac{\mathbf{G}_{2k}(\tau)}{(2\pi i)^{2k-1}} = \frac{2\zeta_{2k}}{(2\pi i)^{2k-1}} + \frac{4\pi i}{(2k-1)!} \sum_{m \ge 1} \frac{m^{2k-1}q^m}{1-q^m}$$
(C.5)

for the Eisenstein series in eq. (C.4) which enters the iterated Eisenstein integrals in eq. (2.18) and where each Fourier coefficient is a rational multiple of $2\pi i$.

The right hand side of eq. (C.4) involves A-cycle eMZVs of smaller length r-1 multiplied by $2\pi i(2\pi i\tau)^l q^k$ with $l \in \mathbb{Z}$ and $k \ge 0$. When r = 1 it is easy to see that $\omega_A(2m|-\frac{1}{\tau}) = -2\zeta_{2m}$ and $\omega_A(2m+1|-\frac{1}{\tau}) = 0$ with $m \ge 0$ are compatible with our claim. Hence, eq. (C.4) implies by induction in r that that the $\tilde{b}_{k,l}(n_1, n_2, \ldots, n_r)$ in eq. (C.1) with k > 0 can be written as $(2\pi i)^l$ times $\mathbb{Q}[2\pi i]$ -linear combinations of MZVs.

In view of the discussion in appendix C.1, there is no need to revisit the constant term of the Laurent polynomial $\tilde{b}_{0,l}(\underline{n})$ which is annihilated by ∂_{τ} , so this concludes the proof.

D Different flavors of iterated Eisenstein integrals

D.1 Another convention for iterated Eisenstein integrals

In this appendix we make precise how to convert the iterated Eisenstein integrals $\mathcal{E}(\underline{k})$ considered in the present work to the differently normalized iterated integrals $\gamma(\underline{k})$ considered in refs. [23,24], defined for

$$G_k(\tau) = \sum_{(m,n)\neq(0,0)} \frac{1}{(m+\tau n)^k} = 2\left(\zeta_k + \frac{(2\pi)^k}{(k-1)!} \sum_{m,n=1}^{\infty} m^{k-1} q^{mn}\right) = 2\zeta_k + G_k^0(\tau)$$
(D.1)

when $k \ge 2$ (even) and $G_0(\tau) = G_0^0(\tau) = -1$ as

$$\begin{split} \gamma(k_1, k_2, \dots, k_r; \tau) &:= \frac{1}{2\pi i} \int_{\tau}^{i\infty} \mathrm{d}\tau_r \ \mathbf{G}_{k_r}(\tau_r) \, \gamma(k_1, k_2, \dots, k_{r-1}; \tau_r) \\ &= \frac{1}{4\pi^2} \int_{0}^{q} \mathrm{dlog} \, q_r \ \mathbf{G}_{k_r}(q_r) \, \gamma(k_1, k_2, \dots, k_{r-1}; q_r) \\ &= \frac{1}{(4\pi^2)^r} \int_{0 < q_1 < q_2 < \dots < q_r < q} \mathrm{dlog} \, q_1 \ \mathbf{G}_{k_1}(q_1) \ \mathrm{dlog} \, q_2 \ \mathbf{G}_{k_2}(q_2) \dots \, \mathrm{dlog} \, q_r \ \mathbf{G}_{k_r}(q_r) \,, \\ \gamma_0(k_1, k_2, \dots, k_r; \tau) &:= \frac{1}{2\pi i} \int_{\tau}^{i\infty} \mathrm{d}\tau_r \ \mathbf{G}_{k_r}^0(\tau_r) \, \gamma_0(k_1, k_2, \dots, k_{r-1}; \tau_r) \\ &= \frac{1}{4\pi^2} \int_{0}^{q} \mathrm{dlog} \, q_r \ \mathbf{G}_{k_r}^0(q_r) \, \gamma_0(k_1, k_2, \dots, k_{r-1}; q_r) \\ &= \frac{1}{(4\pi^2)^r} \int_{0 < q_1 < q_2 < \dots < q_r < q} \mathrm{dlog} \, q_1 \ \mathbf{G}_{k_1}^0(q_1) \ \mathrm{dlog} \, q_2 \ \mathbf{G}_{k_2}^0(q_2) \dots \, \mathrm{dlog} \, q_r \ \mathbf{G}_{k_r}^0(q_r) \,. \end{split}$$
(D.3)

The conversion reads

$$\gamma(k_1, k_2, \dots, k_r; \tau) = (2\pi i)^{k_1 + k_2 + \dots + k_r - 2r} \mathcal{E}(k_1, k_2, \dots, k_r; \tau), \qquad (D.4)$$

$$\gamma_0(k_1, k_2, \dots, k_r; \tau) = (2\pi i)^{k_1 + k_2 + \dots + k_r - 2r} \mathcal{E}_0(k_1, k_2, \dots, k_r; \tau).$$
(D.5)

D.2 Conversion between \mathcal{E}_0 and \mathcal{E}

Recall the generating series

$$\mathbb{E}_{\underline{k}}(Y_0, Y_1, \dots, Y_r; \tau) := \sum_{p_0, p_1, \dots, p_r \ge 0} \frac{1}{(2\pi i)^{2p_0}} \bigg[\prod_{i=1}^r (2\pi i)^{k_i - 2p_i - 1} \bigg]$$

$$\times \mathcal{E}(0^{p_0}, k_1, 0^{p_1}, \dots, k_r, 0^{p_r}; \tau) Y_0^{p_0} Y_1^{p_1} \cdots Y_r^{p_r} .$$
(D.6)

Let us introduce a lighter notation for iterated integrals: we denote the iterated integral of the differential forms $\omega_1(t)dt, \ldots, \omega_r(t)dt$, integrated from ω_r to ω_1 along a path $\gamma \subset \mathbb{C}$ as $\int_{\gamma} \omega_1 \cdots \omega_r$. For instance, on the straight path [0, 1] we have

$$\int_{[0,1]} \omega_1 \cdots \omega_r = \int_0^1 \omega_1(t_1) \, \mathrm{d}t_1 \int_0^{t_1} \omega_2(t_2) \, \mathrm{d}t_2 \cdots \int_0^{t_{r-1}} \omega_r(t_r) \, \mathrm{d}t_r \,, \tag{D.7}$$

while on a path $[\tau, i\infty]$ in the upper half plane we have

$$\int_{[\tau,i\infty]} \omega_1 \cdots \omega_r = \int_{\tau}^{i\infty} \omega_r(t_1) \,\mathrm{d}t_1 \int_{t_1}^{i\infty} \omega_{r-1}(t_2) \,\mathrm{d}t_2 \cdots \int_{t_{r-1}}^{i\infty} \omega_1(t_r) \,\mathrm{d}t_r \,. \tag{D.8}$$

Let us then rewrite our generating series as

$$\mathbb{E}_{\underline{k}}(Y_{0}, Y_{1}, \dots, Y_{r}; \tau) = \sum_{p_{0}, p_{1}, \dots, p_{r} \ge 0} \frac{Y_{0}^{p_{0}} Y_{1}^{p_{1}} Y_{2}^{p_{2}} \cdots Y_{r}^{p_{r}}}{(2\pi i)^{p_{0}+p_{1}+p_{2}+\dots+p_{r}}} \\ \times \left[\int_{[\tau, i\infty]} \underline{G_{0} \cdots G_{0}}_{p_{0}} G_{k_{1}} \underbrace{G_{0} \cdots G_{0}}_{p_{1}} G_{k_{2}} \cdots G_{k_{r}} \underbrace{G_{0} \cdots G_{0}}_{p_{r}} \right]$$
(D.9)
$$= \int_{[\tau, i\infty]} \exp\left(\frac{t_{1}Y_{0}}{2\pi i}\right) G_{k_{1}}(t_{1}) \exp\left(\frac{(t_{2}-t_{1})Y_{1}}{2\pi i}\right) G_{k_{2}}(t_{2}) \cdots \\ \times \cdots G_{k_{r-1}}(t_{r-1}) \exp\left(\frac{(t_{r}-t_{r-1})Y_{r-1}}{2\pi i}\right) G_{k_{r}}(t_{r}) \exp\left(\frac{(\tau-t_{r})Y_{r}}{2\pi i}\right),$$

where in the last step we have used that

$$\int_{[t_i, t_j]} \underbrace{\mathbf{G}_0 \, \mathbf{G}_0 \cdots \mathbf{G}_0}_{p} = \frac{(t_i - t_j)^p}{p!} \tag{D.10}$$

and that (according to the regularization introduced in [59])

$$\int_{[t,i\infty]} \underbrace{\mathbf{G}_0 \cdots \mathbf{G}_0}_{p} = (-1)^p \int_{[0,t]} \underbrace{\mathbf{G}_0 \cdots \mathbf{G}_0}_{p} = \frac{t^p}{p!} \,. \tag{D.11}$$

Let us also introduce a modified generating series

$$\hat{\mathbb{E}}_{\underline{k}}(Y_1,\ldots,Y_r;\tau) := \sum_{p_1,\ldots,p_r \ge 0} \left[\prod_{i=1}^r (2\pi i)^{k_i - 2p_i - 1} \right] \mathcal{E}(k_1,0^{p_1},\ldots,k_r,0^{p_r};\tau) Y_1^{p_1} \cdots Y_r^{p_r}, \quad (D.12)$$

where, in comparison to eq. (D.6), the iterated Eisenstein integrals $\mathcal{E}(0,...)$ with 0 in the first entry are suppressed. It is easy to check repeating for $\hat{\mathbb{E}}_{\underline{k}}$ the same steps of (D.9) that

$$\mathbb{E}_{\underline{k}}(Y_0, Y_1, \dots, Y_r; \tau) = \exp\left(\frac{\tau Y_0}{2\pi i}\right) \hat{\mathbb{E}}_{\underline{k}}(Y_1, \dots, Y_r; \tau), \qquad (D.13)$$

which leads to the explicit formula

$$\mathcal{E}(0^{p_0}, k_1, 0^{p_1}, \dots, k_r, 0^{p_r}; \tau) = \sum_{t+s_1+\dots+s_r=p_0} (-1)^{s_1+\dots+s_r}$$
(D.14)

$$\times \left(\prod_{j=1}^r \binom{p_j+s_j}{s_j}\right) \frac{(2\pi i\tau)^t}{t!} \mathcal{E}(k_1, 0^{p_1+s_1}, \dots, k_r, 0^{p_r+s_r}; \tau).$$

Iterated Eisenstein integrals of the kind $\mathcal{E}(k_1, 0^{p_1}, \ldots, k_r, 0^{p_r}; \tau)$ can be written as the sum of $\mathcal{E}_0(k_1, 0^{p_1}, \ldots, k_r, 0^{p_r}; \tau)$ and other iterated Eisenstein integrals \mathcal{E} 's of strictly lower depth³¹, therefore one can iteratively make use of eq. (D.14) and write any \mathcal{E} in terms of the \mathcal{E}_0 's. For instance, one easily gets

$$\mathcal{E}(k_1, 0^{p_1}; \tau) = \mathcal{E}_0(k_1, 0^{p_1}; \tau) - \frac{2\zeta_{k_1}}{(2\pi i)^{k_1}} \mathcal{E}(0^{p_1+1}; \tau)$$
(D.15)

and

$$\mathcal{E}(k_1, 0^{p_1}, k_2, 0^{p_2}; \tau) = \mathcal{E}_0(k_1, 0^{p_1}, k_2, 0^{p_2}; \tau) - \frac{2\zeta_{k_2}}{(2\pi i)^{k_2}} \mathcal{E}_0(k_1, 0^{p_1 + p_2 + 1}; \tau)$$
(D.16)
$$- \frac{2\zeta_{k_1}}{(2\pi i)^{k_1}} \sum_{s+t=p_1+1} \binom{p_2 + s}{s} \frac{(-1)^s (2\pi i)^t}{t!} \Big(\mathcal{E}_0(k_2, 0^{p_2 + s}; \tau) - \frac{2\zeta_{k_2}}{(2\pi i)^{k_2}} \mathcal{E}_0(0^{p_2 + s + 1}; \tau) \Big).$$

The motivation to do this conversion comes from the fact that we know explicitly³² the *q*-expansion of any \mathcal{E}_0 , which therefore allowed us to exploit the very fast convergence of these series and verify numerically our results and conjectures to arbitrary precision.

D.3 Conversion between \mathcal{E} and \mathcal{G}

The relations eq. (3.28) between the two generating series eq. (3.27) of iterated Eisenstein integrals \mathcal{E} and \mathcal{G} can be proven by writing, as in the previous section,

$$\mathbb{E}_{\underline{k}}(Y_0, Y_1, \dots, Y_r; \tau) = \int_{[\tau, i\infty]} \exp\left(\frac{t_1 Y_0}{2\pi i}\right) \mathcal{G}_{k_1}(t_1) \exp\left(\frac{(t_2 - t_1) Y_1}{2\pi i}\right) \mathcal{G}_{k_2}(t_2) \cdots$$
(D.17)

$$\times \cdots \mathcal{G}_{k_{r-1}}(t_{r-1}) \exp\left(\frac{(t_r - t_{r-1}) Y_{r-1}}{2\pi i}\right) \mathcal{G}_{k_r}(t_r) \exp\left(\frac{(\tau - t_r) Y_r}{2\pi i}\right).$$

as well as writing the generating series $\mathbb{G}_{\underline{k}}$ in eq. (3.27) as

$$\mathbb{G}_{\underline{k}}(T_1,\ldots,T_r;\tau) = \int_{[\tau,i\infty]} \exp\left(\frac{t_1 T_1}{2\pi i}\right) \mathcal{G}_{k_1}(t_1) \exp\left(\frac{t_2 T_2}{2\pi i}\right) \mathcal{G}_{k_2}(t_2) \cdots \exp\left(\frac{t_r T_r}{2\pi i}\right) \mathcal{G}_{k_r}(t_r)$$
(D.18)

by a completely similar computation.

D.4 Examples of modular transformations

We give here one more example in depth two of a special linear combination of iterated Eisenstein integrals which can be S-transformed with our methods, as well as an example of depth three, where one finds one MZV of depth three involved:

$$\begin{split} & 3\mathcal{E}(4,6,0,0,0;-\frac{1}{\tau}) + \mathcal{E}(6,0,4,0,0;-\frac{1}{\tau}) + 3\mathcal{E}(6,4,0,0,0;-\frac{1}{\tau}) = \frac{\zeta_4}{240}\mathcal{E}(4;\tau) - \frac{\zeta_3}{6}\mathcal{E}(0,6;\tau) \\ & + \mathcal{E}(0,0,4,0,6;\tau) + 3\mathcal{E}(0,0,0,4,6;\tau) + 3\mathcal{E}(0,0,0,6,4;\tau) + \frac{i}{T}\Big(-\frac{\zeta_5}{20}\mathcal{E}(4;\tau) + \frac{\zeta_4}{240}\mathcal{E}(0,4;\tau) \\ & - \frac{\zeta_3}{2}\mathcal{E}(0,0,6;\tau) + 3\mathcal{E}(0,0,4,0,0,6;\tau) + 9\mathcal{E}(0,0,0,4,0,6;\tau) + 18\mathcal{E}(0,0,0,0,4,6;\tau) \end{split}$$

³¹Recall that this was defined as the number of non-zero entries of \mathcal{E} .

³²This is a consequence of eq. (2.21) and the fact that our regularization of iterated Eisenstein integrals yields $\mathcal{E}_0(0^p;\tau) = \mathcal{E}(0^p;\tau) = (2\pi i \tau)^p / p!$.

$$+ 3 \mathcal{E}(0,0,0,6,4;\tau) + 18 \mathcal{E}(0,0,0,0,6,4;\tau) + \frac{143\zeta_6}{241920} + \frac{1}{T^2} \left(\frac{\zeta_5}{20} \mathcal{E}(0,4;\tau) - \frac{\zeta_4}{480} \mathcal{E}(0,0,4;\tau) + \frac{3\zeta_3}{4} \mathcal{E}(0,0,0,6;\tau) - \frac{9}{2} \mathcal{E}(0,0,4,0,0,0,6;\tau) - \frac{27}{2} \mathcal{E}(0,0,0,0,4,0,0,6;\tau) - 27 \mathcal{E}(0,0,0,0,4,0,6;\tau) - 45 \mathcal{E}(0,0,0,0,0,4,6;\tau) - \frac{3}{2} \mathcal{E}(0,0,0,6,0,0,4;\tau) - 12 \mathcal{E}(0,0,0,0,6,0,4;\tau) - 45 \mathcal{E}(0,0,0,0,0,6,4;\tau) - \frac{13\zeta_4\zeta_3}{20160} + \frac{7\zeta_7}{1920} + \frac{i}{T^3} \left(\frac{\zeta_5}{40} \mathcal{E}(0,0,4;\tau) + \frac{\zeta_3}{2} \mathcal{E}(0,0,0,0,6;\tau) - 3 \mathcal{E}(0,0,4,0,0,0,6;\tau) - 9 \mathcal{E}(0,0,0,0,0,6;\tau) - 18 \mathcal{E}(0,0,0,0,6;\tau) - 30 \mathcal{E}(0,0,0,0,0,4,0,6;\tau) - 45 \mathcal{E}(0,0,0,0,0,6,6;\tau) - 3 \mathcal{E}(0,0,0,4;\tau) - 15 \mathcal{E}(0,0,0,0,6,6;\tau) - 45 \mathcal{E}(0,0,0,0,0,6;\tau) - 3 \mathcal{E}(0,0,0,0,6;\tau) - 15 \mathcal{E}(0,0,0,0,0,6;\tau) - 45 \mathcal{E}(0,0,0,0,0,6;\tau) - 3 \mathcal{E}(0,0,0,0,6;\tau) - 15 \mathcal{E}(0,0,0,0,0,6;\tau) - 45 \mathcal{E}(0,0,0,0,0,6;\tau) - 3 \mathcal{E}(0,0,0,0,6;\tau) - 15 \mathcal{E}(0,0,0,0,0,6;\tau) - 45 \mathcal{E}(0,0,0,0,0,6;\tau) - 3 \mathcal{E}(0,0,0,0,6;\tau) - 15 \mathcal{E}(0,0,0,0,0,6;\tau) - 45 \mathcal{E}(0,0,0,0,0,6;\tau) - 3 \mathcal{E}(0,0,0,0,0,6;\tau) - 15 \mathcal{E}(0,0,0,0,0,6;\tau) - 15 \mathcal{E}(0,0,0,0,0,6;\tau) - 15 \mathcal{E}(0,0,0,0,0,6;\tau) - 15 \mathcal{E}(0,0,0,0,0,6;\tau) - 45 \mathcal{E}(0,0,0,0,0,6;\tau) - 3 \mathcal{E}(0,0,0,0,6;\tau) - 15 \mathcal{E}(0,0,0,0,0,6;\tau) - 15 \mathcal{E}(0,0,0,0,0,0,6;\tau) - 15 \mathcal{E}(0,0,0,0,0,0,0,6;\tau) - 15 \mathcal{E}(0,0,0,0,0,0,0,0,0,0,0,0,0;\tau) - 15 \mathcal{E}(0,0,0,0,0,0,0,0,0,0,0,0,0,0;\tau) - 15 \mathcal{E}(0,0,0,0,0,0,0,0,0,0,0;\tau) - 15 \mathcal{E}(0,0,0,0,0,0,0,0,0,0,0;\tau) - 15 \mathcal{E}(0,0,0,0,0,0,0,0,0,0,0;\tau) - 15 \mathcal{E}(0,0,0,0,0,0,0,0,0;\tau) - 15 \mathcal{E}(0,0,0,0,0,0,0,0,0;\tau) - 15 \mathcal{E}(0,0,0,0,0,0,0,0;\tau) - 15 \mathcal{E}(0,0,0,0,0,0,0;\tau) - 15 \mathcal{E}(0,0,0,0,0,0,0;\tau) - 15 \mathcal{E}(0,0,0,0,0,0;\tau) - 15 \mathcal{E}(0,0,0,0,0,0;\tau) - 15 \mathcal{E}(0,0,0,0,0,0;\tau) - 15 \mathcal{E}(0,0,0,0,0;\tau) - 15 \mathcal{E}(0,0,0,0,0;\tau) - 15 \mathcal{E}(0,0,0,0;\tau$$

and

$$\begin{split} \mathcal{E}(6,4,4;-\frac{1}{\tau}) &= \frac{1}{8(2\pi i)^8} \bigg[\frac{533\zeta_3\zeta_8}{4050} - \frac{4\zeta_3\zeta_{3,5}}{225} - \frac{4\zeta_5\zeta_3^2}{45} - \frac{4\zeta_{3,5,3}}{225} - \frac{221\zeta_{11}}{5400} \\ &+ \bigg(\frac{16\zeta_5\zeta_3}{15} - \frac{503\zeta_8}{675} + \frac{16\zeta_{3,5}}{75} \bigg) \mathcal{E}(0^2,4;\tau) - \frac{16\zeta_5}{5} \big[\mathcal{E}(0^2,4;\tau) \big]^2 \\ &+ 80640 \, \mathcal{E}(0^7,6,4,0,4;\tau) + 69120 \, \mathcal{E}(0^6,6,0^2,4,4;\tau) + 23040 \, \mathcal{E}(0^5,6,0^3,4,4;\tau) \\ &+ 768 \, \mathcal{E}(0^4,6,0^2,4,0^2,4;\tau) + 161280 \, \mathcal{E}(0^7,6,0,4,4;\tau) + 11520 \, \mathcal{E}(0^6,6,4,0^2,4;\tau) \\ &+ 34560 \, \mathcal{E}(0^6,6,0,4,0,4;\tau) + 11520 \, \mathcal{E}(0^5,6,0^2,4,0,4;\tau) + 23040 \, \mathcal{E}(0^4,6,0^3,4,0,4;\tau) \\ &+ 4608 \, \mathcal{E}(0^4,6,0^4,4,4;\tau) + 322560 \, \mathcal{E}(0^8,6,4,4;\tau) + 3840 \, \mathcal{E}(0^5,6,0,4,0^2,4;\tau) \bigg] \tag{D.20}$$

E A-cycle graph functions at weight five

Among the six A-cycle graph functions at weight five, two examples have been spelt out in terms of eMZVs in eqs. (3.23) and (3.24), and the remaining four are given by

$$\begin{split} \mathbf{A} \bigg[\textcircled{\begin{subarray}{l} \label{eq:A} } & \mathbf{A} \bigg[\textcircled{\begin{subarray}{l} \label{eq:A} \\ & + 120\,\omega_A(0,3)\,\omega_A(0,0,0,0,2) - 312\,\omega_A(0,0,0,0,0,5) + 120\,\omega_A(0,0,0,0,0,1,4) \\ & + 5\,\omega_A(0,0,2)\zeta_3 - 12\,\omega_A(0,3)\zeta_2 - 80\,\omega_A(0,0,2)\,\omega_A(0,0,1,0)\zeta_2 + \frac{25}{9}\,\omega_A(0,0,3,0)\zeta_2 \\ & + 480\,\omega_A(0,0,0,0,0,3)\zeta_2 - 960\,\omega_A(0,0,0,0,1,2)\zeta_2 + \frac{265}{6}\zeta_2\zeta_3 \\ & + \frac{412}{3}\,\omega_A(0,0,1,0)\zeta_4 + 192\,\omega_A(0,0,0,1,0,0)\zeta_4 \\ & (E.1) \\ \mathbf{A} \bigg[\bigstar \bigg] = \frac{7\zeta_5}{80} + \frac{87}{400}\,\omega_A(0,5) + \frac{77}{20}\,\omega_A(0,0,0,5) - \frac{5}{2}\,\omega_A(0,0,2,3) + \omega_A(0,0,2)\,\omega_A(0,0,3,0) \\ & + 15\,\omega_A(0,3)\,\omega_A(0,0,0,0,2) - \frac{381}{10}\,\omega_A(0,0,0,0,5) + 15\,\omega_A(0,0,0,0,1,4) \\ & - \frac{7}{10}\,\omega_A(0,3)\zeta_2 - 6\,\omega_A(0,0,2)\,\omega_A(0,0,1,0)\zeta_2 + \frac{3}{2}\,\omega_A(0,0,3,0)\zeta_2 + 36\,\omega_A(0,0,0,0,0,3)\zeta_2 \\ & - 48\,\omega_A(0,0,0,0,1,2)\zeta_2 + 2\zeta_2\zeta_3 + \frac{47}{10}\,\omega_A(0,0,1,0)\zeta_4 + \frac{48}{5}\,\omega_A(0,0,0,1,0,0)\zeta_4 \\ & (E.2) \\ \mathbf{A} \bigg[\bigstar \bigg] = \frac{3\zeta_5}{20} + \frac{3}{100}\,\omega_A(0,5) + \frac{19}{15}\,\omega_A(0,0,0,5) - \frac{2}{3}\,\omega_A(0,0,2,3) + 4\,\omega_A(0,3)\,\omega_A(0,0,0,0,2) \end{split}$$

$$-\frac{58}{5}\omega_{A}(0,0,0,0,0,0,5) + 4\omega_{A}(0,0,0,0,1,4) - \frac{1}{3}\omega_{A}(0,3)\zeta_{2} + \frac{14}{9}\omega_{A}(0,0,3,0)\zeta_{2} + 32\omega_{A}(0,0,0,0,0,3)\zeta_{2} - 8\omega_{A}(0,0,0,0,1,2)\zeta_{2} + \frac{1}{3}\zeta_{2}\zeta_{3} - \frac{4}{5}\omega_{A}(0,0,1,0)\zeta_{4} - \frac{352}{5}\omega_{A}(0,0,0,1,0,0)\zeta_{4}$$
(E.3)
$$\mathbf{A}\left[\textcircled{\bullet}\right] = \frac{\zeta_{5}}{60} - \frac{7}{900}\omega_{A}(0,5) + \frac{1}{15}\omega_{A}(0,0,0,5) - \frac{2}{5}\omega_{A}(0,0,0,0,0,5) + \frac{1}{10}\omega_{A}(0,3)\zeta_{2} + \frac{1}{3}\omega_{A}(0,0,3,0)\zeta_{2} + 12\omega_{A}(0,0,0,0,1,2)\zeta_{2} - \frac{1}{2}\zeta_{2}\zeta_{3} - \frac{103}{15}\omega_{A}(0,0,1,0)\zeta_{4} + \frac{132}{5}\omega_{A}(0,0,0,1,0,0)\zeta_{4}.$$
(E.4)

F Relations between modular graph functions at weight six

In this appendix, we collect the complete set of relations among modular graph functions of weight six as given in [11]:

$$\begin{split} 0 &= \mathbf{D}\left[\bigoplus\right] - 15 \,\mathbf{D}\left[\bigoplus\right] \mathbf{D}\left[\bigoplus\right] + 30 \,\mathbf{D}\left[\bigoplus\right]^{3} - 10 \,\mathbf{D}\left[\bigoplus\right]^{2} - 60 \,\mathbf{D}\left[\bigoplus\right] + 720 \,\mathbf{D}\left[\bigoplus\right] \\ &+ 240 \,\mathbf{D}\left[\bigtriangleup\right] \mathbf{D}\left[\bigoplus\right] - 720 \,\mathbf{D}\left[\bigoplus\right] \mathbf{D}\left[\bigsqcup\right] - 1440 \,\mathbf{D}\left[\bigtriangleup\right]^{2} - 5280 \,\mathbf{D}\left[\bigoplus\right] \\ &+ 360 \,\mathbf{D}\left[\bigoplus\right] \mathbf{D}\left[\bigoplus\right] - 1280 \,\mathbf{D}\left[\bigoplus\right] + 3380 \,\mathbf{D}\left[\circlearrowright\right] \\ &= 2 \,\mathbf{D}\left[\bigoplus\right] + 3 \,\mathbf{D}\left[\bigoplus\right] - 9 \,\mathbf{D}\left[\bigoplus\right] \mathbf{D}\left[\bigsqcup\right] - 6 \,\mathbf{D}\left[\bigtriangleup\right]^{2} - 18 \,\mathbf{D}\left[\bigoplus\right] \\ &- 24 \,\mathbf{D}\left[\bigoplus\right] - 2 \,\mathbf{D}\left[\bigoplus\right] + 32 \,\mathbf{D}\left[\circlearrowright\right] \\ &= -3 \,\mathbf{D}\left[\bigoplus\right] + 109 \,\mathbf{D}\left[\bigoplus\right] + 408 \,\mathbf{D}\left[\bigoplus\right] + 36 \,\mathbf{D}\left[\bigoplus\right] + 18 \,\mathbf{D}\left[\bigoplus\right] \mathbf{D}\left[\bigtriangleup\right] \\ &+ 12 \,\mathbf{D}\left[\bigtriangleup\right] \mathbf{D}\left[\bigoplus\right] - 211 \,\mathbf{D}\left[\circlearrowright\right] \\ &= 3 \,\mathbf{D}\left[\bigoplus\right] - 18 \,\mathbf{D}\left[\bigoplus\right] - 58 \,\mathbf{D}\left[\bigoplus\right] - 192 \,\mathbf{D}\left[\bigoplus\right] - 3 \,\mathbf{D}\left[\bigoplus\right]^{3} + 24 \,\mathbf{D}\left[\bigtriangleup\right]^{2} \\ &+ 18 \,\mathbf{D}\left[\bigoplus\right] \mathbf{D}\left[\bigsqcup\right] + 46 \,\mathbf{D}\left[\circlearrowright\right] \\ &= 2 \,\mathbf{D}\left[\bigsqcup\right] + 18 \,\mathbf{D}\left[\bigoplus\right] - 36 \,\mathbf{D}\left[\bigoplus\right] - 69 \,\mathbf{D}\left[\bigoplus\right] - 288 \,\mathbf{D}\left[\bigoplus\right] - 6 \,\mathbf{D}\left[\bigoplus\right] \mathbf{D}\left[\bigsqcup\right] \\ \\ &- 18 \,\mathbf{D}\left[\bigoplus\right] \mathbf{D}\left[\bigsqcup\right] - 36 \,\mathbf{D}\left[\bigtriangleup\right]^{2} + 183 \,\mathbf{D}\left[\circlearrowright\right] \\ &= 3 \,\mathbf{D}\left[\bigsqcup\right] + 6 \,\mathbf{D}\left[\bigoplus\right] - 10 \,\mathbf{D}\left[\bigoplus\right] - 48 \,\mathbf{D}\left[\bigoplus\right] - 12 \,\mathbf{D}\left[\bigoplus\right] \\ \\ &= 18 \,\mathbf{D}\left[\bigoplus\right] - 9 \,\mathbf{D}\left[\bigoplus\right] - 10 \,\mathbf{D}\left[\hookrightarrow\right] \\ &= 48 \,\mathbf{D}\left[\bigsqcup\right] - 9 \,\mathbf{D}\left[\bigoplus\right] - 20 \,\mathbf{D}\left[\bigoplus\right] - 60 \,\mathbf{D}\left[\bigoplus\right] + 9 \,\mathbf{D}\left[\bigoplus\right] \mathbf{D}\left[\bigsqcup\right] \\ \\ &+ 18 \,\mathbf{D}\left[\bigtriangleup\right]^{2} - 10 \,\mathbf{D}\left[\bigvee\right] \\ \\ &= 3 \,\mathbf{D}\left[\bigsqcup\right] - \mathbf{D}\left[\bigoplus\right] - 12 \,\mathbf{D}\left[\bigoplus\right] \\ &= 3 \,\mathbf{D}\left[\bigsqcup\right] - \mathbf{D}\left[\bigoplus\right] - 12 \,\mathbf{D}\left[\bigoplus\right] \\ \end{aligned}$$

G Explicit modular graph forms and modular graph functions

In this appendix we gather explicit representations of all the modular graph forms and modular graph functions which appear in the Cauchy–Riemann equations up to weight six and have not

been spelt out in the main text.

G.1 Cauchy Riemann derivatives

In order to supplement the discussion of the Cauchy–Riemann equations in section 4.2.3, all the modular graph forms on their right hand side will be spelt out in this subsection. Starting from the expression eq. (2.33) for non-holomorphic Eisenstein series, repeated action of the Cauchy–Riemann derivative eq. (2.55) gives rise to

$$\pi \nabla E_2 = \frac{2y^3}{45} - \zeta_3 + 24y^2 \mathcal{E}_0(4) + 12y \mathcal{E}_0(4, 0) + 6 \operatorname{Re}[\mathcal{E}_0(4, 0, 0)]$$
(G.1)

$$\pi \nabla E_3 = \frac{2y^4}{315} - \frac{3\zeta_5}{2y} + 240y^2 \mathcal{E}_0(6,0) + 360y \mathcal{E}_0(6,0,0) + 180 \mathcal{E}_0(6,0,0,0)$$

$$+ 180 \operatorname{Re}[\mathcal{E}_0(6,0,0,0)] + \frac{180 \operatorname{Re}[\mathcal{E}_0(6,0,0,0,0)]}{180 \operatorname{Re}[\mathcal{E}_0(6,0,0,0,0)]}$$
(G.2)

$$(\pi \nabla)^{2} E_{3} = \frac{8y^{5}}{315} + \frac{3\zeta_{5}}{2} - 960y^{4} \mathcal{E}_{0}(6) - 960y^{3} \mathcal{E}_{0}(6,0) - 720y^{2} \mathcal{E}_{0}(6,0,0) - 360y \mathcal{E}_{0}(6,0,0,0) - 180 \operatorname{Re}[\mathcal{E}_{0}(6,0,0,0,0)]$$
(G.3)

$$\pi \nabla E_4 = \frac{4y^5}{4725} - \frac{15\zeta_7}{8y^2} + 3360y^2 \mathcal{E}_0(8, 0^2) + 10080y \mathcal{E}_0(8, 0^3) + 12600 \mathcal{E}_0(8, 0^4)$$
(G.4)
+
$$\frac{6300 \mathcal{E}_0(8, 0^5)}{y} + 5040 \operatorname{Re}[\mathcal{E}_0(8, 0^4)] + \frac{12600 \operatorname{Re}[\mathcal{E}_0(8, 0^5)]}{y} + \frac{9450 \operatorname{Re}[\mathcal{E}_0(8, 0^6)]}{y^2}$$

$$(\pi\nabla)^{2} \mathcal{E}_{4} = \frac{4y^{6}}{945} + \frac{15\zeta_{7}}{4y} - 13440y^{4} \mathcal{E}_{0}(8,0) - 33600y^{3} \mathcal{E}_{0}(8,0^{2}) - 50400y^{2} \mathcal{E}_{0}(8,0^{3})$$
(G.5)

$$-50400y \mathcal{E}_{0}(8,0^{4}) - 25200 \mathcal{E}_{0}(8,0^{5}) - 12600 \operatorname{Re}[\mathcal{E}_{0}(8,0^{5})] - \frac{18900 \operatorname{Re}[\mathcal{E}_{0}(8,0^{5})]}{y} (\pi\nabla)^{3} \operatorname{E}_{4} = \frac{8y^{7}}{315} - \frac{15\zeta_{7}}{4} + 53760y^{6} \mathcal{E}_{0}(8) + 80640y^{5} \mathcal{E}_{0}(8,0) + 100800y^{4} \mathcal{E}_{0}(8,0^{2}) + 100800y^{3} \mathcal{E}_{0}(8,0^{3}) + 75600y^{2} \mathcal{E}_{0}(8,0^{4}) + 37800y \mathcal{E}_{0}(8,0^{5}) + 18900 \operatorname{Re}[\mathcal{E}_{0}(8,0^{6})].$$
(G.6)

At depth two, the Cauchy–Riemann derivative of the modular graph function $E_{2,2}$ in eq. (4.44) is given by

$$\begin{aligned} \pi \nabla \mathcal{E}_{2,2} &= -\frac{2y^5}{10125} + \frac{y^2 \zeta_3}{45} - \frac{5\zeta_5}{12} + \frac{\zeta_3^2}{2y} + \left(\frac{4y^3}{15} - 6\zeta_3\right) \mathcal{E}_0(4,0) - \left(\frac{2y^2}{15} + \frac{6\zeta_3}{y}\right) \operatorname{Re}[\mathcal{E}_0(4,0,0)] \\ &+ \frac{2y^2}{5} \mathcal{E}_0(4,0,0) + 36y \, \mathcal{E}_0(4,0)^2 + 36 \, \mathcal{E}_0(4,0) \operatorname{Re}[\mathcal{E}_0(4,0,0)] + \frac{18 \operatorname{Re}[\mathcal{E}_0(4,0,0)]^2}{y} \quad (G.7) \\ &+ 144y^2 \, \mathcal{E}_0(4,4,0) + 72y \left(\mathcal{E}_0(4,4,0,0) + \frac{1}{360} \, \mathcal{E}_0(4,0,0,0)\right) \\ &+ 36 \operatorname{Re}[\mathcal{E}_0(4,0,4,0,0) + 3 \, \mathcal{E}_0(4,4,0,0) + \frac{1}{360} \, \mathcal{E}_0(4,0,0,0,0)] \,, \end{aligned}$$

which enters on the right hand side of the differential equation (4.40) for $E_{2,2,2}$.

G.2 Modular graph functions at weight six

Using the method in section 4.2.2, the Cauchy–Riemann equations (4.39) to (4.39) give rise to the following expressions for $E'_{3,3}$ and $E_{2,4}$, respectively.

$$\mathbf{E}_{3,3}' = -\frac{y^6}{18753525} + \frac{y\zeta_5}{630} + \frac{3\zeta_7}{160y} - \frac{7\zeta_9}{480y^3} - \left(\frac{4y^2}{105} - \frac{9\zeta_5}{y^3}\right) \operatorname{Re}[\mathcal{E}_0(6,0,0,0)]$$

$$-\frac{540 \operatorname{Re}[\mathcal{E}_{0}(6,0,0,0)]^{2}}{y^{2}} - \frac{1080 \operatorname{Re}[\mathcal{E}_{0}(6,0,0,0)] \operatorname{Re}[\mathcal{E}_{0}(6,0,0,0,0)]}{y^{3}}$$

$$-\frac{4y}{21} \operatorname{Re}[\mathcal{E}_{0}(6,0,0,0,0)] - 1440 \operatorname{Re}[\mathcal{E}_{0}(6,0,6,0,0,0)] - \frac{11}{21} \operatorname{Re}[\mathcal{E}_{0}(6,0,0,0,0,0)]$$

$$-\frac{2160 \operatorname{Re}[\mathcal{E}_{0}(6,0,0,6,0,0,0)]}{y} - \frac{4320 \operatorname{Re}[\mathcal{E}_{0}(6,0,6,0,0,0,0)]}{y}$$

$$-\frac{13 \operatorname{Re}[\mathcal{E}_{0}(6,0,0,0,0,0,0)]}{14y} - \frac{1080 \operatorname{Re}[\mathcal{E}_{0}(6,0,0,0,6,0,0,0)]}{y^{2}}$$

$$-\frac{2160 \operatorname{Re}[\mathcal{E}_{0}(6,0,0,6,0,0,0,0)]}{y^{2}} + \frac{1080 \operatorname{Re}[\mathcal{E}_{0}(6,6,0,0,0,0,0,0)]}{y^{2}}$$

$$-\frac{\operatorname{Re}[\mathcal{E}_{0}(6,0,0,0,0,0,0,0)]}{y^{2}} + \frac{1080 \operatorname{Re}[\mathcal{E}_{0}(6,0,0,0,0,0,0,0)]}{y^{3}}$$

$$+\frac{5400 \operatorname{Re}[\mathcal{E}_{0}(6,0,0,6,0,0,0,0,0)]}{y^{3}} + \frac{16200 \operatorname{Re}[\mathcal{E}_{0}(6,0,6,0,0,0,0,0,0)]}{y^{3}}$$

$$+\frac{37800 \operatorname{Re}[\mathcal{E}_{0}(6,6,0,0,0,0,0,0,0,0)]}{y^{3}} - \frac{\operatorname{Re}[\mathcal{E}_{0}(6,0,0,0,0,0,0,0,0,0)]}{2y^{3}}$$

$$\begin{split} \mathbf{E}_{2,4} &= -\frac{y^6}{70875} + \frac{y^3\zeta_3}{525} + \frac{3\zeta_7}{40y} + \frac{25\zeta_9}{8y^3} - \frac{135\zeta_3\zeta_7}{32y^4} - \left(\frac{2y^3}{175} - \frac{405\zeta_7}{16y^4}\right) \operatorname{Re}[\mathcal{E}_0(4,0,0)] \\ &- \left(504y - \frac{11340\zeta_3}{y^2}\right) \operatorname{Re}[\mathcal{E}_0(8,0,0,0,0)] - \left(2520 - \frac{28350\zeta_3}{y^3}\right) \operatorname{Re}[\mathcal{E}_0(8,0,0,0,0,0)] \\ &- \left(\frac{5670}{y} - \frac{42525\zeta_3}{2y^4}\right) \operatorname{Re}[\mathcal{E}_0(8,0,0,0,0,0)] - \frac{68040 \operatorname{Re}[\mathcal{E}_0(4,0,0)] \operatorname{Re}[\mathcal{E}_0(8,0,0,0,0)]}{y^2} \\ &- \frac{170100 \operatorname{Re}[\mathcal{E}_0(4,0,0)] \operatorname{Re}[\mathcal{E}_0(8,0,0,0,0,0)] - \frac{127575 \operatorname{Re}[\mathcal{E}_0(4,0,0)] \operatorname{Re}[\mathcal{E}_0(8,0,0,0,0,0,0)]}{y^4} \\ &- 45360 \operatorname{Re}[\mathcal{E}_0(8,0,0,4,0,0)] - 136080 \operatorname{Re}[\mathcal{E}_0(8,0,4,0,0,0)] - 272160 \operatorname{Re}[\mathcal{E}_0(8,0,0,0,0,0)] \\ &- 272160 \operatorname{Re}[\mathcal{E}_0(4,8,0,0,0,0)] - \frac{3}{20} \operatorname{Re}[\mathcal{E}_0(4,0,0,0,0,0)] - \frac{136080 \operatorname{Re}[\mathcal{E}_0(8,0^3,4,0^2)]}{y} \\ &- \frac{1360800 \operatorname{Re}[\mathcal{E}_0(4,8,0^5)]}{y} - \frac{9\operatorname{Re}[\mathcal{E}_0(4,0^6)]}{20y} - \frac{136080 \operatorname{Re}[\mathcal{E}_0(8,0^3,4,0^2)]}{y} \\ &- \frac{408240 \operatorname{Re}[\mathcal{E}_0(8,0^2,4,0^3)]}{y^2} - \frac{816480 \operatorname{Re}[\mathcal{E}_0(8,0,4,0^4)]}{y^2} - \frac{1360800 \operatorname{Re}[\mathcal{E}_0(8,0^4,4,0^5)]}{y} \\ &- \frac{340200 \operatorname{Re}[\mathcal{E}_0(4,0,8,0^5)]}{y^2} - \frac{2551500 \operatorname{Re}[\mathcal{E}_0(4,8,0^6)]}{y^2} - \frac{9\operatorname{Re}[\mathcal{E}_0(4,0^7)]}{16y^2} \\ &- \frac{170100 \operatorname{Re}[\mathcal{E}_0(8,0,4,4,0^2)]}{y^2} - \frac{510300 \operatorname{Re}[\mathcal{E}_0(8,0^3,4,0^3)]}{y^3} - \frac{6615 \operatorname{Re}[\mathcal{E}_0(8,0^5,4,0^2)]}{32y^3} \\ &- \frac{255150 \operatorname{Re}[\mathcal{E}_0(4,0,8,0^6)]}{y^3} - \frac{1786050 \operatorname{Re}[\mathcal{E}_0(8,0^3,4,0^4)]}{y^3} - \frac{85050 \operatorname{Re}[\mathcal{E}_0(8,0^2,4,0^5)]}{y^3} \\ &- \frac{255150 \operatorname{Re}[\mathcal{E}_0(8,0,4,4,0^3)]}{y^3} - \frac{510300 \operatorname{Re}[\mathcal{E}_0(8,0^3,4,0^4)]}{y^3} - \frac{85050 \operatorname{Re}[\mathcal{E}_0(8,0^2,4,0^5)]}{y^3} \\ &- \frac{255150 \operatorname{Re}[\mathcal{E}_0(8,0,4,4,0^3)]}{y^3} - \frac{1786050 \operatorname{Re}[\mathcal{E}_0(8,0^3,4,0^4)]}{y^3} - \frac{85050 \operatorname{Re}[\mathcal{E}_0(8,0^2,4,0^5)]}{y^3} \\ &- \frac{1275750 \operatorname{Re}[\mathcal{E}_0(8,0,4,4,0^3)]}{y^3} - \frac{1786050 \operatorname{Re}[\mathcal{E}_0(8,0^3,4,0^4)]}{y^3} - \frac{85050 \operatorname{Re}[\mathcal{E}_0(8,0^2,4,0^5)]}{y^3} \\ &- \frac{1275750 \operatorname{Re}[\mathcal{E}_0(8,0,4,0^6)]}{y^3} - \frac{1786050 \operatorname{Re}[\mathcal{E}_0(8,0^3,4,0^4)]}{y^3} - \frac{6615 \operatorname{Re}[\mathcal{E}_0(8,0^8,0)]}{2y^3} \\ \end{array}\right)$$

The contributing expressions for ∇E_3 and $\nabla^{j=1,2,3}E_4$ can be found in the previous subsection.

References

- M. B. Green and P. Vanhove, "The Low-energy expansion of the one loop type II superstring amplitude", Phys.Rev. D61, 104011 (2000), hep-th/9910056.
- [2] M. B. Green, J. G. Russo and P. Vanhove, "Low energy expansion of the four-particle genus-one amplitude in type II superstring theory", JHEP 0802, 020 (2008), arxiv:0801.0322.
- [3] E. D'Hoker, M. B. Green and P. Vanhove, "On the modular structure of the genus-one Type II superstring low energy expansion", JHEP 1508, 041 (2015), arxiv:1502.06698.
- [4] E. D'Hoker, M. B. Green and P. Vanhove, "Proof of a modular relation between 1-, 2- and 3-loop Feynman diagrams on a torus", arxiv:1509.00363.
- [5] A. Basu, "Poisson equation for the Mercedes diagram in string theory at genus one", Class. Quant. Grav. 33, 055005 (2016), arxiv:1511.07455.
- [6] E. D'Hoker, M. B. Green, O. Guerdogan and P. Vanhove, "Modular Graph Functions", Commun. Num. Theor. Phys. 11, 165 (2017), arxiv:1512.06779.
- [7] E. D'Hoker and M. B. Green, "Identities between Modular Graph Forms", arxiv:1603.00839.
- [8] A. Basu, "Poisson equation for the three loop ladder diagram in string theory at genus one", Int. J. Mod. Phys. A31, 1650169 (2016), arxiv:1606.02203.
- [9] A. Basu, "Proving relations between modular graph functions", Class. Quant. Grav. 33, 235011 (2016), arxiv:1606.07084.
- [10] A. Basu, "Simplifying the one loop five graviton amplitude in type IIB string theory", Int. J. Mod. Phys. A32, 1750074 (2017), arxiv:1608.02056.
- [11] E. D'Hoker and J. Kaidi, "Hierarchy of Modular Graph Identities", JHEP 1611, 051 (2016), arxiv:1608.04393.
- [12] A. Kleinschmidt and V. Verschinin, "Tetrahedral modular graph functions", JHEP 1709, 155 (2017), arxiv:1706.01889.
- [13] E. D'Hoker and W. Duke, "Fourier series of modular graph functions", arxiv:1708.07998.
- [14] D. Zagier, "The Bloch-Wigner-Ramakrishnan polylogarithm function", Math. Ann. 286, 613 (1990).
- [15] A. Levin, "Elliptic polylogarithms: An analytic theory", Compositio Mathematica 106, 267 (1997).
- [16] F. Brown and A. Levin, "Multiple elliptic polylogarithms", arxiv:1110.6917v2.
- [17] O. Schlotterer and S. Stieberger, "Motivic Multiple Zeta Values and Superstring Amplitudes", J.Phys. A46, 475401 (2013), arxiv:1205.1516.
- [18] S. Stieberger, "Closed superstring amplitudes, single-valued multiple zeta values and the Deligne associator", J.Phys. A47, 155401 (2014), arxiv:1310.3259.
- [19] S. Stieberger and T. R. Taylor, "Closed String Amplitudes as Single-Valued Open String Amplitudes", Nucl.Phys. B881, 269 (2014), arxiv:1401.1218.
- [20] O. Schnetz, "Graphical functions and single-valued multiple polylogarithms", Commun. Num. Theor. Phys. 08, 589 (2014), arxiv:1302.6445.
- [21] F. Brown, "Single-valued Motivic Periods and Multiple Zeta Values", SIGMA 2, e25 (2014), arxiv:1309.5309.
- [22] B. Enriquez, "Analogues elliptiques des nombres multizétas",
 Bull. Soc. Math. France 144, 395 (2016), arxiv:1301.3042.
- [23] J. Broedel, C. R. Mafra, N. Matthes and O. Schlotterer, "Elliptic multiple zeta values and one-loop superstring amplitudes", JHEP 1507, 112 (2015), arxiv:1412.5535.
- [24] J. Broedel, N. Matthes and O. Schlotterer, "Relations between elliptic multiple zeta values and a special derivation algebra", J. Phys. A49, 155203 (2016), arxiv:1507.02254.

- [25] J. Broedel, N. Matthes, G. Richter and O. Schlotterer, "Twisted elliptic multiple zeta values and non-planar one-loop open-string amplitudes", arxiv:1704.03449.
- [26] M. B. Green and J. H. Schwarz, "Infinity Cancellations in SO(32) Superstring Theory", Phys.Lett. B151, 21 (1985).
- [27] F. Brown, "Notes on motivic periods", Communications in Number Theory and Physics 11, 557 (2015), arxiv:1512.06410.
- [28] F. Brown, "A class of non-holomorphic modular forms I", Res. Math. Sci. 5, 5:7 (2018).
- [29] F. Brown, "A class of non-holomorphic modular forms II : equivariant iterated Eisenstein integrals", arxiv:1708.03354.
- [30] C. R. Mafra, O. Schlotterer and S. Stieberger, "Complete N-Point Superstring Disk Amplitude I. Pure Spinor Computation", Nucl. Phys. B873, 419 (2013), arxiv:1106.2645.
- [31] J. Broedel, O. Schlotterer and S. Stieberger, "Polylogarithms, Multiple Zeta Values and Superstring Amplitudes", Fortsch. Phys. 61, 812 (2013), arxiv:1304.7267.
- [32] J. Polchinski, "String theory. Vol. 1: An introduction to the bosonic string", Cambridge University Press (2007).
- [33] T. Terasoma, "Selberg Integrals and Multiple Zeta Values", Compositio Mathematica 133, 1 (2002).
- [34] F. Brown, "Multiple zeta values and periods of moduli spaces $\mathfrak{M}_{0,n}$ ", Ann. Sci. Éc. Norm. Supér. (4) 42, 371 (2009), math/0606419.
- [35] S. Stieberger, "Constraints on Tree-Level Higher Order Gravitational Couplings in Superstring Theory", Phys.Rev.Lett. 106, 111601 (2011), arxiv:0910.0180.
- [36] L. A. Barreiro and R. Medina, "5-field terms in the open superstring effective action", JHEP 0503, 055 (2005), hep-th/0503182.
- [37] D. Oprisa and S. Stieberger, "Six gluon open superstring disk amplitude, multiple hypergeometric series and Euler-Zagier sums", hep-th/0509042.
- [38] S. Stieberger and T. R. Taylor, "Multi-Gluon Scattering in Open Superstring Theory", Phys.Rev. D74, 126007 (2006), hep-th/0609175.
- [39] S. Stieberger and T. R. Taylor, "Supersymmetry Relations and MHV Amplitudes in Superstring Theory", Nucl. Phys. B793, 83 (2008), arxiv:0708.0574.
- [40] R. H. Boels, "On the field theory expansion of superstring five point amplitudes", Nucl. Phys. B876, 215 (2013), arxiv:1304.7918.
- [41] G. Puhlfuerst and S. Stieberger, "Differential Equations, Associators, and Recurrences for Amplitudes", Nucl. Phys. B902, 186 (2016), arxiv:1507.01582.
- [42] J. Broedel, O. Schlotterer, S. Stieberger and T. Terasoma, "All order α'-expansion of superstring trees from the Drinfeld associator", Phys.Rev. D89, 066014 (2014), arxiv:1304.7304.
- [43] C. R. Mafra and O. Schlotterer, "Non-abelian Z-theory: Berends-Giele recursion for the α' -expansion of disk integrals", JHEP 1701, 031 (2017), arxiv:1609.07078.
- [44] J. Broedel, O. Schlotterer and S. Stieberger, http://mzv.mpp.mpg.de.
- [45] H. Kawai, D. C. Lewellen and S. H. H. Tye, "A Relation Between Tree Amplitudes of Closed and Open Strings", Nucl. Phys. B269, 1 (1986).
- [46] S. Bloch and P. Vanhove, "The elliptic dilogarithm for the sunset graph", J. Number Theory 148, 328 (2015), arxiv:1309.5865.
- [47] S. Bloch, M. Kerr and P. Vanhove, "A Feynman integral via higher normal functions", Compos. Math. 151, 2329 (2015), arxiv:1406.2664.
- [48] L. Adams, C. Bogner and S. Weinzierl, "The two-loop sunrise graph in two space-time dimensions with arbitrary masses in terms of elliptic dilogarithms", J.Math.Phys. 55, 102301 (2014), arxiv:1405.5640.

- [49] L. Adams, C. Bogner and S. Weinzierl, "The two-loop sunrise integral around four space-time dimensions and generalisations of the Clausen and Glaisher functions towards the elliptic case", J.Math.Phys. 56, 072303 (2015), arxiv:1504.03255.
- [50] L. Adams and S. Weinzierl, "Feynman integrals and iterated integrals of modular forms", arxiv:1704.08895.
- [51] E. Remiddi and L. Tancredi, "Schouten identities for Feynman graph amplitudes; The Master Integrals for the two-loop massive sunrise graph", Nucl. Phys. B880, 343 (2014), arxiv:1311.3342.
- [52] E. Remiddi and L. Tancredi, "Differential equations and dispersion relations for Feynman amplitudes. The two-loop massive sunrise and the kite integral", Nucl. Phys. B907, 400 (2016), arxiv:1602.01481.
- [53] E. Remiddi and L. Tancredi, "An Elliptic Generalization of Multiple Polylogarithms", Nucl. Phys. B925, 212 (2017), arxiv:1709.03622.
- [54] J. Broedel, C. Duhr, F. Dulat and L. Tancredi, "Elliptic polylogarithms and iterated integrals on elliptic curves I: general formalism", arxiv:1712.07089.
- [55] J. Broedel, C. Duhr, F. Dulat and L. Tancredi, "Elliptic polylogarithms and iterated integrals on elliptic curves II: an application to the sunrise integral", arxiv:1712.07095.
- [56] B. Enriquez, "Elliptic associators", Selecta Math. (N.S.) 20, 491 (2014).
- [57] N. Matthes, "Elliptic multiple zeta values", PhD thesis, Universität Hamburg, 2016.
- [58] F. Zerbini, "Elliptic multiple zeta values, modular graph functions and superstring scattering amplitudes", PhD Thesis, to appear 20, F. Zerbini (2017).
- [59] F. Brown, "Multiple modular values and the relative completion of the fundamental group of $M_{1,1}$ ", arxiv:1407.5167v4.
- [60] N. Matthes, "On the algebraic structure of iterated integrals of quasimodular forms", Algebra & Number Theory 11-9, 2113 (2017), arxiv:1708.04561.
- [61] M. B. Green, J. H. Schwarz and L. Brink, "N=4 Yang-Mills and N=8 Supergravity as Limits of String Theories", Nucl. Phys. B198, 474 (1982).
- [62] D. M. Richards, "The One-Loop Five-Graviton Amplitude and the Effective Action", JHEP 0810, 042 (2008), arxiv:0807.2421.
- [63] M. B. Green, C. R. Mafra and O. Schlotterer, "Multiparticle one-loop amplitudes and S-duality in closed superstring theory", JHEP 1310, 188 (2013), arxiv:1307.3534.
- [64] C. R. Mafra and O. Schlotterer, "One-loop superstring six-point amplitudes and anomalies in pure spinor superspace", JHEP 1604, 148 (2016), arxiv:1603.04790.
- [65] F. Zerbini, "Single-valued multiple zeta values in genus 1 superstring amplitudes", Commun. Num. Theor. Phys. 10, 703 (2016), arxiv:1512.05689.
- [66] D. Zagier, "Evaluation of S(m, n). Appendix to [2]".
- [67] D. Zagier, "Genus 0 and genus 1 string amplitudes and multiple zeta values (in preparation)".
- [68] A. Tsuchiya, "More on One Loop Massless Amplitudes of Superstring Theories", Phys.Rev. D39, 1626 (1989).
- [69] S. Stieberger and T. Taylor, "NonAbelian Born-Infeld action and type 1. heterotic duality 2: Nonrenormalization theorems", Nucl. Phys. B648, 3 (2003), hep-th/0209064.
- [70] C. R. Mafra and O. Schlotterer, "The double-copy structure of one-loop open-string amplitudes", arxiv:1711.09104.
- [71] J. Broedel, N. Matthes and O. Schlotterer, https://tools.aei.mpg.de/emzv.
- [72] D. Zagier, "Periods of modular forms and Jacobi theta functions", Invent. Math. 104, 449 (1991).
- [73] K. Haberland, "Perioden von Modulformen einer Variabler und Gruppencohomologie, I", Mathematische Nachrichten 112, 245 (1983).

- [74] D. Calaque, B. Enriquez and P. Etingof, "Universal KZB equations: the elliptic case", in: "Algebra, arithmetic, and geometry: in honor of Yu. I. Manin. Vol. I", Birkhäuser Boston, Inc., Boston, MA (2009), 165–266p.
- [75] R. Hain, "Notes on the universal elliptic KZB equation", arxiv:1309.0580.
- [76] M. B. Green, J. Schwarz and E. Witten, "Superstring Theory. Vol. 2: Loop amplitudes, anomalies and phenomenology", Cambridge, UK: Univ. Pr. (1987) (Cambridge Monographs on Mathematical Physics) (1987).
- [77] S. Hohenegger and S. Stieberger, "Monodromy Relations in Higher-Loop String Amplitudes", Nucl. Phys. B925, 63 (2017), arxiv:1702.04963.
- [78] M. B. Green and J. H. Schwarz, "Supersymmetrical Dual String Theory. 3. Loops and Renormalization", Nucl. Phys. B198, 441 (1982).
- [79] P. Tourkine and P. Vanhove, "Higher-loop amplitude monodromy relations in string and gauge theory", Phys. Rev. Lett. 117, 211601 (2016), arxiv:1608.01665.
- [80] E. D'Hoker and M. B. Green, "Zhang-Kawazumi Invariants and Superstring Amplitudes", Journal of Number Theory 144, 111 (2014), arxiv:1308.4597.
- [81] E. D'Hoker, M. B. Green, B. Pioline and R. Russo, "Matching the D⁶R⁴ interaction at two-loops", JHEP 1501, 031 (2015), arxiv:1405.6226.
- [82] E. D'Hoker, M. B. Green and B. Pioline, "Higher genus modular graph functions, string invariants, and their exact asymptotics", arxiv:1712.06135.
- [83] C. M. Hull and P. K. Townsend, "Unity of superstring dualities", Nucl. Phys. B438, 109 (1995), hep-th/9410167.
- [84] M. B. Green and M. Gutperle, "Effects of D instantons", Nucl. Phys. B498, 195 (1997), hep-th/9701093.
- [85] M. B. Green, H.-h. Kwon and P. Vanhove, "Two loops in eleven-dimensions", Phys. Rev. D61, 104010 (2000), hep-th/9910055.
- [86] M. B. Green and P. Vanhove, "Duality and higher derivative terms in M theory", JHEP 0601, 093 (2006), hep-th/0510027.
- [87] M. B. Green, S. D. Miller and P. Vanhove, "SL(2,ℤ)-invariance and D-instanton contributions to the D⁶R⁴ interaction", Commun. Num. Theor. Phys. 09, 307 (2015), arxiv:1404.2192.
- [88] V. G. Drinfeld, "Quasi-Hopf algebras", Algebra i Analiz 1, 114 (1989).
- [90] T. Le and J. Murakami, "Kontsevich's integral for the Kauffman polynomial", Nagoya Math J. 142, 93 (1996).
- [91] G. Racinet, "Doubles mélanges des polylogarithmes multiples aux racines de l'unité", Publ. Math. Inst. Hautes Études Sci., 185 (2002).